Det här verket har digitaliserats vid Göteborgs universitetsbibliotek. Alla tryckta texter är OCRtolkade till maskinläsbar text. Det betyder att du kan söka och kopiera texten från dokumentet. Vissa äldre dokument med dåligt tryck kan vara svåra att OCR-tolka korrekt vilket medför att den OCR-tolkade texten kan innehålla fel och därför bör man visuellt jämföra med verkets bilder för att avgöra vad som är riktigt.

This work has been digitized at Gothenburg University Library. All printed texts have been OCR-processed and converted to machine readable text. This means that you can search and copy text from the document. Some early printed books are hard to OCR-process correctly and the text may contain errors, so one should always visually compare it with the images to determine what is correct.



GÖTEBORGS UNIVERSITET göteborgs universitetsbibliotek

DOKTORSAVHANDLINGAR vid CHALMERS TEKNISKA HÖGSKOLA

Nr 32

CONTRIBUTIONS TO THE DYNAMICS OF FLEXIBLE STRINGS

With particular Attention to the Steady Rotatory Motion of the Elastic String

BY

ERLAND YHLAND



GÖTEBORG 1961





CONTRIBUTIONS TO THE DYNAMICS OF FLEXIBLE STRINGS

With particular Attention to the Steady Rotatory Motion of the Elastic String

AV

ERLAND YHLAND

AKADEMISK AVHANDLING

SOM MED TILLSTÅND AV CHALMERS TEKNISKA HÖGSKOLA FÖR TEKNOLOGIE DOKTORSGRADS VINNANDE TILL OFFENTLIG GRANSKNING FRAMLÄGGES Å FÖRELÄSNINGS-SALEN FÖR FYSIK, GIBRALTARGATAN 5 B, GÖTEBORG, LÖRDAGEN DEN 17 JUNI 1961 KL. 10

GÖTEBORG Elanders boktryckeri aktiebolag 1961

DOKTORSAVHANDLINGAR vid CHALMERS TEKNISKA HÖGSKOLA

CONTRIBUTIONS TO THE DYNAMICS OF FLEXIBLE STRINGS

With Particular Attention to the Steady Rotatory Motion of the Elastic String

 $\mathbf{B}\mathbf{Y}$

ERLAND YHLAND



CHALMERS UNIVERSITY BOOKS/GUMPERTS. GÖTEBORG GYLDENDALSKE BOGHANDEL/NORDISK FORLAG, KØBENHAVN AKATEEMINEN KIRJAKAUPPA/AKADEMISKA BOKHANDELN, HELSINGFORS WILLIAM HEINEMANN LTD, LONDON, MELBOURNE, TORONTO

SCANDINAVIAN UNIVERSITY BOOKS

Gyldendalske Boghandel / Nordisk Forlag, København Svenska Bokförlaget / P. A. Norstedt & Söner — Albert Bonnier, Stockholm Akademiförlaget / Gumperts, Göteborg Akateeminen Kirjakauppa / Akademiska Bokhandeln, Helsingfors William Heinemann Ltd, London, Melbourne, Toronto

> GÖTEBORG 1961 ELANDERS BOKTRYCKERI AKTIEBOLAG

Acknowledgements

The more important parts of the present thesis have been written during the time I have held the appointment as assistant lecturer in the Institution of Classical Mechanics in the Chalmers University of Technology. Professor ADOLF ANZELIUS was then head of the department and I should wish to express my deep gratitude to him for the encouragement he has given me.

Mr. JAN PETERSSON, Lecturer at the Chalmers University of Technology, has helped me a great deal, particularly on points of mathematics, and I should like to express my thanks.

Mr. ESKIL STERZEL, Fil. Lic., has revised the English translation with patience and care.

My thanks are also due to Dr. ERIK HEMLIN, Chief Librarian of the Chalmers University Library, and to his staff for their helpfullness and courtesy.

To Miss IRMGARD MOLIN and Miss TULLAN LARSSON my thanks are due for their work in transcribing my manuscript.

To my wife I can only try to express my gratitude for her help in my work.

A SERVICE PROPERTY OF A

And a second second

with path car and the f

in the second state of the second s

The Mine Service of the Service Street Service of the Service Service

and and willing a new solidy are deviced as an an electronic and the second second second second second second In the works

Introduction

The present thesis is devoted to a theoretical study of some problems concerning the dynamics of strings.* Preponderant interest is centred on a particular class of problems related to the steady rotatory motion of an elastic string. The study of such motions is of importance in textile mechanics.

The treatment is theoretical in the sense that no experiments are made in order to verify the results, and no particular attention has been paid to problems which may have greater interest from the practical point of view. Although admittedly highly desirable, a systematic investigation of such problems would be very voluminous as well as laborious. It should also be taken into consideration that theoretical works in this field are rare, indeed, some themes discussed here have not been treated previously.

No serious endeavours have been made to attain a high degree of concentration, neither has the author's efforts primarily been directed towards making the exposition mathematically rigorous at every phase. In some places a more penetrating analysis might be expected and there the author has duly commented on this fact.

The thesis is divided into six chapters and two appendices.

In the first chapter the author derives the basic equations related to arbitrary motions of a *homogeneous* and *linearly elastic* string. These equations are valid in an arbitrary curvilinear coordinate system in space. Tensor notation is used to some extent.

The second chapter is divided into two main parts. The first part contains a brief discussion of the mathematical character of the basic equations. These equations constitute a system of quasi-linear partial differential equations which are shown to be hyperbolic for the elastic string, while they are non-hyperbolic, if the string is inextensible. The second part of the chapter is devoted to a short discussion of discontinuities propagating along the string.

^{*} By a string we understand here a one-dimensional, continuous medium of perfect flexibility. The terms 'thread' and 'filament' are also common in the literature.

In the third chapter the basic equations of string motion are specialized to be valid in a system of cylindrical coordinates. Dimensionless variables are introduced, and a linearization process is outlined for a small time-dependent motion superimposed on a steady rotatory string motion; it is assumed that no external forces do act on the string.

The fourth and fifth chapters contain the principal results obtained by the author. These chapters are devoted to the study of inextensible and elastic strings in steady rotatory motion, in the absence of external forces. A more precise description of the problems treated is given infra.

In Chapter six we consider a time-dependent motion in the form of small vibrations superimposed on a simple kind of steady rotatory motion of an inextensible string. This chapter is intended as an introduction to the investigation of more general, time-dependent motions of a rotating string. Such investigations may assume an interest with reference to the stability of steady motions.

Two appendices on mathematical minutiae conclude the thesis.

As mentioned above the main subject of the work is the study of a particular class of steady rotatory string motions.

A string motion is called *steady rotatory* (or *steady rotational*), if the string moves with a velocity, the tangential velocity which does not depend on time, along a curve which is fixed in a frame of reference, the latter rotating with constant angular velocity round an axis fixed in space.

As regards the properties of the string we confine ourselves to strings that are *linearly elastic* and *homogeneous*. A string is *linearly elastic* in our sense of the term, *if the logarithmic strain depends linearly on the string tension*. In the case of small strain the results obtained are valid for most kinds of elastic strings. A homogeneous string in unstressed state has the same mass per unit length in every point. An inextensible string is obtained by putting the elastic constant equal to zero.

As to the kinds of motion treated we restrict ourselves to cases where *no external forces act on the string*. This implies that we disregard gravity and air resistance. Such limitations need no particular justification in a theoretical work like this. We also ignore the special motions which take place in a plane perpendicular to the axis of rotation.

By a string property we understand any dependent variable of the basic equations formulated in Chapter 1, or equations that can be

deduced from the aforesaid basic equations. The properties of the steady string motions specified here are governed by a system of ordinary nonlinear differential equations. The latter system is derived in Chapter 3.

From our theoretical point of view it is of interest to investigate and solve this differential system under as general conditions as possible. In this matter we are confronted with certain difficulties as there are no theorems available on the existence and uniqueness of general boundary value problems in connection with nonlinear differential systems. If we exclude string motions involving finite strain in connection with non-zero tangential velocity, however, the system will become amenable to elementary methods of integration.

In order to facilitate the determination of explicit solutions we divide the class of problems, consisting of all possible boundary value problems connected with our differential system, into a number of subclasses.

Within each subclass treated in this thesis we establish a basic problem which is representative of the class in the following sense: If a problem belonging to a certain subclass has a solution (*i. e.*, if the corresponding motion really exists), then that solution is contained in the solution of the basic problem. The basic problems are suitably chosen as one-point (initial value) problems or closely related problems. Their solutions are unique, but this circumstance does not imply uniqueness of the solution of a particular boundary value problem; in fact, several problems which possess an interest in this connection actually are eigenvalue problems.

The separation into subclasses is made according to three different bases of division.

In the first place we must distinguish between string motions that take place with *positive reduced string tension* and motions with *negative reduced string tension*. In the author's terminology the *reduced* string tension is the quantity $\overline{T}=T-mv^2$, where $T\geq 0$ is the string tension, *m* the mass of the unit length, and *v* the tangential velocity of the string. Steady rotatory motions with $\overline{T}>0$ are treated in Chapter 4, and those with $\overline{T}<0$ are briefly discussed in Chapter 5.

The second basis of division is related to the magnitude of the strain of the string and we consequently establish the following division: (i) *inextensible strings*; (ii) *elastic strings subject to small strain*; (iii) *linearly elastic strings subject to finite strain*. As concerns

the last kind of strings we confine the treatment to motions which take place without tangential velocity.

The third basis of division is related to the curve assumed by the string as observed in a co-rotating frame of reference. If this curve *intersects* the *axis of rotation*, we have a special (although important) case which must be treated separately from the *general non-intersection* case.

We must make a further division for motions with negative reduced string tension, $\overline{T} < 0$, as shown in Chapter 5.

A schematic diagram, arranged according to the subclasses enumerated above, is placed at the end of this introduction; it shows the designations of those basic problems which are solved in the thesis.

We must now briefly survey the literature in the field and in neighbouring fields. The survey follows the contents of the chapters of the thesis and is not intended to be complete.

The basic equations of inextensible and elastic strings with a fixed, rectangular Cartesian coordinate system as frame of reference are stated in several textbooks; in some of these works particular kinds of problems are discussed, see for instance BEGHIN, [B1], and ROUTH, [R2]. HAMEL, [H3], discusses the basic string equations as special cases of rod equations, and valuable information on the string theory may be extracted from the rigorous treatment of stress and strain in connection with finite deformations of rods (and shells) by ERICK-SEN and TRUESDELL, [E1].

No discussion on the transformation of the basic string equations into curvilinear frames of reference seems to exist in the literature. Neither has the author been able to find any general investigations of the mathematical nature of the basic equations or the propagation of discontinuities along elastic strings. Strong discontinuities on inextensible strings are analysed by BEGHIN, [B1], and PAILLOUX, [P1], [P3], and in connection with a particular problem by HAMEL, [H4].

A large number of papers are devoted to the study of spinning problems; a survey and bibliography containing 37 papers up to 1955 may be found in three papers by DETRY, [D1]. The factor of gravity is disregarded in all these papers on spinning theory and practice, and the string is regarded throughout as inextensible; solutions which also take air resistance into consideration, however, are of particular interest. The tangential velocity is neglected since it is of minor

importance in most spinning problems, but MACK, [M1], has shown that it is possible to solve the problems here called (I1) and (I2) by means of quadrature.

Steady rotatory motions of inextensible strings taking place in a plane perpendicular to the axis of rotation (the gravitational effects being neglected) are treated by HAMEL, [H5], and MACK, [M2].

The problem of an inextensible string intersecting the axis of rotation and being in steady rotatory motion without tangential velocity round a vertical axis is rigorously analysed by KOLODNER, [K2], whose treatment also takes gravity into account. The stability of particular motions is studied by CAUGHEY, [C5], HALL and HUNTER, [H6], KOLODNER, [K2], and NERONOFF, [N2].

A special type of vibrational motion superimposed on a steadily rotating string is discussed by STEVENSON, [S4], cf. also WRIGHT, [W1].

We shall end this introduction with a few comments on the disposition of the thesis and on particular concepts and notations that are used throughout the work.

The decimal paragraphing system is employed and the equations are numbered conformably. The footnotes are numbered consecutively throughout the thesis.

The bibliography is arranged in approximate alphabetic order and each separate work is given a code number, e. g. [B1] refers to H. BEGHIN: Cours de mécanique théorique et appliquée. These code numbers are used in the references in the text of the thesis.

Each chapter contains a list of the main symbols and conventions used in that chapter.

The O-notation is used throughout the work. A function f(x) satisfies

$$f(x) = O(\Phi(x)), x \to 0$$
, if $\frac{f(x)}{\Phi(x)}$ is bounded for $x \to 0$.

The term analytic (function) is used in the sense regular analytic. The symbol \equiv is employed in three different connotations: (i) $f(x) \equiv \equiv$ const. means that equality holds throughout the interval of defini-

tion of f(x); (ii) $u \equiv \frac{\partial h}{\partial \tau}$ that u is defined as $\frac{\partial h}{\partial \tau}$; (iii) $x' \equiv x'(u, t) \equiv \equiv (x_1, \ldots, x_9, T, u)$ merely that we use three different ways of writing (the row matrix) x'.

The symbol (a, b) signifies an open interval with a denoting the left-hand end point and b denoting the right-hand end point. [a, b] signifies the closed interval.

		HOMOGENEOUS	PERFECTLY	FLEXIBLE STRINGS
		JEXTENSIBLE TRINGS	ELASTIC STRI	NGS OTHER KINDS OF EXTENSIBLE STRINGS
			LINEARLY OTHER ELASTIC S. OF EL	ASTIC S.
			FINITE SMALL STRAIN STRAIN	EINITE STRAIN
MOTION WITH TANG	SENTIAL VELOCITY	-	-	
POSITIVE REDUCED	GENERAL CASE	► PROBLEM (11)	PRO	BLEM (E1)
STRING TENSION	STRING INTERSECTS AXIS OF ROTATION	► PROBLEM (12)	PRO	BLEM (E2)
NEGATIVE REDUCED	GENERAL CASE	1		
STRING TENSION	STRING INTERSECTS AXIS OF ROTATION	PROBLEMS (IN2a),(IN2b)		
MOTION WITHOUT T	ANGENTIAL VELOCITY			
STRING TENSION =	GENERAL CASE	PECIAL CASE OF (11)	PROBLEM (E3) SPE	CIAL CASE (E1)
TENSION	STRING INTERSECTS AXIS OF ROTATION	PECIAL CASE P	PROBLEM (E4) OF	CIAL CASE (E2)
Schematic diagram showing	those problems concerning ste	ady rotatory motion of v	arious kinds of flexible	strings that are treated in the thesis.

Broken rectangles indicate problems that have not been the objects of treatment.

Contents

	Page
Acknowledgements	. 3
Introduction	. 5
Contents	. 11
1. The Basic Equations of String Motions	. 13
1.1. Description of the String and its Motion	. 14
1.2. Kinematics and Kinetics of the String	. 16
1.3. The Flow of String Mass Through a Surface	. 26
1.4. The String Motion in a Fixed Path	. 28
2. Discussion of the String Equations	. 30
2.1. Classification of the String Equations	. 31
2.2. The Propagation of Discontinuities Along the String	. 38
3. The String Equations in Cylindrical Coordinates	. 47
3.1. Dimensionless Variables	. 48
3.2. String Equations for a Small Motion Dependent on Time and Superimposed on a Steady Rotatory Motion; External Forces being Abse	r- nt 52
4. The Steady Rotatory Motion of the Linearly Elastic String in the Case of Positive Reduced String Tension and in the Absence	ne of
External Forces	. 59
4.1. Introduction	. 59 of
Positive Reduced String Tension	. 64
4.5. The Steady Rotatory Rotatory and Small Strain	. 74
4.4. The Steady Rotatory Motion Without Tangential Velocity of Elastic String	an 87
5. The Steady Rotatory Motion of an Inextensible String in the Ca of Negative Reduced String Tension and in the Absence of E	se x-
ternal Forces	93
5.1. Introduction	93
5.2. The String Equations. The Two Cases	94

6.	A Vibration Problem of the Inextensible String	100
	6.1. Introduction	102
	6.2. Transformation of the Basic System. The Vibration Problem	104
	6.3. Outline of a Perturbation Method	108
Ap	opendices	120
Bi	bliography	125

1. The Basic Equations of String Motions

Symbols and conventions:

$y^{i}, i=1,2,3$	Rectangular, right-handed, Cartesian coordinates, fixed in space. Coordinates of the string points.
$x^{i}, i = 1, 2, 3$	Arbitrary, curvilinear coordinates, fixed in space. Coordinates of the string points.
σ	Coordinate along the curve of the string. $\sigma = \text{const.}$ denotes a particular string point.
τ, t	Time coordinates.
8	Arc length of the string curve.
S	Surface generated by the string during its motion.
v^i	Velocity vector of the string point.
f^i	Acceleration vector of the string point.
z	Independent variable, defined by Eq. $(1.2-2)$.
u, q	String properties, defined by Eq. $(1.2-4)$.
m	Mass of the unit length of the string.
m_0	Mass of the unit length of the string for $T=0$.
T	String tension (force).
k	Elastic constant, see Eq. $(1.2-18)$.
A	Arbitrary string property, scalar or vector.
A^i	Arbitrary, contravariant string property.
g_{ik}	Metric tensor of the curvilinear coordinate system x^i .
$egin{pmatrix} i \ km \end{pmatrix}$	Christoffel symbol of the second kind.
δ^p_k	Kronecker symbol.
$F^i = F_i$	Rectangular Cartesian components of the external force acting on the unit length of the string.

- K^i, K_i Contravariant and covariant components of the external force in a curvilinear coordinate system.
- *H* Quantity of string mass passing per unit of time through a particular surface $F(y^i, t)=0$ which is fixed in space or moving through space.

Symbols not included in the above list will occur now and then in the text. All indices have the range of values 1, 2, 3. The summation convention is used. Concerning the tensor notation see for instance [M3].

1.1. Description of the String and its Motion

In this section we shall give a description of the fundamental properties of the string and the motions that we are going to study in this work.

1.11. The String

The string is defined and various assumptions concerning it are made as enunciated in the following points.

1) A point of the string, or *string point*, is a point which is fixed with respect to the material of the string. (A string point consequently follows a certain 'molecule' of the string.)

2) The string is assumed to have zero thickness and its mass is therefore distributed along a curve, the string curve, consisting of the string points.

3) An *element* of the string consists of the string material carried by an arc element of the string curve. The definitions of the length and the mass of the string element are self-evident.

4) The string is *perfectly flexible* and the state of stress in a string point is completely determined by *the string tension* which is directed along the tangent of the string curve. It is postulated that *the string tension cannot be negative*, *i. e.* the string cannot sustain compressive forces.

5) The string is *elastic*, *i. e.* the length of the string element is solely determined by the string tension. We shall in general confine the treatment to a *linearly elastic string*, *i. e.* to the case of a linear relationship

between strain (defined in a certain manner, see section 1.25) and the string tension.

6) The string is *homogeneous*, i. e. the mass of the unit length is the same in every point, when the string is in unstressed state.

1.12. The Motion of the String

y1

1) When a string (of finite or infinite length) moves through space, it generates a surface S. Let the equation of S (in a rectangular, Cartesian coordinate system) be

$$y^{i} = y^{i}(\sigma, \tau).$$
 (1.1-1)

Let the independent variable τ be the time. It is then clear that a curve on the surface S with the equation $y^i = y^i(\sigma, \tau_0), \tau_0 = \text{const.}$, is the string curve at the time τ_0 .

Let the curve $y^i = y^i(\sigma_0, \tau)$, σ_0 = arbitrary constant, be the path of a certain string point; the velocity vector of that string point is then directed along the tangent to the curve $y^i = y^i(\sigma_0, \tau)$. Let us specify σ as the unstretched length of the string measured from some arbitrary string point. It is obvious that the independent variables σ and τ constitute a coordinate system on the surface S.

2) In general we shall assume that there exists no point on S in which the tangent vectors to the coordinate lines $\sigma = \text{const.}$ and



Fig. I. Schematic sketch showing part of the surface S generated by a string, and the coordinate lines $\sigma = \sigma_0$ and $\tau = \tau_0$ on S. The velocity vector v^i of the string point $\sigma = \sigma_0$ at the time $\tau = \tau_0$ is also shown.

 τ =const. are collinear. This assumption implies that we exclude cases where adjacent points of the string have the same path. When every point of the string describes the same path we have the kind of motion that is discussed in section 1.4.

3) We shall assume that the functions $y^i(\sigma, \tau)$ possess sufficient properties of continuity. They must in general be twice continuously differentiable with respect to σ and τ . Along discrete lines on S, however, there may appear *discontinuities* of different kinds, *cf*. Ch. 2.

4) Later on we shall use other surface coordinates than σ , τ and also introduce curvilinear space coordinates instead of the rectangular, Cartesian coordinates y^i used so far.

1.2. Kinematics and Kinetics of the String

It is the purpose of this section to establish a system of equations for the linearly elastic, homogeneous string. The system will be complete in the sense that we can deduce from it any equation which, in terms of mechanics, has a meaning for such a string.¹ Naturally, the string may also have to satisfy initial and boundary conditions. The mentioned system can be formulated in different ways. We shall start from the expressions for the velocity and acceleration vectors with σ, τ as independent variables and a rectangular, Cartesian system as the frame of reference. The corresponding equations of motion and kinematical conditions are well known and may be found in the standard textbooks, e.g. [B1], [R2], [S3]. We next introduce a new set of surface coordinates z, t on S and use them as independent variables, and then a system of curvilinear coordinates as a frame of reference in space. Then we shall discuss the elastic properties of the string and finally put down the equations of motion and gather our expressions in a system of equations which will be called the string equations.

The transformation into curvilinear coordinates may be carried out in conventional terms of the intrinsic geometry of a surface embedded in a three-dimensional Euclidean space, cf. [M3]. For later applications, however, we will find it preferable to employ a more specialized technique.

¹ In certain points of discontinuity conditions obtain which cannot be deduced from our complete system, cf. Ch. 2.

1.21. Velocity and Acceleration Vectors in Lagrangian Formulation

The velocity vector v^i and the acceleration vector f^i of the string point are determined by the expressions

$$v^{i} = v^{i}(\sigma, \tau) = \frac{\partial y^{i}}{\partial \tau},$$

$$f^{i} = f^{i}(\sigma, \tau) = \frac{\partial^{2} y^{i}}{\partial \tau^{2}}.$$
(1.2-1)

where the variables are the same as in section 1.1.

1.22. A New Set of Independent Variables

Let z and t be coordinates on S defined by

$$z = h(\sigma, \tau),$$

$$t = \tau.$$
(1.2-2)

We shall assume that this transformation is one to one, and that h is at least twice continuously differentiable with respect to σ and τ . We also assume that

$$\frac{\partial(z,t)}{\partial(\sigma,\tau)} = \frac{\partial h}{\partial\sigma} \neq 0. \tag{1.2-3}$$

From now on we shall regard every property of the string, i. e. every dependent variable, as a function of z and t.

Let q=q(z, t), u=u(z, t) be string properties defined by

$$\begin{array}{l} \frac{\partial h}{\partial \sigma} \equiv \frac{m_0}{mq} \,, \\ \\ \frac{\partial h}{\partial \tau} \equiv u \,, \end{array} \tag{1.2-4}$$

where m = m(T) = m(T(z, t)) is the mass of the unit length of the string. For an elastic and homogeneous string m depends explicitly of the string tension T only, and m_0 is the value of m when T=0. The physical interpretations of q and u depend on the choice of h and will be discussed later.



Fig. II. Sketch illustrating the significance of the transformation Eq. (1.2-2) from the surface coordinates σ, τ into the surface coordinates z, t.

We now get the following expressions for the transformation of the derivatives of an arbitrary string property A (scalar or vector)

$$\frac{\partial A}{\partial \sigma} = \frac{m_0}{mq} \frac{\partial A}{\partial z},$$

$$\frac{\partial A}{\partial \tau} = \frac{\partial A}{\partial t} + u \frac{\partial A}{\partial z} \equiv \frac{dA}{dt},$$
(1.2-5)

where, from the mechanics of continuous media, we adopt the term substantial time derivative, symbolized by $\frac{d}{dt}$. From Eq. (1.2–1) we obtain the following expressions for the velocity and acceleration vectors of the string point

$$v^{i} = \frac{\partial y^{i}}{\partial t} + u \frac{\partial y^{i}}{\partial z} = \frac{dy^{i}}{dt},$$

$$f^{i} = \left(\frac{\partial}{\partial t} + u \frac{\partial}{\partial z}\right) \left(\frac{\partial y^{i}}{\partial t} + u \frac{\partial y^{i}}{\partial z}\right) = \frac{d^{2}y^{i}}{dt^{2}}.$$
 (1.2-6)

It may be observed that the substantial time derivative generally does not commute with the partial derivatives.

We may remark that if, for some special string problem, we know the string properties m, q, u as functions of z and t, we naturally can also determine the paths of the string points, *i. e.* the curves $\sigma = \text{const.}$ on the surface S. Let the inverse transformation to Eq. (1.2-2) be

$$\sigma = g(z, t),$$

$$\tau = t.$$

Then, from the identities in Eq. (1.2-4), we have

$$\frac{\partial g}{\partial t} = -\frac{mqu}{m_0}.$$

The right member of this differential equation is a known function of z and t, and together with an initial condition, for instance,

$$g(z, t_0) = g_0(z)$$

the differential equation determines the function g. The path of an arbitrary string point $\sigma = \sigma_1$ is then the curve $g(z, t) = \sigma_1$.

We shall now discuss the physical interpretations of the string properties q and u.

Let s be the arc length of the string curve measured from an arbitrary point. By the definition of σ we then obtain the following equation for the mass of a string element of the length ds

$$mds = m_0 d\sigma. \tag{1.2-7}$$

Studying the change of an arbitrary string property A due to a small displacement along the string curve we obtain the equality

$$\frac{\partial A}{\partial \sigma} \, d\sigma = \frac{\partial A}{\partial z} \, dz.$$

Together with Eq. (1.2-5) this yields $d\sigma = \frac{mq}{m_o} dz$; and from Eq. (1.2-7) we then get

$$dz = \frac{1}{q} ds. \tag{1.2-8}$$

Observing that z is a coordinate along the string curve (t=const.), we may interpret q as a measure of the longitudinal scale along that curve.

$$q(z, t_0) = q_0(z)$$



Fig. III. The decomposition of the velocity v^i into components along the coordinate lines z=const. and t=const. The arrows along the coordinate lines indicate positive directions of the coordinates, and in the case reproduced here q and u are both assumed to be positive quantities.

By the definition of s the unit tangent vector along the string curve is $\frac{\partial y^i}{\partial s} = \frac{1}{q} \frac{\partial y^i}{\partial z}$. From the expressions for the velocity vector v^i in Eq. (1.2-6) we can then conclude that uq may be interpreted as the component of the velocity vector along the tangent to the string curve (cf. Fig. III).

Finally we consider two special choices of the coordinate z, *i. e.* the function $h(\sigma, \tau)$.

1) Let z be the arc length of the string curve, *i. e.* let $z \equiv s$. Choose an arbitrary curve on the surface S which does not touch any string curve and let this curve have the equation $y^i = y^i(s_0, t), s_0 = \text{const.}$ Provided the motion is known, every curve s = const. on S is then determined. From Eq. (1.2-8) we have

$$q = q(z, t) \equiv 1$$
 (1.2-9)

and u=u(z, t) is then the component of the velocity vector along the string tangent.

2) Let z be one of the coordinates y^i , say y^3 , and let the string curves nowhere touch a plane $y^3 = \text{const.}$ Then the curves $y^i = y^i(z, t)$, z = const., are lines of intersection between the planes $y^3 = \text{const.}$ and the surface S. We get

$$\frac{\partial y^3}{\partial z} \equiv 1, \ \frac{\partial y^3}{\partial t} \equiv 0.$$
 (1.2–10)

According to Eq. $(1.2-8)\frac{1}{q}$ becomes the direction cosine of the tangent to the string curve with respect to the y^3 axis. By perpendicular projection on the y^i axes we obtain u as the y^3 component of the velocity vector.

1.23. Kinematical Conditions

Let z, t be the same set of coordinates on the surface S as previously. Let s, t represent this set when z=s is chosen as the arc length of the string curve. Since $y^i=y^i(s,t)$ is the equation of the surface S, $\frac{\partial y^i}{\partial s}$ is the unit tangent vector to the string curve and the equations

$$\frac{\partial y^{i}}{\partial s} \frac{\partial y^{i}}{\partial s} = 1, \qquad (1.2-11)$$

$$\frac{\partial y^{i}}{\partial s} \frac{\partial^{2} y^{i}}{\partial s^{2}} = \frac{\partial y^{i}}{\partial s} \frac{\partial^{2} y^{i}}{\partial t \partial s} = 0, \qquad (1.2-12)$$

$$\frac{\partial y^{i}}{\partial s} \frac{d}{dt} \left(\frac{\partial y^{i}}{\partial s}\right) = 0$$

hold everywhere on the surface S, provided that the derivatives exist. Eq. (1.2–12) imply that $\frac{\partial y^i}{\partial s}$ is perpendicular to its own derivative in an arbitrary direction on S. Instead of Eq. (1.2–11) we get in the general case

$$\frac{\partial y^i}{\partial z} \frac{\partial y^i}{\partial z} = q^2. \tag{1.2-13}$$

There is another kinematical condition. The string properties q and u are both derived from the function $h=h(\sigma,\tau)$ which defines the curves z=const. on S, cf. Eq. (1.2-4). By hypothesis, the partial derivatives of h with respect to σ and τ commute, and by means of the expressions (1.2-5) we get

$$\frac{\partial(mq)}{\partial t} + \frac{\partial(mqu)}{\partial z} = 0. \tag{1.2-14}$$

In the special case of z=s we have $q\equiv 1$ and Eq. (1.2-14) will then be slightly simplified. Eq. (1.2-14) corresponds to the equation of continuity for a general continuous medium. It can be derived directly from a kinematical study of the motion of a string element during a small interval of time.

1.24. Curvilinear Coordinates

In the preceding sections we have studied the motion of the string with a rectangular Cartesian system (coordinates designated y^i , i=1,2,3) as a frame of reference in space. We now introduce a general, curvilinear system with the coordinates x^i , i=1,2,3, and the metric tensor g_{ik} , cf. [M3]. We transform the expressions (1.2-6) for the velocity and acceleration vectors and the kinematical conditions (1.2-13) and (1.2-14) to be valid in a general curvilinear coordinate system. For the contravariant tangent vectors $\frac{\partial y^i}{\partial z}$ and $\frac{\partial y^i}{\partial t}$ we have

$$\frac{\partial y^{i}}{\partial z} \rightarrow \frac{\partial x^{i}}{\partial z},$$

$$\frac{\partial y^{i}}{\partial t} \rightarrow \frac{\partial x^{i}}{\partial t},$$
(1.2–15)

where \rightarrow means: corresponds to.

For the partial derivatives of an arbitrary contravariant vector with space components A^i , defined on the surface S, we obtain

$$\frac{\partial A^{i}}{\partial z} \rightarrow \frac{\delta A^{i}}{\delta z} \equiv \frac{\partial A^{i}}{\partial z} + \begin{cases} i\\ km \end{cases} A^{k} \frac{\partial x^{m}}{\partial z},$$

$$\frac{\partial A^{i}}{\partial t} \rightarrow \frac{\delta A^{i}}{\delta t} \equiv \frac{\partial A^{i}}{\partial t} + \begin{cases} i\\ km \end{cases} A^{k} \frac{\partial x^{m}}{\partial t},$$
(1.2-16)

where $\frac{\delta A^{i}}{\delta z}$ is called the intrinsic derivative of A^{i} with respect to z. In Eq. (1.2–16) $\binom{i}{km}$ is the CHRISTOFFEL symbol of the second kind

defined by

$$egin{cases} i\ km \end{pmatrix} = rac{1}{2} g^{ip} igg(rac{\partial g_{mp}}{\partial x^k} + rac{\partial g_{kp}}{\partial x^m} - rac{\partial g_{km}}{\partial x^p} igg),$$

where the conjugate metric tensor g^{ip} is defined by

$$g_{pi}g^{ik} = \delta_p^k = \begin{cases} 1 & \text{if } k = p \\ 0 & \text{if } k \neq p \end{cases}.$$

and δ_p^k is the KRONECKER symbol.

Scalar functions defined on S are not affected by the change of reference system in space. We therefore get the following expressions corresponding to Eqs. (1.2-6), (1.2-13) and (1.2-14)

$$\begin{aligned} v^{i} &= \frac{dx^{i}}{dt} = \frac{\partial x^{i}}{\partial t} + u \frac{\partial x^{i}}{\partial z}, \\ f^{i} &= \frac{\partial v^{i}}{\partial t} + u \frac{\partial v^{i}}{\partial z} = \frac{\partial v^{i}}{\partial t} + u \frac{\partial v^{i}}{\partial z} + \begin{Bmatrix} i \\ km \end{Bmatrix} v^{k}v^{m}, \\ g_{ik} \frac{\partial x^{i}}{\partial z} \frac{\partial x^{k}}{\partial z} = q^{2}, \\ \frac{\partial (mq)}{\partial t} + \frac{\partial (mqu)}{\partial z} = 0. \end{aligned}$$
(1.2-17)

1.25. Conditions Governing the Properties of an Elastic String

So far we have not made any assumptions concerning the elastic properties of the material in the string.

We shall assume that the string is in a state of simple tension and confine ourselves to the case of a linearly elastic string, which we shall define as a string in which the strain is proportional to the string tension. It may be remarked that owing to the lateral contraction of the string the stress-strain relation in general will not be linear.

By the strain we shall understand the logarithmic or natural strain, cf. [H1], p. 9. The strain is generally assumed to be finite, and some of the results obtained later may be extended to apply to a wider range of elastic strings. For an element of the length l of a linearly elastic string subjected to the tension T we have by definition

$$l = l_0 \exp(kT), \ m = m_0 \exp(-kT), \ (1.2-18)$$

where l_0 and m_0 are the length and the mass per unit length, respectively, of the unstretched element; $k \ge 0$ is the elastic constant of the string and k=0 means that the string is inextensible.²

From Eq. (1.2-18) we have

$$l = l_0 (1 + kT) + O((kT)^2), \quad m = m_0 (1 - kT) + O((kT)^2), \quad kT \to 0. \quad (1.2 - 19)$$

Neglecting terms $O((kT)^2)$ we get the classical expressions valid for small strain.

1.26. Equations of Motion. The Complete Equations

Taking s to be the arc length of the string curve, as we have done previously, we get the following wellknown equation of motion for the string element

$$\frac{\partial}{\partial s} \left(T \frac{\partial y^i}{\partial s} \right) - m f^i + F^i = 0, \qquad (1.2 - 20)$$

where T is the string tension, f^i the acceleration vector, F^i the external force on the unit length of the string,³ and m the mass of the unit length. y^i are rectangular, Cartesian coordinates in space. From sections 1.22, 1.23 and 1.25 we have expressions for f^i , the kinematical conditions and the conditions valid for a linearly elastic string.

Gathering all these equations into a single system we get the following equations for a linearly elastic string

$$\frac{\partial}{\partial s} \left(T \frac{\partial y^{i}}{\partial s} \right) - m \frac{d^{2}y^{i}}{dt^{2}} + F^{i} = 0,$$

$$\frac{\partial y^{i}}{\partial s} \frac{\partial y^{i}}{\partial s} = 1,$$

$$\frac{\partial m}{\partial t} + \frac{\partial (mu)}{\partial s} = 0,$$

$$m = m_{0} \exp(-kT).$$
(1.2-21)

This system is complete in the sense discussed in the beginning of section 1.2. System (1.2-21) consists of five (scalar) partial differential

² The elastic properties of the material of the string cannot, of course, be completely described by a single constant.

³ If the external force on the unit mass of the string is P^{i} , we must replace F^{i} by mP^{i} .

equations and one transcendental equation for the six dependent variables y^i , T, u, m, and we have s, t as independent variables. By means of inner multiplication by $\frac{\partial y^i}{\partial s}$ in the first equation of the system and taking into consideration the conditions of Eq. (1.2-12) we get the following equations

$$\frac{\partial T}{\partial s} - m \left(\frac{du}{dt} + \frac{\partial y^{i}}{\partial s} \frac{\partial^{2} y^{i}}{\partial t^{2}} \right) + F^{i} \frac{\partial y^{i}}{\partial s} = 0,$$

$$\frac{\partial}{\partial s} \left(T - mu^{2} \right) - \frac{\partial}{\partial t} \left(mu \right) - m \frac{\partial y^{i}}{\partial s} \frac{\partial^{2} y^{i}}{\partial t^{2}} + F^{i} \frac{\partial y^{i}}{\partial s} = 0.$$
(1.2-22)

These two equations are equivalent and are equations of motion in the direction of the string tangent.

We can, of course, formulate the equations (1.2-21) with z and t as independent variables (s simply means a particular choice of z). We then obtain

$$\begin{aligned} \frac{\partial}{\partial z} \left(\frac{T}{q} \frac{\partial y^{i}}{\partial z} \right) &- mq \frac{d^{2}y^{i}}{dt^{2}} + qF^{i} = 0, \\ \frac{\partial y^{i}}{\partial z} \frac{\partial y^{i}}{\partial z} = q^{2}, \\ \frac{\partial (mq)}{\partial t} &+ \frac{\partial (mqu)}{\partial z} = 0, \\ m = m_{0} \exp(-kT). \end{aligned}$$
(1.2-23)

Finally we transform the string equations (1.2-23) to be valid in a general, curvilinear coordinate system x^i .

By means of the expressions (1.2-16) and (1.2-17) we get

$$\frac{\delta}{\delta z} \left(\frac{T}{q} \frac{\partial x^{i}}{\partial z} \right) - mq \left(\frac{\delta}{\delta t} + u \frac{\delta}{\delta z} \right) \left(\frac{\partial x^{i}}{\partial t} + u \frac{\partial x^{i}}{\partial z} \right) + qK^{i} = 0,$$

$$g_{ik} \frac{\partial x^{i}}{\partial z} \frac{\partial x^{k}}{\partial z} = q^{2},$$

$$(1.2-24)$$

$$\frac{\partial (mq)}{\partial t} + \frac{\partial (mqu)}{\partial z} = 0,$$

$$m = m_{0} \exp\left(-kT\right).$$

In this system the contravariant force components, K^i , are determined from the Cartesian components $F^i = F_i$ by means of the expression

$$K^{i} = F^{p} \frac{\partial x^{i}}{\partial y^{p}} = g^{ki} F_{p} \frac{\partial y^{p}}{\partial x^{k}}. \qquad (1.2-25)$$

Essentially, the systems (1.2-23) and (1.2-24) possess the same mathematical properties. The following comments on system (1.2-24), with obvious modifications, are consequently valid also for system (1.2-23). System (1.2-24) consists of five (scalar) partial differential equations and one transcendental equation for seven (scalar) dependent variables: x^i , T, m, q, u. There are two independent variables z and t. We have one more dependent variable than equations, but we are free to choose the curves z=const. in an arbitrary manner on the surface S, generated by the string in its motion.

In the following considerations the dependent variables will generally be called *string properties*, and the systems (1.2-24) or (1.2-23) will be called *the string equations*. These equations naturally do not yield unique solutions of problems concerning the motions of a string. In addition to the string equations we must possess a suitable system of initial values and boundary conditions.

The questions of the existence and uniqueness of the solutions to the system (1.2-24) present difficult problems which sometimes become even more complicated by the presence of strong discontinuities in the string properties. A brief discussion of the string equations, starting from system (1.2-21), is given in Chapter 2.

1.3. The Flow of String Mass Through a Surface

Sometimes it is of interest to determine the quantity of string mass passing through a particular surface in space during the motion of a string. The surface may be fixed or it may be moving through space.

1) Let us consider a surface D in a rectangular, Cartesian system in space, and let t be the time. Let the equation of D be

$$F(y^i, t) = 0,$$
 (1.3-1)

where F is continuously differentiable with respect to y^i and t. Let h^i be the velocity vector of a point P on D at the time t, and n_i a normal unit vector of D at P.⁴

⁴ It should be observed that Eq. (1.3-1) does not completely determine h^i but only the component along n_i . There is here no necessity of a more precise statement.

2) Let the positive side of D, at the time t, be defined as the set consisting of all the points y^i in space that satisfy the inequality F > 0. Let the negative side of D be defined accordingly.

3) Let v^i be the velocity vector of a string point coincident with P at the time t.

From Eq. (1.3-1) we then obtain

$$\frac{\partial F}{\partial y^i} h^i + \frac{\partial F}{\partial t} = 0, \qquad (1.3-2)$$

$$n_i = \frac{\partial F}{\partial y^i} \left(\frac{\partial F}{\partial y^k} \frac{\partial F}{\partial y^k} \right)^{-1/2}.$$
 (1.3-3)

As usual we take s to be the arc length of the string curve, and $y^i = y^i(s,t)$ to be the equation of the surface S, generated by the motion of the string. We shall assume that the string does not touch the surface D in the point P, and we then obtain

$$n_i \frac{\partial y^i}{\partial s} \neq 0. \tag{1.3-4}$$

The quantity of string mass H passing through the surface D in the vicinity of P per unit of time is determined by

$$H = m \left(n_i \frac{\partial y^i}{\partial s} \right)^{-1} n_k (v^k - h^k).$$
(1.3-5)

Combining the last equation with Eqs. (1.3-2) and (1.3-3) we get

$$H = m \left(\frac{\partial F}{\partial y^i} \frac{\partial y^i}{\partial s} \right)^{-1} \left(v^k \frac{\partial F}{\partial y^k} + \frac{\partial F}{\partial t} \right).$$

We next write the equation of the surface S in the form $y^i = y^i(z,t)$ with z as in section 1.22. Finally, adopting a general, curvilinear coordinate system x^i , and writing the equation of the surface D as $F(x^i,t)=0$, we get

$$H = mq \left[u + \left(\frac{\partial F}{\partial x^i} \frac{\partial x^i}{\partial z} \right)^{-1} \left(\frac{\partial F}{\partial x^k} \frac{\partial x^k}{\partial t} + \frac{\partial F}{\partial t} \right) \right], \qquad (1.3 - 6)$$

where by Eq. (1.3-4) $\frac{\partial F}{\partial x^i} \frac{\partial x^i}{\partial z} \neq 0$. It should be observed that H may have positive or negative sign. If we choose the arc length s in

such a manner that it increases on passing from the negative to the positive side of D, then $n_i \frac{\partial y^i}{\partial s} > 0$ and H > 0 means that string mass is flowing from the negative into the positive side of D.

1.4. The String Motion in a Fixed Path

We remarked in section 1.12, point (2) that the case in which the string moves in a fixed path must be discussed separately. We now consider such a motion in a rectangular, Cartesian system y^i , fixed in space, and take s to be the arc length of the string curve measured from a fixed point. Using the same notations as previously we have the following string properties:

 $y^i = y^i(s)$ equation of the string curve; u = u(s,t) velocity of the string point, directed along the tangent vector $\frac{\partial y^i}{\partial s}$;

T = T(s,t) string tension; m = m(s,t) mass of the unit length of the string.

Taking $\frac{\partial y^i}{\partial t} \equiv 0$ into consideration, simple reflection will indicate that we may make use of the string equations (1.2-21). Rewriting the first of the latter equations we get

$$\begin{split} & \frac{\partial}{\partial s} \left[(T - mu^2) \frac{\partial y^i}{\partial s} \right] + \frac{\partial (mu)}{\partial t} \frac{\partial y^i}{\partial s} + F^i = 0, \\ & \frac{\partial y^i}{\partial s} \frac{\partial y^i}{\partial s} = 1, \\ & \frac{\partial m}{\partial t} + \frac{\partial (mu)}{\partial s} = 0, \\ & m = m_0 \exp\left(-kT\right). \end{split}$$
(1.4-1)

We shall not attempt a general discussion of this system but merely make a few comments.

1) If the external force F^i on the unit length of the string does not depend on time, and if the string is inextensible, *i. e.* k=0, then the motion may take place with a constant velocity u. We then obtain

$$\frac{\partial}{\partial s} \left[(T - mu^2) \frac{\partial y^i}{\partial s} \right] + F^i = 0.$$
 (1.4-2)

This equation implies that the string curve will be the same as the curve of a string which is in statical equilibrium subjected to the external force F^i , provided that m is the same and similar boundary conditions hold. If the string tension is $\overline{T} = \overline{T}(s)$ in the equilibrium case, the tension of the moving string is $T = \overline{T} + mu^2$. This result is well known, cf. [R2], p. 400.

2) If the external force is $F^i \equiv 0$, the system (1.4-1) is satisfied by $T = mu^2 = \text{const.}, m, u = \text{const.},$ for arbitrary functions $y^i(s)$. The string curve may thus have an arbitrary shape and may also contain points of strong discontinuities.

The latter fact is a consequence of results which will be presented in section 2.22. (A strong discontinuity propagates with the velocity $c = \left(\frac{T}{m}\right)^{1/2}$ relative to the string points, and from the preceding formulae we get c=u. The discontinuity is then fixed in space, on condition

that it moves backwards along the string; in the opposite case the string cannot possibly move in a fixed path.)

When the string is inextensible the motion now discussed is a special case of the motion discussed in (1).

The result obtained for a linearly elastic string may be used for all other kinds of elastic strings, since T=const. implies m=const. for every homogeneous elastic string.

2. D	iscussion of the String Equations
Sy	mbols and conventions:
y^i	Fixed, rectangular, right-handed, Cartesian coordinates. Coordinates of the string points.
8	Arc length of the string curve.
t	Time coordinate.
T	String tension.
m	Mass of the unit length of the string.
m_0	——————————————————————————————————————
u	Tangential velocity of the string point.
F^i	Rectangular, Cartesian components of the external force acting on the unit length of the string.
$x = [x_k]$	Column matrix with elements x_k , see Eq. (2.1-2).
$x' = (x_k)$	Row matrix, transpose of x .
A, B	Square matrices, see Eq. $(2.1-4)$, Eq. $(2.1-5)$.
A^{-1}	Inverse of A .
Ι	Unit (three-row) matrix.
$h = [h_k]$	Column matrix with elements h_k , see Eq. (2.1-3).
$\tau^{(i)}$	Row matrix, eigenvector with ordinal number i of the matrix pair A , B , see Eq. (2.1-8).
$\varrho^{(i)}$	Eigenvalue with ordinal number i of the matrix pair A , B .
P, Q	Variables defined by Eq. $(2.1-14)$.
E	Domain of the complex x, s, t space.
ξ, η	Coordinates in the s, t plane.
D	Jacobian of ξ , η with respect to s , t .
$\{A\}$	Weak discontinuity in the string property A , see Eq. (2.2-3).

- $l^{(i)}$ Path of propagation in the *s*, *t* plane of a weak discontinuity; *i* is the ordinal number.
- $w^{(i)}$ Propagation velocity in relation to the string points of a weak discontinuity propagating along the path $l^{(i)}$.

$$au^i = rac{\partial y^i}{\partial s}$$
 Unit vector along the string tangent.

U Elastic potential of the unit mass of the string.

- Propagation velocity of a strong discontinuity in relation to the string points.
- C The quantity of string mass passed by a strong discontinuity per unit of time.

Symbols not included in the above list will occur now and then in the text. Indices run from one to three, if not otherwise specified. The summation convention is used. Partial derivatives are generally

written by the use of indices, e. g. f_s means $\frac{\partial f}{\partial s}$.

2.1. Classification of the String Equations

2.11. Introduction

С

A definite problem concerning the motion of a string naturally cannot be regarded as properly formulated unless the string equations have a unique solution. This proposition immediately leads to an inquiry into existence and uniqueness theorems for initial and boundary value problems connected with the string equations. Particular interest will be centred on initial value problems of the infinite string and mixed initial — boundary value problems of the semi-infinite and finite string. General problems of these kinds are complicated, especially because of the presence of strong discontinuities,⁵ and we shall consider it to be outside the scope of this work to give a thorough discussion of such problems. In order to get some information concerning these questions, however, we shall classify the string equations in the usual manner according to the theory of partial differential systems. We shall also briefly discuss the propagation of discontinuities along the string.

⁵ Such cases are discussed by HAMEL, [H4], and PAILLOUX, [P1], [P3].
Anticipating the results, we assert that the string equations form a *quasi-linear*, differential system which in the case of a *linearly elastic* string (and all other kinds of elastic strings, too) is *hyperbolic* in the sense used by COURANT and others, *cf.* [C2], and [S1], p. 127. For the *inextensible string* the system is *not hyperbolic*. Further, the string curve is a *characteristic curve* for both kinds of string.⁶ The latter fact means that a problem with initial values prescribed along the string (which may be finite or infinite) generally will not be a proper problem.

Special kinds of mixed initial — boundary value problems have been investigated in connection with quasi-linear hyperbolic systems, cf. [B3], [C2], [C6]. Some of these problems may be of interest with reference to the elastic string. The results obtained will not be valid for the inextensible string.

To sum up, most dynamical problems related to a string, extensible or inextensible, finite or infinite, are not easily attacked by rigorous methods.

2.12. A First Order System of String Equations

We now turn to the study of the string in a rectangular, Cartesian coordinate system y^i , and use the arc length s of the string curve and the time t as independent variables. From the string equations (1.2-21) we easily obtain the following, first order differential system⁷

$$Ax_s + Bx_t + h = 0.$$
 (2.1-1)

In this equation we have adopted the index notation for the partial derivatives of the dependent variable $x, e. g. x_s = \frac{\partial x}{\partial s}$, &c. Further, x and h are column vectors. The coefficients A and B are square matrices. Writing x' and h' for the transposes of x and h we have (with y^i as in Chapter 1)

$$\left. \begin{array}{c} x^{i} \equiv x^{i}(u, t) \equiv (x_{1}, \dots, x_{9}, T, u), \text{ where} \\ x_{i} \equiv y^{i}, i \equiv 1, 2, 3, \\ x_{i+3} \equiv y^{i}_{s}, \\ x_{i+6} \equiv y^{i}_{t}; \end{array} \right\}$$

$$(2.1-2)$$

⁶ Anticipating later treatment we may mention here that strong discontinuities as well as weak ones propagate along characteristics, *cf.* section 2.22.

⁷ The symbols x, x_i in this chapter should not be confused with the curvilinear coordinates x^i appearing in Chapter 1.

$$\begin{array}{c} h' \equiv (h_1, \dots, h_{11}), \text{ where } \\ h_k = -x_{k+3}, k = 1, 2, 3, \\ h_k = F^{k-3}, k = 4, 5, 6, \\ h_k \equiv 0, k = 7, \dots, 11. \end{array}$$

$$(2.1-3)$$

The coefficient matrices A and B are expressed (in partitioned form) by

In these expressions I stands for the unit three-row matrix and every excluded element is identically zero, *i. e.* zero for every x, *s*, *t.* F^i , i=1,2,3, appearing in Eq. (2.1-3) are the rectangular, Cartesian components of the external force. In equation (2.1-1) we have preferred to take the last of the equations (1.2-12) as a kinematical condition instead of Eq. (1.2-11) which was used in Eq. (1.2-21). In order to make the system Eq. (2.1-1) equivalent to Eq. (1.2-21)the condition

$$x_i x_i = 1, i = 4,5,6$$
 (2.1-6)

must hold along some non-characteristic curve, cf. [K1], p. 11.

Since the sum Eq. (2.1-6) appears frequently in subsequent deductions, we shall, for the sake of brevity, imply $x_i x_i = 1$ throughout. In Eq. (2.1-1) m and T are connected by the elastic condition $_3$

$$m = m_0 \exp(-kT),$$

see section 1.25. This equation is also implied throughout.

The equation (2.1-1) forms a quasi-linear system of partial differential equations in two independent variables s, t and eleven dependent variables $x \equiv (x_1, \ldots, x_9, T, u)$.⁸ The coefficient matrices A and B are functions of the dependent variable x, but not of s and t. Further, A and B are analytic everywhere in the (complex, eleven-dimensional) x space. This is evident, since the elements of A and B are polynomials in m and the elements of the vector x and, by the elastic condition above, m is analytic in T for every T (which is an element of x). In the most general case the vector h=h(s, t, x), *i. e.* $F^{i}=F^{i}(s, t, x)$, but as we do not discuss the inhomogeneous string h is unlikely to depend on s explicitely. If F^i emanates from a scleronomic force field h does not depend on t explicitly, and in the common, gravitational field F^i depends on the string tension T only. If the external force $F^{i}(s, t, x)$ in the general case is analytic in its variables in a domain of the (thirteen-dimensional, complex) s, t, x space, then, by definition, the differential system Eq. (2.1-1) is analytic in that domain.

If the coefficient matrix A is non-singular, *i. e.* det $A \neq 0$, the inverse A^{-1} of A exists, and it is then possible to solve Eq. (2.1-1) with respect to x_s . det $A \neq 0$ holds in those points of the x space where

$$u \neq 0,$$

 $T - mu^2 \neq 0,$
 $1 - kmu^2 \neq 0,$
 $x_4, x_5, x_6 \text{ not all} = 0$
(2.1-7)

hold true simultaneously. The last condition is satisfied by the implication of Eq. (2.1-6). In several cases, however, the first two conditions are not satisfied everywhere on the string, cf. the discussion at the end of section 2.14.

2.13. Characteristics of the String Equations Hyperbolic Systems

Proceeding as in the works [S1] or [C2], we seek scalars ρ and vectors τ and p which satisfy the equation

$$\tau'(Ax_s + Bx_t + h) = p'(x_s + \varrho x_t) + \tau' h = 0$$
(2.1-8)

⁸ Concerning the term quasi-linear see for instance [S1], p. 36.

The vectors τ are called *eigenvectors* and the scalars ρ *eigenvalues* of the matrix pair A, B. This procedure means that we seek linear combinations of the equations in the system (2.1-1) which will contain the derivatives of each of the components of x in one and the same direction in the s, t plane. The latter direction is called a characteristic direction; it is determined by the ordinary differential equation

$$\frac{dt}{ds} = \varrho. \tag{2.1-9}$$

The integral curves of this differential equation are called *characteris*tics. The differential system Eq. (2.1-1) is a special one, in so far as A, B and h depend explicitly on x but not on the independent variables s and t. The same is consequently true of ρ , τ and p also. It is, however, of no particular importance as concerns the theoretical treatment of the system (2.1-1). From Eq. (2.1-8) we get

$$\tau'(B - \rho A) = 0,$$
 (2.1-10)

$$\det(B - \varrho A) = 0. \tag{2.1-11}$$

The latter equations determine ρ and τ , and if A and B are of the order n, with A nonsingular, there are n roots $\rho^{(k)}$ of the equation (2.1-11). Further, $\rho^{(k)}$, $k=1, \ldots, n$, is an eigenvalue of the matrix $A^{-1}B$, provided A is nonsingular. If A and B are both real, $\rho^{(k)}$ may be real or complex; distinct or multiple.

The differential system Eq. (2.1-1) is called hyperbolic in a point (x_0, s_0, t_0) of the n+2-dimensional x, s, t space, if every eigenvalue $\varrho^{(k)}$ is real and the eigenvectors $\tau^{(i)}$ span the *n*-dimensional space at that point.⁹ If a system is hyperbolic in every point of a domain E in the n+2-dimensional space it is called hyperbolic in E. The fact that a system is hyperbolic implies the existence of matrices K and K^{-1} such that $K^{-1}A^{-1}BK$ is a diagonal matrix. If A, B, K and the vector h are analytic in E, the system is called analytic hyperbolic in E, see [L1], p. 235.

It is well known that weak discontinuities in the functions $x_i(s, t)$ propagate along the characteristics, *cf.* section 2.21.

The equations (2.1-8) are frequently called the characteristic system (corresponding to Eq. (2.1-1)). Finally, we may point out

⁹ The eigenvectors span the *n*-dimensional space, if, and only if, the rank of $B \cdot \varrho A$ is $n \cdot q_k$ for a q_k -fold eigenvalue $\varrho^{(k)}$, and the same holds true for every eigenvalue.

that the significance of the characteristics and the characteristic system is not affected by transformations of the independent variables, cf. [C3], p. 143.

2.14. The Elastic String

In the case of a linearly elastic string we have the elastic constant k > 0. We easily obtain the following expressions:

where Eq. (2.1-6) is implied to hold and

$$P = T - mu^{2},$$

$$Q = 1 - mku^{2}.$$
(2.1-14)

We may remark that

1) If T > 0, *i.e.* if the string tension is positive, every eigenvalue $\varrho^{(i)}$, $i=1,\ldots,11$, is real, and by an investigation of the rank of $B-\varrho^{(i)}A$ we can conclude that the eigenvectors $\tau^{(i)}$ span the eleven-dimensional space.

2) We let the domain E of the s, t, x space consist of all points satisfying¹⁰

$$T > 0,$$

 $u, P, Q \neq 0,$ (2.1-15)
 $x_k x_k = 1, k = 4, 5, 6.$

¹⁰ The last condition in Eq. (2.1-15) may be replaced by a less restrictive one, for instance, x_4, x_5, x_6 not all zero, *cf.* remarks in connection with Eq. (2.1-6).

The differential system Eq. (2.1-1), or Eq. (2.1-8), is then analytic hyperbolic in E. It must be pointed out that while T < 0 is excluded by hypothesis, *vide* section 1.11, T, u, P=0 may occur in string points that are interesting from the physical aspect. At a free end of a string T=0, in a string point fixed in space u=0, and motions where P=0are discussed in section 1.4. All non-hyperbolic points of the s, t, xspace are therefore not without interest.

3) The curves
$$x_i = x_i(s, t), \frac{dt}{ds} = \varrho^{(k)}, i = 1, 2, 3; k = 1, ..., 11, i. e.$$
 the

curves on the surface S (generated by the string in its motion through space) corresponding to the characteristics, have simple physical interpretations. The eigenvalue $\varrho=0$ corresponds to the string curve, $\varrho^{(5)}$ to the paths of the string points and the others to the paths of propagation of certain discontinuities which will be discussed in section 2.2. The fact that the string curve itself corresponds to a characteristic complicates the treatment of an initial value problem (in the strict sense of the term), as was pointed out in the introduction to section 2.1.

2.15. The Inextensible String

The inextensible string is characterized by k=0, implying $m=m_0=$ =const. We then get the following expressions

$$\det A = uP^{2}, \qquad (2.1-16)$$

$$\varrho^{(1)}, \varrho^{(2)}, \varrho^{(3)}, \varrho^{(4)}, \varrho^{(6)}, \varrho^{(7)} \equiv 0,$$

$$\varrho^{(5)} = \frac{1}{u},$$

$$\varrho^{(8)}, \varrho^{(9)} = -\frac{1}{P} \left[m_{0}u + (m_{0}T)^{1/2} \right], \qquad (2.1-17)$$

$$\varrho^{(10)}, \varrho^{(11)} = -\frac{1}{P} \left[m_{0}u - (m_{0}T)^{1/2} \right].$$

By investigating the rank of B we find that only five linearly independent eigenvectors correspond to the sixfold eigenvalue $\varrho \equiv 0$. The eigenvectors therefore do not span the eleven-dimensional space, and the system Eq. (2.1-1), or Eq. (2.1-8), is consequently not hyperbolic in any point of the s, t, x space in the case of an inextensible string.

2.2. The Propagation of Discontinuities Along the String

It is well known from the theory of the inextensible string that discontinuities of the string properties may propagate along a string, see for instance [B1], p. 412, [H4] and [P3].

Since there does not appear to exist any general survey of the subject, the author considers it suitable to discuss the matter here.

We shall consider the string motion with rectangular Cartesians as space coordinates and the s, t coordinates of Chapter 1 as independent variables. We shall use the terms *weak* and *strong discontinuities*.

 Weak discontinuities are discontinuities in yⁱ_{ss}, yⁱ_{st}, yⁱ_{tt}, (i=1,2,3), T_s, T_t, m_s, m_t, u_s, u_t, or discontinuities of derivatives of higher order.
 Strong discontinuities are discontinuities in yⁱ_s, vⁱ, (i=1,2,3),

T, m, u.

It should be pointed out that weak discontinuities may be analyzed by using the string equations in their characteristical form as a starting point, while the presence of strong discontinuities necessitates a particular treatment.

2.21. Weak Discontinuities

In essential the discussion in this section follows that in [S1], p. 135 et seq. We start from the string equations (1.2-21) and assume the external force F^i to be continuous in s, t and y^i . This assumption means that we disregard discontinuities caused by an external force.¹¹ If the latter is a field force, for instance the weight of the string, our assumption is justified; but it is not valid in several cases of physical interest.

We now introduce a new set of coordinates ξ , η in the s, t plane by the one to one transformation

$$\left. \begin{array}{c} \xi = \xi(s, t), \\ \eta = \eta(s, t), \end{array} \right\}$$

$$(2.2-1)$$

$$D = \frac{\partial(\xi, \eta)}{\partial(s, t)} \neq 0.$$
 (2.2–2)

38

¹¹ To make the investigation valid for discontinuities in derivatives of higher order than those appearing in Eq. (1.2-21) we must, of course, make further assumptions regarding the differentiability of F^{i} .

Assuming ξ and η to be sufficiently smooth we can use them as independent variables in the string equations (1.2–21). The curve $\xi(s, t)=0$ divides the s, t plane into two domains, (1) and (2). An arbitrary property $A=A(s, t)=A(s(\xi, \eta), t(\xi, \eta))$ (scalar or vector) of the string is said to possess a discontinuity $\{A\}$ along $\xi(s, t)=0$ if

$$\{A\} = \lim_{P_1, P_2 \Rightarrow P} [A(P_1) - A(P_2)] \neq 0, \qquad (2.2 - 3)$$

where A(Q) is the value of A(s, t) at a point Q; P is a point on $\xi=0$ and P_1 , P_2 are points in the domains (1) and (2) approaching P. The property A then shows a jump $\{A\}$ when crossing the curve $\xi=0$. Let the following conditions hold true:

1) The string properties satisfy Eq. (1.2-21) in the open domains (1) and (2).

2) y_s^i, y_t^i, T, m, u are continuous when crossing the curve $\xi = 0$.

3) The intrinsic derivatives¹² $(y_s^i)_{\eta}$, $(y_t^i)_{\eta}$, T_{η} , m_{η} , u_{η} are continuous on $\xi=0$.

4) The external derivatives $(y_s^i)_{\xi}$, $(y_t^i)_{\xi}$, T_{ξ} , m_{ξ} , u_{ξ} may be discontinuous on $\xi=0$.

For the intrinsic derivative A_{η} of the string property A we obtain (by equations (2.2-1), (2.2-2))

$$A_{\eta} = A_s \, s_{\eta} + A_t \, t_{\eta} = - \frac{1}{D} \left(A_s \, \xi_t - A_t \, \xi_s \right).$$

Taking A to be any one of the properties listed in condition (2), we obtain by condition (3)

$$\{A_s\} + \varrho\{A_t\} = 0,$$
 (2.2-4)

where

$$\varrho = \frac{dt}{ds} = -\frac{\xi_s}{\xi_t} \tag{2.2-5}$$

is the derivative of t with respect to s along the curve $\xi(s, t)=0$. From the string equations (1.2-21) and the kinematical conditions Eq. (1.2-12) we obtain the following system for the weak discontinuities of the string properties,

¹² *i. e.* the derivatives along the curve $\xi = 0$.

$$\begin{aligned} &\{T_{s}-muu_{s}-mu_{t}\} y_{s}^{i}+(T-mu^{2}) \{y_{ss}^{i}\}-m\{y_{tt}^{i}\}-2mu\{y_{st}^{i}\}=0, \\ &y_{s}^{i}\{y_{ss}^{i}\}=y_{s}^{i}\{y_{st}^{i}\}=0, \\ &\{u_{s}\}=k\{T_{t}+uT_{s}\}, \\ &\{m_{s}\}=-km\{T_{s}\}, \{m_{s}\}=-km\{T_{s}\}, \end{aligned}$$

$$(2.2-6)$$

where we have summation on i=1,2,3.¹³ The Eqs. (2.2-6) together with Eq. (2.2-4) and Eq. (2.2-5) give the essential information about the weak discontinuities and their paths of propagation. There are some differences between the elastic string, k>0, and the inextensible string, k=0. It is easy to verify that the possible paths of propagation of discontinuities coincide with the characteristics of the string equations discussed in sections 2.13, 2.14 and 2.15. No discontinuity can propagate, however, along the characteristics determined by $\varrho = \frac{1}{u}$.¹⁴

We shall now discuss the elastic and the inextensible string separately.

a) For the *elastic string* we have k > 0. Let the path of propagation (in the s, t plane), corresponding to the directional derivative $\frac{dt}{ds} = \varrho^{(p)}$, be symbolized by $l^{(p)}$, $p=1, \ldots, 5$.

We conclude from Eq. (2.2-6) that along $l^{(1)}$

$$\begin{array}{l} \varrho^{(1)} = 0, \\ \{y^i_t\} = -\{u_t\} y^i_s \neq 0, \ i = 1, 2, 3, \end{array}$$
(2.2-7)

while all other properties remain continuous on $l^{(1)}$. For the vector y_{tt}^i a discontinuity parallel to the tangent vector y_s^i (of the string) consequently propagates instantaneously along the string. It should be observed that the acceleration vector remains continuous, since the discontinuities $\{y_{tt}^i\}$ and $\{u_t\}$ cancel each other.

¹³ The application of this method to problems containing discontinuities of vectors in common space requires no particular investigation.

¹⁴ It may be pointed out that this discrepancy is due to the fact that on discussing the characteristics we only make use of the last of the three conditions in Eq. (1.2-12), while we here apply the first two.

We next obtain, from Eq. (2.2-6), the following relations, valid along $l^{(2)}$ and $l^{(3)}$.¹⁵

$$\begin{split} \varrho^{(2)}, \, \varrho^{(3)} &= -\frac{1}{P} \left[m u \pm (mT)^{1/2} \right], \\ \{y^i_{ss}\} &= -\varrho^{(p)} \, \{y^i_{st}\} = (\varrho^{(p)})^2 \{y^i_{tt}\} \neq 0, \\ p &= 2, \, 3; \, i = 1, 2, 3, \end{split}$$

$$(2.2-8)$$

where $P=T-mu^2$. The other properties of the string remain continuous along $l^{(2)}$ and $l^{(3)}$. The physical interpretation of the above discontinuities is a jump in the acceleration vector perpendicular to the tangent vector y_s^i of the string curve.

Along $l^{(4)}$ and $l^{(5)}$, finally,

$$\begin{split} \varrho^{(4)}, \, \varrho^{(5)} &= -\frac{1}{Q} \left[mku \pm (mk)^{1/2} \right], \\ \{u_s\} &= \pm \{T_s\} \left(\frac{k}{m} \right)^{1/2}, \\ \{T_t\} &= (\varrho^{(p)})^{-1} \{T_s\}, \ p = 4, 5, \\ \{m_s\} &= -km \{T_s\}, \ \{m_t\} = -km \{T_t\}, \end{split}$$

$$(2.2-9)$$

while the second order derivatives of y^i remain continuous. Physically, these discontinuities are interpreted by a jump in the acceleration vector parallel to y_s^i , a jump in the derivatives of the string tension T and corresponding jumps in the derivatives of the mass m of the unit length of the string.

Concerning the determination of the path $l^{(i)}$, $i=1, \ldots, 5$, we may remark, that owing to the nonlinear character of the string equations the differential equations (2.2-5) of the $l^{(i)}$'s contain the dependent variables of the string equations. The paths $l^{(i)}$ consequently cannot be determined until the string equations are solved. For the propagation velocities $w^{(p)}$ in relation to the string points we obtain

$$w^{(p)} = (\varrho^{(p)})^{-1} - u, \qquad (2.2 - 10)$$

since we know from section 1.22 that u (in connection with the s, t coordinates) means the velocity of the string point along the string. Therefore

¹⁵ The notation here is different from that used in sections 2.14 and 2.15.

$$w^{(2)}, w^{(3)} = \mp \left(\frac{T}{m}\right)^{1/2},$$
 (2.2-11)
 $w^{(4)}, w^{(5)} = \mp (mk)^{-1/2}.$

As mentioned previously the first expression means that the corresponding discontinuity propagates instantaneously. The expressions for $w^{(2)}$ and $w^{(4)}$, $(w^{(3)}$ and $w^{(5)})$ imply that the corresponding discontinuities, see Eq. (2.2–8), propagate with the same relative velocity in either direction. Comparison with the results relating to small strain in the theory of elasticity indicates that $w^{(2)}$, $w^{(3)}$ correspond to the propagation velocity of transversal waves along a tight string, while $w^{(4)}$, $w^{(5)}$ correspond to the propagation velocity of longitudinal waves along a thin rod, cf. [R1], pag. 271 and 245.¹⁶ These facts become clear, if we let the string possess the cross-section area A, YOUNG's modulus of the material E, the stress σ , and the density q. We then get

21,(1)____

This agreement is hardly surprising, remembering the physical interpretation of the discontinuities.

Finally, we may remark that the equations (2.2-6) only give information about the relative magnitudes of the discontinuities. It is well known that it is possible to deduce conditions determining the jumps $\{A\}$ of the string properties A from the string equations in their characteristic form, Eq. (2.1-8). These conditions appear as ordinary differential equations for the $\{A\}$'s along the characteristics (*i. e.* the paths of propagation of the discontinuities). In the absence of solutions to the string equations the above equations will yield

¹⁶ It may be remarked that our results are valid for finite strain, provided the material of the string is linearly elastic in the sense stated in section 1.25.

only general, well-known results, and we therefore confine ourselves to pointing out their most important consequences. The latter may be stated as follows. There are only two possibilities for the jump $\{A\}$, viz.

$$\{A\}{\equiv}0$$
 on the characteristic, or
$$\{A\}{\neq}0. \tag{2.2-13}$$

This means that a weak discontinuity cannot (in the absence of discontinuous external forces) be formed in or vanish from an interior point of a string. It must be forced on the end points or be present in the initial state.

b) For the *inextensible string*, k=0, we obtain from Eq. (2.2-6)

$$\begin{split} \varrho^{(1)} &= 0, \\ \{y^i_{tt}\} &= -\{u_t\} y^i_s \neq 0, \ i = 1, 2, 3, \\ \{T_t\} &= 0. \end{split} \tag{2.2-14}$$

In addition to the discontinuities occurring on an elastic string, a jump in the time derivate of the string tension may then propagate instantaneously along the string. The expressions of Eq. (2.2-8) hold (with the obvious change from m to $m_0=$ const.) for the paths of propagation $l^{(2)}$ and $l^{(3)}$. The propagation velocities given in Eq. (2.2-11) and the comments on the elastic string also hold with appropriate modifications (the paths $l^{(4)}$ and $l^{(5)}$ may be regarded as contained in $l^{(1)}$).

2.22. Strong Discontinuities

We now consider the case when our linearly elastic string has a discontinuity in one or more of the properties y_s^i , v^i , T, m, u, i. e. the string tangent, the velocity of the string point, the tension, the mass of the unit length, and the velocity component along the string tangent. We shall assume that the discontinuities are not caused by external point forces. It is well known that only one kind of strong discontinuity can occur on the inextensible string, see for instance [B1], [P1]. We shall find that this is the case also of the linearly elastic string, and probably of all other kinds of elastic strings, too.



Fig. IV. Schematic sketch of the propagation of a (strong) discontinuity in the tangent vector $\frac{\partial y^i}{\partial s}$. The path of the point of discontinuity through space is designated Γ .

We adopt the following notations (see Fig. IV):

- L'The string curve at the time t.L''-----t+dt. Γ The path of the discontinuity point in space.
- *s* The arc length of the string curve.
- $s_{\Gamma} = s_{\Gamma}(t)$ The value of s at the point of discontinuity (at the time t). Indices 1 and 2 denote string properties immediately to the left, $s = s_{\Gamma} 0$, of the point s_{Γ} and to the right, $s = s_{\Gamma} + 0$, respectively.
- $dl_1(dl_2)$ The length of an element of the string (passed by the discontinuity during the time dt) in the left (right) position; dM is the mass of the element.

 $\tau^i = y_s^i, i = 1,2,3$ The unit tangent vector of the string curve. U = U(T) The elastic potential of the unit mass of the string.¹⁷ $C = \frac{dM}{dt}$ The quantity of string mass passed by the discontinuity per unit of time.

 $c = \frac{dl}{dt}$ The propagation velocity of the discontinuity in relation to the string.

The figure is drawn with reference to a discontinuity propagating in the direction of positive s.

¹⁷ The existence U(T) is an immediate consequence of our assumption that the length of an element depends explicitly on the tension only; c_{f} , section 1.25.

Neglecting low order terms, we obtain

$$\begin{split} & m_1 \exp{(kT_1)} = m_2 \exp{(kT_2)} = m_0, \\ & m_1 dl_1 = m_2 dl_2 = dM, \\ & \tau_1^i dl_1 + v_1^i dt = \tau_2^i dl_2 + v_2^i dt, \\ & (v_1^i - v_2^i) dM = (T_2 \tau_2^i - T_1 \tau_1^i) dt, \\ & (U_1 - U_2 + \frac{1}{2} v_1^i v_1^i - \frac{1}{2} v_2^i v_2^i) dM = (T_2 \tau_2^i v_2^i - T_1 \tau_1^i v_1^i) dt, \end{split}$$

where we have summation on i=1,2,3. In this system the first equation is the elastic condition, the second implies conservation of mass, the third is a kinematical condition obtained from Figure IV, the fourth implies conservation of momentum, and the fifth is an energy equation. It may be remarked that a discontinuous external force will not affect the discontinuity, provided it is not a point force. After some further calculations we get from Eq. (2.2-15)

$$\begin{split} & m_1 c_1 {=} m_2 c_2 {=} C, \\ & [T_1 {-} m_1 (c_1)^2] \, \tau_1^i {=} [T_2 {-} m_2 (c_2)^2] \, \tau_2^i, \\ & 2 \, (C)^2 \, (U_2 {-} U_1) {=} (T_2)^2 {-} (T_1)^2. \end{split} \tag{2.2-16}$$

As the string is linearly elastic we get the following expression for U

$$U = \frac{k}{m_0} \int T \exp(kT) dT$$

= $U_0 + (m_0 k)^{-1} \exp(kT) (kT - 1),$ (2.2-17)

where U_0 is a constant.

We conclude from Eq. (2.2-16) and Eq. (2.2-17) that only one kind of discontinuity can exist. We get

$$\begin{aligned} \tau_{1}^{i} &= \tau_{2}^{i} \implies T_{1} = T_{2} = T, \\ m_{1} = m_{2} = m, \\ c_{1} = c_{2} = c = \pm \left(\frac{T}{m}\right)^{1/2}. \end{aligned} \tag{2.2-18}$$

Only the tangent vector is consequently discontinuous.

This kind of discontinuity is possible, of course, also in the case of an inextensible string. On comparing the above expression with Eq. (2.2-11) we can observe that the strong discontinuity moves along the string at the same relative velocity as a weak discontinuity of that component of acceleration which is perpendicular to the string tangent. The strong discontinuity therefore propagates along a characteristic of the string equations, as HAMEL, [H4], points out (in a case of plane motion).

46

3.	The Str	ring Equations in Cylindrical Coordinates
	Symbols	and conventions:
x^1		Cylindrical coordinate, radial distance.
x^2		, polar angle.
x^3		, axial distance.
8		Arc length of the string curve.
z		Independent variable, defined by Eq. $(3.1-6)$.
t		The time.
T		String tension.
\overline{T}		Reduced string tension, defined by Eq. $(3.1-10)$.
q, u		String properties, defined as in Chapter 1 and physically interpreted on pag. 50.
m		Mass of the unit length of the string.
K^i		Contravariant, cylindrical components of the external force acting on the unit length of the string.
a		Characteristical length.
ω		Angular velocity of the string at steady rotatory motion. $\frac{1}{\omega}$ is used as a characteristical time.
T_{0}		Characteristical force.
τ		Independent variable, dimensionless time, defined by Eq. $(3.1-6)$.
ξ		Dependent variable, dimensionsless radial distance, defined by Eq. $(3.1-8)$.
$\varphi =$	x^2	Dependent variable, see above.
σ		Dependent variable, dimensionless string tension, defined by Eq. $(3.1-8)$.
ō		Dependent variable, dimensionless reduced string tension, defined by Eq. $(3.1-10)$.

γ Dependent dimensionless variable, defined by Eq. $(3.1-8)$ and physically interpreted on pag. 50.
 Dependent variable, dimensionless string mass per uni length.
μ Dependent variable, dimensionless axial velocity of th string point, see Eq. (3.1-8).
p_1, α Dimensionless string parameters, defined by Eq. (3.1-9).
$y_i, i=1, \ldots, 6$ Dimensionless, dependent variables for a small, time dependent motion superimposed on a steady rotatory motion of a string, see Eq. (3.2-2).
h Dimensionless constant, see Eq. $(3.2-5)$.

In this chapter derivatives are generally written by means of indices, e.g. $y_z = \frac{\partial y}{\partial z}$.

3.1. Dimensionless Variables

In this section we shall establish a complete system of equations for the linearly elastic string which hold in a cylindrical coordinate system fixed in space. In subsequent applications it will be convenient to use dimensionless variables, and we shall introduce such variables from the first. As we shall not deal with plane motion, the time and the axial distance of the point (on the string curve) will serve as independent variables in the string equations. Let us start from the system (1.2-24) and let x^1 be the distance from the axis, x^2 the polar angle, and x^3 the axial coordinate of a point on the string curve. For the metric tensor we have

$$\begin{array}{l} g_{11}\!\equiv\!1, \ g_{22}\!=\!(x^1)^2, \ g_{33}\!\equiv\!1, \\ g_{ik}\!\equiv\!0, \ i\!=\!k. \end{array} \tag{3.1-1}$$

The CHRISTOFFEL symbols are determined by

$$\begin{cases} 2\\1 2 \end{cases} = \begin{cases} 2\\2 1 \end{cases} = \frac{1}{x^1}, \ \begin{cases} 1\\2 2 \end{cases} = -x^1;$$
(3.1-2)

all other Christoffel symbols $\equiv 0$.

After some calculations we get from system (1.2-24)

$$\begin{split} & \left(\frac{T}{q} x_{z}^{1}\right)_{z} - x^{1} \frac{T}{q} x_{z}^{2} - mq \left[\frac{d^{2}x^{1}}{dt^{2}} - x^{1} \left(\frac{dx^{2}}{dt}\right)^{2}\right] + qK^{1} = 0, \\ & \left(\frac{T}{q} x_{z}^{2}\right)_{z} + \frac{2}{x^{1}} \frac{T}{q} x_{z}^{1} x_{z}^{2} - mq \left[\frac{d^{2}x^{2}}{dt^{2}} + \frac{2}{x^{1}} \frac{dx^{1}}{dt} \frac{dx^{2}}{dt}\right] + qK^{2} = 0, \\ & \left(\frac{T}{q} x_{z}^{3}\right)_{z} - mq \frac{d^{2}x^{3}}{dt^{2}} + qK^{3} = 0, \\ & \left(x_{z}^{1}\right)^{2} + (x^{1}x_{z}^{2})^{2} + (x_{z}^{3})^{2} = (q)^{2}, \\ & \left(mq\right)_{t} + (mqu)_{z} = 0, \\ & m = m_{0} \exp (-kT), \end{split}$$

$$(3.1-3)$$

where

4

$$\begin{aligned} x_{z}^{1} &= \frac{\partial x^{1}}{\partial z}, \quad x_{t}^{1} &= \frac{\partial x^{1}}{\partial t}, \\ \frac{dx^{1}}{dt} &= x_{t}^{1} + ux_{z}^{1}, \text{ etc.} \end{aligned}$$
(3.1-4)

 K^i , i=1,2,3, are the contravariant, cylindrical components of the external force (exerted on the unit length of the string), determined from the Cartesian components by Eq. (1.2-25). In order to facilitate the transition to dimensionless dependent and independent variables we now introduce characteristical quantities of length, time, and force:

$$a = characteristical length,$$

 $\frac{1}{\omega} = characteristical time,$ (3.1-5)
 $T_0 = characteristical force.$

On studying particular problems we shall later identify these parameters with well-defined physical quantities. Let now z and τ be dimensionless independent variables defined by

$$x^{3} = az,$$

$$t = \frac{\tau}{\omega}.$$
(3.1-6)

By this definition we have

$$\begin{array}{ll} x_{z}^{3} \equiv a, & x_{zz}^{3} \equiv x_{t}^{3} \equiv 0, \\ \\ \frac{\partial}{\partial t} = \omega & \frac{\partial}{\partial \tau} \,. \end{array} \tag{3.1-7}$$

From the discussion at the end of section 1.22 we can conclude (putting $y^3 = x^3 = az$) that $\frac{a}{q}$ is the direction cosine of the tangent vector of the string with respect to the x^3 axis. Further, au is the component of velocity in the direction of the x^3 axis. The string variables $q(z, \tau)$ and $u(z, \tau)$ are then given physical interpretations. The transformation Eq. (1.2-2), by which the independent variable z was introduced, has a sense only if $\frac{1}{q} \neq 0$, *i.e.* if the string curve does not touch any plane $x^3 = \text{const.}$ Assuming this condition to hold we choose $q(z, \tau) \ge a > 0$ (it should be noted that the equations (3.1-3) are invariant under a change from q to -q). The velocity component aumay, of course, be positive, zero or negative.

In order to complete the transition to dimensionless variables we introduce the variables ξ , φ , σ , γ , μ , ϱ , functions of z, τ , by means of the following expressions:¹⁸

$$x^{1} = a\xi, \ x^{2} = \varphi, \ \frac{T}{q} = \frac{T_{0}}{a}\sigma,$$

$$q = a\gamma, \ u = \omega\mu, \ m = m_{0}\varrho.$$
(3.1-8)

The variables defined in this manner have simple physical interpretations. ξ is the dimensionless form of the distance of the point (ξ, φ, z) (on the string curve) from the z axis, φ is the polar angle measured from a line fixed in space. Concerning σ we observe that $\frac{a}{q}T$ is the axial component of the string tension vector and that σ therefore is the dimensionless form of this force component. $\frac{1}{\gamma}$ is the axial component of the unit tangent vector of the string curve, hence $\gamma \ge 1$. μ is the dimensionless form of the unit length of the string, restricted by $1 \ge \rho > 0$.

50

¹⁸ The variables σ , τ which occur in this chapter should not be confused with the σ , τ in Chapter 1.

We shall find it convenient to use two dimensionless parameters p_1 and α , defined by

$$p_1 = \frac{m_0 a^2 \omega^2}{2T_0}, \qquad (3.1-9)$$

 $\alpha = kT_0.$

The constants p_1 and α , $p_1 > 0$, $\alpha \ge 0$, will be called *string parameters*. It can easily be verified that α is the logarithmic strain of a string element subjected to the tension T_0 . The inextensible string is characterized by $\alpha = 0$.

New variables may also suitably be introduced instead of T and σ . Let \overline{T} and $\overline{\sigma}$ be defined by

$$\overline{T} = T - mu^2 q^2,$$

$$\overline{\sigma} = \overline{T} \frac{a}{qT_0} = \sigma - 2p_1 \varrho \gamma \mu^2.$$
(3.1-10)

 \overline{T} will be called the *reduced string tension* and $\overline{\sigma}$ the *dimensionless* reduced string tension, although the latter is not the dimensionless form of \overline{T} .

After straight-forward calculations the system (3.1-3) takes the form¹⁹

$$\begin{split} (\bar{a}\xi_z)_z &-\bar{a}\xi\varphi_z^2 - 2p_1[(\varrho\gamma\mu\xi_z)_\tau + \varrho\gamma(\xi_{\tau\tau} + \mu\xi_{z\tau}) - \varrho\gamma\xi(\varphi_\tau^2 + 2\mu\varphi_z\varphi_\tau)] + \\ &+ \gamma \frac{a}{T_0}K^1 = 0, \\ (\bar{a}\xi^2\varphi_z)_z &- 2p_1[(\varrho\gamma\mu\xi^2\varphi_z)_\tau + \varrho\gamma(\xi^2\varphi_\tau)_\tau + \varrho\gamma\mu(\xi^2\varphi_\tau)_z] + \\ &a^2 \end{split}$$

$$+\gamma\xi^2 \frac{a^2}{T_0} K^2 = 0, \qquad (3.1 - 11)$$

$$\begin{split} \bar{\sigma}_z - 2p_1 \left(\varrho \gamma \mu \right)_{\tau} + \gamma \frac{a}{T_0} K^3 = 0, \\ \bar{\xi}_z^2 + (\bar{\xi}\varphi_z)^2 + 1 = \gamma^2, \\ (\varrho \gamma)_{\tau} + (\varrho \gamma \mu)_z = 0, \\ \rho = \exp\left[-\alpha \gamma (\bar{\sigma} + 2p, \rho \gamma u^2) \right]. \end{split}$$

¹⁹ The superscripts in K^1, K^2, K^3 indicate the components of the force vector as in Eq. (3.1-3), but everywhere else a superscript means raising to a power. This inconsistency should not cause any confusion.

Combining the first three equations in this system we get

$$\begin{split} (\bar{\sigma}\gamma)_z &- 2p_1[(\varrho\gamma^2\mu)_\tau + \varrho\gamma\mu\gamma_\tau + \varrho\xi_z(\xi_{\tau\tau} - \xi\varphi_\tau^2) + \varrho\varphi_z(\xi^2\varphi_\tau)_\tau] + \\ &+ \frac{a}{T_0}(\xi_z K^1 + a\xi^2\varphi_z K^2 + K^3) = 0. \end{split} (3.1-12)$$

This equation corresponds to the second one in Eq. (1.2-22); it is the equation of motion (for the string element) along the string tangent.

The system (3.1-11) consists of five nonlinear partial differential equations and one transcendental equation, the elastic condition. There are two independent variables, z and τ , and six unknown functions, ξ , φ , $\tilde{\sigma}$, γ , μ , ϱ . We can add the equation (3.1-12) to the system without changing its main properties.

3.2. String Equations for a Small Motion Dependent on Time and Superimposed on a Steady Rotatory Motion; External Forces Being Absent

In this section we shall specialize the string equations (3.1-11) and (3.1-12) to apply to a small time-dependent motion superimposed on a steady rotatory motion. These motions will be described more exactly below. It is assumed that no external forces affect the motions. Linearizing the string equations in the usual manner, they will be separated into 'steady' terms and 'time-dependent' terms. In Chapter 6 we shall make an investigation into a simple special case of such a motion.

3.21. Description of the Motion. Linearization

The following conditions are postulated to hold for the motion:

1) No external forces act on the string. Therefore

$$K^1 = K^2 = K^3 = 0 \tag{3.2-1}$$

hold everywhere in the strip $z \in (z_1, z_2), \tau \ge 0$ which we shall take to be the domain of definition of the motion. A point force equal to the positive (negative) of the string tension vector acts on the string at the boundary $z=z_2$, $(z=z_1)$. 2) Within the strip the dependent variables can be expanded in power series of a small, real parameter ε according to the following expressions²⁰

$$\begin{split} &\xi(z,\tau) = \xi_0(z) + \varepsilon y_1(z,\tau) + O(\varepsilon^2), \quad \varepsilon \to 0, \\ &\varphi(z,\tau) = \tau + \varphi_0(z) + \varepsilon y_2(z,\tau) + O(\varepsilon^2), \\ &\bar{o}(z,\tau) = \bar{o}_0(z) + \varepsilon y_3(z,\tau) + O(\varepsilon^2), \\ &\gamma(z,\tau) = \gamma_0(z) + \varepsilon y_4(z,\tau) + O(\varepsilon^2), \\ &\mu(z,\tau) = \mu_0(z) + \varepsilon y_5(z,\tau) + O(\varepsilon^2), \\ &\varrho(z,\tau) = \varrho_0(z) + \varepsilon y_6(z,\tau) + O(\varepsilon^2). \end{split}$$
(3.2-2)

Condition (1) implies that we neglect the effect of gravity on the string and the air-drag.²¹ Terms not containing ε in Eq. (3.2–2) will be called steady terms and the others time-dependent terms. Thus $\varepsilon=0$ corresponds to a *steady motion*. The time-dependent terms constitute the so-called *time-dependent motion*. The functions $y_i(z, \tau)$, $i=1, \ldots, 6$, approximate the time-dependent motion. The expressions (3.2–2) imply that, in the steady motion every string point, viewed from a coordinate system rotating about the z axis with the constant angular velocity ω , will describe the same path $\xi = \xi_0(z), \varphi = \varphi_0(z)$ with a velocity uq which we shall prefer to call the tangential velocity. It is easily concluded from Eq. (3.1–7) that the angular velocity is ω . We have consequently now given a *physical interpretation of the char*-

acteristical time $\frac{1}{\omega}$.

Adding the equation (3.1-12) to the system (3.1-11) and substituting the expressions (3.2-2) for the unknown functions, we get a system each equation of which may be arranged according to powers of ε .

We now assume this system to be satisfied in such a way that the coefficient of each power of ε is zero in each of the equations.

We then get

a) A non-linear system of ordinary differential equations (including a transcendental equation, the elastic condition) for the steady motion.

²⁰ Naturally, we must assume that the expansion coefficients (functions of z and τ) are sufficiently regular to guarantee the validity of the subsequent deductions.

²¹ A special case of steady rotatory motion in a gravitational field has been treated by KOLODNER, [K2], and steady rotatory motion influenced by air-drag by MACK, [M1]; in those papers the string is assumed to be inextensible.

b) A homogeneous, linear system of partial differential equations for the unknown functions $y_i(z, \tau)$, $i=1, \ldots, 6$, together with an algebraic equation linear in y_i (and homogeneous). The coefficients of this system are functions of z formed by the steady terms.

c) A rest-system consisting of homogeneous, linear partial differential equations for higher order terms.

It is of great interest to study a time-dependent motion caused by a small external disturbance acting on a string which initially has a steady rotatory motion. The functions $y_i(z, \tau)$, solutions of the system (b), form a first order approximation of the time-dependent motion. In such a case the parameter ε is an appropriate measure of the disturbance. The system (b) will be non-homogeneous, if the disturbance has the form of an external, distributed force acting on the string in the interval in question. In order to be able to discuss the time-dependent system (b) one must first, of course, determine the essentials of a steady motion which is governed by (a).

The procedure outlined for the time-dependent motion may be called a *linearization*, and (in general) the following two problems will then result:

a') A two-point boundary value problem for the system (a).

b') A mixed initial-boundary value problem for the system (b).

Questions of convergence of the method (viz. the expansion of the time-dependent motion in powers of ε) will also arise.

We shall thoroughly discuss the equations of steady motion in Chapter 4. A strange type of steady motion is briefly discussed in Chapter 5, and in Chapter 6 we shall study a simple vibration problem for an inextensible string.

3.22. The Equations of Steady Motion

From Eqs. (3.1-11) and (3.2-2) we obtain the following system valid for the steady terms

$$\begin{split} &(\bar{\sigma}\xi_{z})_{z} - \bar{\sigma}\xi\varphi_{z}^{2} + 2p_{1}\varrho\gamma\xi + 4p_{1}\varrho\gamma\mu\xi\varphi_{z} = 0, \\ &(\bar{\sigma}\xi^{2}\varphi_{z})_{z} - 2p_{1}\varrho\gamma\mu(\xi^{2})_{z} = 0, \\ &\bar{\sigma}_{z} = 0, \\ &(\bar{\sigma}\gamma)_{z} + 2p_{1}\varrho\xi\xi_{z} = 0, \\ &(\bar{\sigma}\gamma)_{z} + 2p_{1}\varrho\xi\xi_{z} = 0, \\ &\xi_{z}^{2} + (\xi\varphi_{z})^{2} + 1 = \gamma^{2}, \\ &(\varrho\gamma\mu)_{z} = 0, \\ &\varrho = \exp\left[-\alpha\gamma(\bar{\sigma} + 2p_{1}\varrho\gamma\mu^{2})\right], \end{split}$$
(3.2-3)

where index zero has been omitted for the sake of brevity. The fourth equation of the system corresponds to Eq. (3.1-12) and the first four equations are consequently not independent. From the equation $(\varrho\gamma\mu)_z=0$ we get $\varrho\gamma\mu=$ const. along the string. This fact has a simple physical interpretation as is shown later, see remark (1). Since $\varrho(z)$, $\gamma(z) < \infty$ according to assumptions made previously, we have two possibilities for $\mu(z)$, viz.

$$\mu(z) \neq 0$$
, or $\mu(z) \equiv 0$.

The tangential velocity of a string point is $uq = a\omega\mu\gamma$; in a steady rotatory motion this velocity consequently cannot be zero in one point unless it is zero everywhere, a result which is almost self-evident. For $\mu(z) \neq 0$ the system Eq. (3.2-3) gives

$$\begin{split} \bar{\sigma}(\xi_{zz} - \xi \varphi_z^2) + 2p_1 h \frac{1}{\mu} \xi + 4p_1 h \xi \varphi_z = 0, \\ \bar{\sigma}(\xi^2 \varphi_z)_z - 4p_1 h \xi \xi_z = 0, \\ \bar{\sigma} \gamma_z + 2p_1 \varrho \xi \xi_z = 0, \\ \xi_z^2 + (\xi \varphi_z)^2 + 1 = \gamma^2, \end{split}$$
(3.2-4)

where the following equations hold

$$\varrho = \exp\left[-\alpha\gamma \left(\bar{\sigma} + 2p_1 h \mu\right)\right], \\ \rho\gamma\mu = h = \text{const.}$$
(3.2-5)

and

$$\bar{\sigma} = \text{const.}$$
 (3.2–6)

For a motion where $\mu(z) \equiv 0$ the first equation of the system (3.2-4) becomes

$$\bar{\sigma}(\xi_{zz} - \xi \varphi_z^2) + 2p_1 \varrho \gamma \xi = 0,$$
(3.2-7)

while the others are valid if we put h=0. In the case of an inextensible string $\alpha=0\Rightarrow \varrho\equiv 1$ and the tangential velocity of the string point is therefore the same everywhere.

The system Eq. (3.2-4) is a nonlinear system of differential equations for the unknown functions $\xi(z)$, $\varphi(z)$, $\gamma(z)$, $\mu(z)$ and $\varrho(z)$; these functions are further connected by Eq. (3.2-5). α , p_1 are constants, cf. Eq. (3.1-9). The system is determinate since there are three independent differential equations and two other equations. These equations, or equivalent equations, may be derived directly from the conservation laws of classical mechanics, as is generally done for the case $\alpha=0$ (inextensible string) in works on textile mechanics, see for instance [M1].

In the following considerations we shall take it as understood that the equations (3.2-4) to (3.2-7) are defined on some interval (z_1, z_2) of the z axis. We may remark that

1) The same quantity of string mass flows per unit of time through every plane z=const. This is readily apparent from section 1.3, Eq. (1.3-6). In the latter equation we put $F(x^i, t)=x^3-\text{const.}=0$, and

consequently $\frac{\partial F}{\partial x^1} = \frac{\partial F}{\partial x^2} \equiv 0$ and by Eq. (3.1-7) $\frac{\partial x^3}{\partial t} \equiv 0$. Hence $\frac{\partial F}{\partial x^i} \frac{\partial x^i}{\partial t} \equiv 0 \Rightarrow H = mqu = m_0 a \omega h = \text{const.}$

2) From Eq. (3.1–10) and Eq. (3.2–6) we obtain $\overline{T}\frac{a}{q} = \overline{a}T_0 = \text{const.}$

Now, q > 0 which implies that \overline{T} cannot change sign along the string. Since \overline{T} is the difference between two functions which are positive definite, three different possibilities exist (we shall take no interest in the pathological case T < 0), viz.

a)
$$\overline{T} > 0$$
 $\Rightarrow \overline{\sigma} > 0$,
b) $-mu^2 q^2 \leq \overline{T} < 0$ $\Rightarrow -2p_1 \varrho \gamma \mu^2 \leq \overline{\sigma} < 0$
c) $\overline{T} = 0$ $\Rightarrow \overline{\sigma} = 0$

Case (c) does not possess any interest, since by Eq. (3.2-4) it implies either $\xi \equiv 0$, or $p_1 = 0 \Rightarrow \omega = 0$ (though a strange case of motion with $\xi \equiv \text{const.} \pm 0$, $\varphi_z = -1/2\mu = \text{const.}$ has, unfortunately, been passed over). The other two cases correspond to entirely different types of steady motion. Case (a) is exhaustively discussed in Chaper 4. Several special types of case (a) have been treated by various authors. Case (b), which obviously is possible only if $\mu \pm 0$ (*i. e.* non-zero tangential velocity), has apparently not been discussed before. We shall devote Chapter 5 to a brief investigation of this case.

3) In case (a) we find it suitable to choose $\bar{\sigma}=1$, implying $T_0=T a/q$. The denomination of T_0 as the z component of the reduced string

56

tension \overline{T} is justified, since $\frac{a}{q}$ is the direction cosine of the string tangent with respect to the z axis.²³ In this way the characteristical force T_0 is given a physical interpretation. In case (b) we analogously choose $\bar{\sigma} = -1$.

3.23. A Special Case of the Linearized Time-Dependent Motion

The mixed initial-boundary value problem (b') in section 3.21, which results from the linearization process, will in general turn out to be quite complicated. In the case of an elastic string the differential system is a linear hyperbolic system,²⁴ but the coefficient functions determined by the steady motion will become complicated. For the inextensible string the differential system is not hyperbolic, but on the other hand the coefficient functions are comparatively simple.

We now deduce the equations of a linearized time-dependent rotary motion for an inextensible string in the case of a steady motion characterized by the following condition

$$\varphi_0(z) = \mu_0(z) \equiv 0.$$
 (3.2-8)

This means that in steady motion the string is contained in a plane which rotates with constant angular velocity about the z axis. The string itself has no tangential velocity. The steady motion is therefore a simple and well-known motion, satisfying the following equations

$$\xi_{zz} + 2p_1\gamma\xi = 0,$$

 $\gamma_z + 2p_1\xi\xi_z = 0,$ (3.2-9)
 $\xi_z^2 + 1 = \gamma^2.$

Putting $\bar{\sigma}=1$ (cf. remark (3) in section 3.22) these equations are immediately obtained from Eqs. (3.2-4) and (3.2-7), and it can easily be seen that only two of them are independent. We next apply

²³ Actually, we may define the reduced string tension as the contravariant vector $\overline{T} \; \frac{\partial x^i}{\partial s}.$

²⁴ In a recent paper ŽDANOVIČ, [Z1], has investigated initial value problems for hyperbolic differential systems on the plane with homogeneous two-point boundary conditions (by means of separation of the variables).

the linearization process outlined in section 3.21 to the system (3.1–-11). For $\alpha=0$, $K^i\equiv 0$ (i=1,2,3), $\varphi_0(z)=\mu_0(z)\equiv 0$ we then get

$$y_{1zz} - 2p_{1}\gamma(y_{1\tau\tau} - y_{1} - y_{1z}\xi\xi_{z}\gamma^{-2} - 2y_{2\tau}\xi) + y_{3}\xi_{zz} = 0,$$

$$y_{2zz} - 2p_{1}\gamma(y_{2\tau\tau} + 2y_{1\tau}\xi^{-1} + 2y_{5}\xi^{-1}\xi_{z}) + 2y_{2z}\xi^{-1}\xi_{z} = 0,$$

$$y_{3z} - 2p_{1}\gamma y_{5\tau} = 0,$$

$$y_{1z\tau}\xi_{z} + \gamma(y_{5}\gamma)_{z} = 0,$$

$$y_{4} - y_{1z}\xi_{z}\gamma^{-1} = 0,$$

(3.2-10)

where we have omitted index zero for the steady terms just as in equations (3.2-9). This is a homogeneous system of linear partial differential equations for the unknown functions $y_i(z, \tau), i=1, \ldots, 5$. The coefficient functions satisfy Eq. (3.2-9) and they consequently depend on z but not on τ ; p_1 is a string parameter, cf. Eq. (3.1-9). The system is not hyperbolic, and this was not to be expected since we are dealing with an inextensible string, cf section 2.15.

The unknown function $y_4(z, \tau)$ appears only in the last equation. This equation may therefore be regarded as defining y_4 , and we can exclude it without changing the properties of the system. We shall return to this kind of motion in Chapter 6.

4. The Steady Rotatory Motion of the Linearly Elastic String in the Case of Positive Reduced String Tension and in the Absence of External Forces

Symbols and conventions:

Most of the symbols used in this chapter are the same as those in Chapter 3; the following list is therefore incomplete.

 p_2 Dimensionless string parameter, defined by Eq. (4.2-4).

- k Modulus of various elliptic functions and integrals. k also denotes an elastic constant.

$k_i, i=1,, 4$	Constants,	determined	by	Eq.	(4.2 - 13)	or	Eq.
direct in succession	(4.3 - 14).						
					Contraction of the second		

 m, n_1, n_2, n'_2 Constants, determined by Eqs. (4.3-13), (4.3-14) and (4.3-16).

 λ, δ Constants, determined by Eq. (4.3-19).

 $\varkappa_0, \varkappa_1, \varkappa_2$ Constants, determined by Eq. (4.3-20).

The symbols denoting elliptic functions and integrals agree with those used by BYRD and FRIEDMAN, [B2].

4.1. Introduction

Before we start the systematic treatment of the steady rotatory motion of the elastic string we may suitably give a survey of those steady-motion problems which will be treated later.



Fig. V. Sketch of the steady motion described in section 4.11. The broken curve A'B' is the trace of the string curve in the plane $x^3=z=0$. In the case reproduced here the string is assumed to flow from A to B.

4.11. A Two-Point Steady-Motion Problem

The following problem may be considered typical for a certain class of problems in textile mechanics. *The following premisses are given*:

1) A flexible, linearly elastic string with the mass m_0 per unit length in unstrained state, and the constant of elasticity k (k defined as in section 1.25).

2) A cylindrical coordinate system x^1 , x^2 , x^3 fixed in space, *cf*. Fig. V. We define the dimensionless coordinate z by $x^3 = az$, where a is the characteristical length in the problem.

3) Two moving points A and B. Their motions are defined by

$$x_A^1, z_A = \text{const.}, x_A^2 = \omega t,$$

 $x_B^1, z_B = \text{const.}, x_B^2 = \omega t + \delta$

where ω , $\delta = \text{const.}$, and $z_B > z_A$. The points A and B are consequently fixed in a coordinate system rotating with the constant angular velocity ω about the z axis.

4) A quantity of string mass H=const. which flows through A towards B (or in the opposite direction) per unit of time along a string curve containing A and B. It is assumed that the motion is steady in our sense of the term, cf. section 3.21; the string curve therefore does not change its shape.

5) No external forces act on the string in the interval (z_A, z_B) , *i. e.* the effects of gravity and air-drag are neglected.

6) One more condition in addition to those given in (3). This condition may be

a) Information concerning the total mass of the string elements in the interval (z_A, z_B) . This is a probable condition for motions without tangential velocity, *i. e.* for H=0.

b) Information concerning the magnitude of the string tension in some string point on the interval $[z_A, z_B]$.

c) A more general condition than (a) or (b), e. g. a connection between the total string mass in the interval (z_A, z_B) and the string tension at the point A. Such a condition and still more general ones may occur in problems of textile mechanics.

Necessary requirements include

7) All data for the string in the interval (z_A, z_B) .²⁵ We consequently require:

a) The equation of the string curve

$$\begin{aligned} x^{1} = x^{1} (z), \\ x^{2} = \omega t + \varphi (z) \end{aligned}$$

b) The tangential velocity or the z component of the velocity vector. c) The string tension T=T(z) which by the elastic condition determines m=m(z).

It may be remarked that

8) The string equations of the steady rotatory motion, Eq. (3.2-4), must hold on (z_A, z_B) .

9) It is not known from the beginning whether the reduced string tension \overline{T} is positive or negative, *cf.* section 3.22, remark (2). Apparently a solution of the problem always exists for $\overline{T} > 0$. Solutions for $\overline{T} < 0$ exist, at least in some special cases, *cf.* Chapter 5.

10) From a physical point of view it might be expected that the informations in points (1) to (6) should uniquely determine the solution

²⁵ In the case of positive reduced string tension (cf. section 3.22, remark (2)) the interval of definition of the occurring functions may always be extended to be the whole z axis.

of the problem, but this will not be true in general.²⁶ Apparently no existence or uniqueness theorems directly applicable to two-point problems of the present type are to be found in the mathematical literature. Questions concerning existence and uniqueness of two-point problems are left open in this thesis.

11) It is not easy to obtain explicit solutions of a two-point problem by direct attack. We shall therefore set one-point problems (initial value problems) with conditions for the string tension of the kind (6) (b). It will then be demonstrated that if a two-point problem does have a solution it is contained in the solution of a one-point problem. The uniqueness of the solutions of the one-point problems occurring here follows from elementary theorems on ordinary differential systems.

4.12. Two Steady-Motion Problems Relating to an Inextensible String

We now set two basic problems for an inextensible string. They are equivalent to the two-point problems discussed in the preceding section. The first one, called (I1), refers to a case, where the string curve does not intersect the z axis. In the second one, (I2), the string curve has at least one point of intersection with the z axis. Both problems, of course, start from the system Eq. (3.2-4).

(I1) Problem: To find functions $\xi(z)$, $\varphi(z)$, $\gamma(z)$, defined on some interval (a, b), a < 0 < b, such that they satisfy

a) The differential system

$$\begin{aligned} &(\xi^{2}\varphi_{z})_{z} - 4p_{1}h\xi\xi_{z} = 0, \\ &\gamma_{z} + 2p_{1}\xi\xi_{z} = 0, \\ &\xi_{z}^{2} + (\xi\varphi_{z})^{2} + 1 = \gamma^{2}. \end{aligned} \tag{4.1-1}$$

b) The initial values²⁷

$$\xi(0) = 1, \ \varphi(0) = 0, \ \varphi_z(0) = b, \ \gamma(0) = (1+b^2)^{1/2},$$
 (4.1-2)

 $^{^{26}}$ In an (unpublished) paper on the theory of ringspinning the author obtained an infinitely enumerable set of solutions to such a problem. Instable motions, indicating non-uniqueness, are well-known in textile technology, cf. also [N1].

 $^{^{27}}$ The initial value b is not to be confused with the right end point of the interval (a,b).

where the following inequality is valid

$$b^2 - 2p_1(1+b^2)^{1/2} - 4p_1hb \ge 0.$$
 (4.1-3)

(12) Problem: To find functions $\xi(z)$, $\varphi(z)$, $\gamma(z)$, defined on some interval (a, b), a < 0 < b, such that they satisfy

a) The differential system Eq. (4.1-1)

b) The initial values

$$\xi(0) = \varphi(0) = 0. \tag{4.1-4}$$

c) The condition

$$\xi_{z}=0 \text{ for } \xi=1.$$
 (4.1-5)

In problems (I1) and (I2) p_1 and h are real constants and $p_1>0$, while h is not restricted. Omitting the first of the equations and putting $\alpha=0$, $\bar{\sigma}=1$ in Eq. (3.2-4) we immediately obtain the system Eq. (4.1-1). Problems (I1) and (I2) will be completely solved in section 4.2. Anticipating the results in that section we may remark that

1) The existence and uniqueness of the solutions of (I1) and (I2) are guaranted by basic theorems in the theory of ordinary differential systems. The solutions will be obtained in terms of elliptic functions and integrals by straight-forward calculations. The latter property of the steady rotatory motion of an inextensible string has also been discovered by MACK, cf. [M1]. Mack used a method of representation entirely different from ours and does not make a penetrating study of the matter as the air-drag problem constitutes his major interest. We shall examine the subject thoroughly, especially since we need the results when studying the elastic string.

2) Problem (I1) is a one-point problem in the strict sense of the term, whereas (I2) is not.

3) The functions $\xi(z)$ in (I1) and (I2) turn out to be periodic functions, we may take their intervals of definition to be the whole z axis. The condition Eq. (4.1-2) means that $\xi(0)=1$ is an extremum of $\xi(z)$, and condition Eq. (4.1-3) that we choose it to be a minimum. These two conditions therefore do not imply any specialization, but merely a normalization of the function $\xi(z)$. In fact, $\xi(0)=1$ means that the characteristical length a, cf. (3.1-5), is the smallest distance between the string curve and the z axis. Similarly, the condition Eq. (4.1-5) for problem (I2) means that a is the greatest distance between the string curve and the z axis. 4) Problems (I1) and (I2) will prove to be equivalent to the two-point problems of section 4.11, *cf.* section 4.23.

4.13. Two Steady-Motion Problems for an Elastic String in the Case of Small Strain

In section 4.3 we shall treat those two basic problems for an elastic string which are analogous to problems (I1) and (I2). We shall assume that the strain $\alpha = kT_0$ is small and so obtain solutions valid in the vicinity of $\alpha = 0$ and correct up to and including terms $O(\alpha)$, $\alpha \rightarrow 0$. A common linearization process will lead to equations that are singular at the turning points $\xi_z = 0$, and in order to avoid such difficulties we shall adopt a special method. The solutions so obtained will be valid on any finite z interval. We shall call these problems (E1) and (E2), corresponding to (I1) and (I2), respectively.

4.14. Two Steady-Motion Problems for a Linearly Elastic String and Finite Strain

In section 4.4 we shall treat two basic problems related to the steady motion of a linearly elastic string, when the latter has no tangential velocity. These problems are analogous to problems (I1) and (I2) and will be called (E3) and (E4), respectively. No restrictions are imposed on the magnitude of the strain.

4.2. The Steady Rotatory Motion of the Inextensible String in the Case of Positive Reduced String Tension

Two problems, (I1) and (I2), related to the inextensible string were set in section 4.12. We now proceed to the solution of these problems, and at the end of the section we shall briefly discuss the possibilities of finding solutions of two-point problems, starting from the solutions of (I1) or (I2).

4.21. The Existence and Uniqueness of the Solution of Problem (I1)

The differential system Eq. (4.1-1) is easily transformed into the vector differential equation

$$x_z = A(x),$$
 (4.2–1)

where x and A(x) are column vectors in the five-dimensional space. Writing x' for the transpose of x we have

$$\begin{aligned} x' &\equiv (x_i) = (\xi, \, \xi_z, \, \varphi, \, \xi \varphi_z, \, \gamma), \\ i &= 1, \dots, 5. \end{aligned}$$

$$(4.2-2)$$

and

$$A(\mathbf{x}) = \begin{bmatrix} x_2 \\ x_4^2 x_1^{-1} - 2p_1 x_1 x_5 - 2p_2 x_4 \\ x_4 x_1^{-1} \\ 2p_2 x_2 - x_2 x_4 x_1^{-1} \\ -2p_1 x_1 x_2 \end{bmatrix}, \quad (4.2-3)$$

where p_2 is a constant

$$p_2 = 2p_1h.$$
 (4.2–4)

The initial values defined by Eq. (4.1-2) obviously correspond to

$$x'(0) \equiv k' = (1, 0, 0, b, (1+b^2)^{1/2}). \tag{4.2-5}$$

From the above equations we can conclude that the elements A_i of A are rational functions of the elements x_n of x; $i, n=1, \ldots, 5$. We write G for the domain $Re(x_1)>0$ of the complex five-dimensional space, and it is evident that A(x) is analytic and regular in G and also that $x(0)=k\in G$. From theorems on ordinary differential systems it therefore follows that problem (I1) has a unique solution, x(z), which is analytic in the neighbourhood of the point z=0, cf. for instance [H2], [P2]. This neighbourhood may be extended to consist of all points z such that x(z) is in the interior of G. Moreover, the solution is analytic in the parameters p_1 , p_2 and the initial value b.

4.22. The Existence and Uniqueness of the Solution of problem (12)

The transformation of the string equations from rectangular Cartesians to cylindrical coordinates is singular for $x_1=0$. Strictly speaking, this fact means that the equations (4.1-1) become meaningless for 5

 $\xi=0$ ($x_1=a\xi$, a=const.) and that consequently problem (I2) is not a proper one. From the first of the equations (4.1-1), however, we conclude that

$$\xi^2 \varphi_z - p_2 \xi^2 = \text{const.}$$

Now, this constant may be zero for some kind of motion and then

$$\varphi_z = p_2 = \text{const.} \tag{4.2-6}$$

Further, a string curve intersecting the axis of rotation ($\xi=0$) is permitted by this kind of motion, a motion which can easily be shown to correspond to the solution of problem (I2).

We now set a one-point problem equivalent to the problem (I2).

(12') Problem: To find functions $x_i(z)$, i=1,2,3,5, defined on some interval (a, b), a < 0 < b, such that they satisfy

a) A differential system Eq. (4.2-1) where, using the same notations as in the preceding section

$$x' \equiv (x_i) = (x_1, x_2, x_3, x_5),$$

(4.2 - 7)

$$A(x) = \begin{bmatrix} x_2 \\ -p_2^2 x_1 - 2p_1 x_1 x_5 \\ p_2 \\ -2p_1 x_1 x_2 \end{bmatrix}.$$

b) The initial value

$$x'(0) = (0, b_2, 0, (1+b_2^2)^{1/2}).$$
 (4.2-8)

As previously, p_1 , p_2 are real constants and b_2 is a positive constant. The initial value $x_2(0)=b_2$ is equivalent to the condition $\xi_z=0$ for $\xi=1, cf$. Eq. (4.1-5), since the solution of problem (I2) yields $\xi_z(0) > 0$, say $\xi_z(0)=b_2$, see Eq. (4.2-20).

Now, the elements of A(x) are integral functions of x_i , i=1,2,3,5, so that the solution of (I2'), and in consequence the solution of (I2), is analytic in z for every z such that x_i is finite. Moreover, the solution is analytic in the parameters p_1 , p_2 and the initial value b_2 .



Fig. VI. The graph explains the situation of case (1). For $\xi \in (\xi_3, \xi_2)$ the conditions $\xi_z^2 > 0$ and $\gamma > 1$ are satisfied. For $\xi > \xi_1$ one has $\xi_z^2 > 0$ but $\gamma < 0$, cf. note 28. In case (2) the two points of intersection, $\xi = \xi_2$ and $\xi = \xi_3$, disappear.

4.23. The Solutions of Problems (I1) and (I2)

Writing $p_2 = 2p_1h$ we get from Eq. (4.1-1) for problem (I1)

$$\begin{split} &\gamma + p_1 \xi^2 = A, \\ &\xi^2 \varphi_z - p_2 \xi^2 = B, \\ &\xi_z^2 = (A - p_1 \xi^2)^2 - 1 - \xi^{-2} (B + p_2 \xi^2)^2, \end{split} \tag{4.2-9}$$

where A and B are real constants (arbitrary, so far). We take $B \neq 0$, since B=0 leads to problem (I2).

There are now two possible cases:

1) A and B are such that ξ_z regarded as a function of ξ has three real zeros, $\xi_1 > \xi_2 > \xi_3 > 0.^{28}$ It is not difficult to show that the expressions Eq. (4.2-9) are physically significant in the interval $[\xi_3, \xi_2]$ only (see Fig. VI).

2) A and B are such that ξ_z has only one zero. It is easily demonstrated that this case has no physical significance.

It is now clear that for any steady rotatory motion with positive reduced string tension and such that the string curve does not intersect the axis of rotation, the function $\xi(z)$ can assume the values $\xi_3 \leq \xi \leq \xi_2$ only. Further, $\xi_z = 0$ for $\xi = \xi_2$ and $\xi = \xi_3$. It is evident that no physical

²⁸ It should be noted that (by definition) $\gamma > 1$ and that we may take ξ to be positive. The constant p_1 is also positive.
specialization is imposed on such a motion by the introduction of the initial values Eq. (4.1-2). Together with the restrictions formulated in Eq. (4.1-3) the initial values imply that letting $\xi_3=1$ we normalize the string curve (cf. section 4.12, point (3)) and that we fix it so that $\xi=1$ occurs at $\varphi=0, z=0$.

By varying b we can cover any possible string motion of the type in question here. It should be observed that for given $p_1 > 0$ and p_2 we cannot choose b arbitrarily but are restricted by Eq. (4.1-3). The latter restriction implies that b must be chosen outside the open interval (b_2, b_1) where $b_2 < 0$ and $b_1 > 0$ are roots of the equation

$$f(b) \equiv b^2 - 2p_1(1+b^2)^{1/2} - 2p_2b = 0. \tag{4.2-10}$$

From Eq. (4.2-9) we now obtain

$$\begin{split} &(\xi\xi_z)^2 = p_1^2(\xi^2 - 1)(k_2 - \xi^2)(k_1 - \xi^2),\\ &\xi^2\varphi_z = b + p_2(\xi^2 - 1),\\ &\gamma = (1 + b^2)^{1/2} - p_1(\xi^2 - 1), \end{split} \tag{4.2-11}$$

where we have to regard

$$\xi^2 \in [1, k_2]. \tag{4.2-12}$$

We have introduced k_1 and k_2 instead of ξ_1^2 and ξ_2^2 . The following relations are valid:

$$\begin{split} k_1 &= Q + \frac{1}{p_1} \left[p_1^2 Q^2 - (b - p_2)^2 \right]^{1/2}, \\ k_2 &= Q - \frac{1}{p_1} \left[p_1^2 Q^2 - (b - p_2)^2 \right]^{1/2}, \\ Q &= \frac{1}{2} \left[1 + \left(\frac{p_2}{p_1} \right)^2 + \frac{2}{p_1} (1 + b^2)^{1/2} \right], \\ f(b) &\equiv b^2 - 2p_1 (1 + b^2)^{1/2} - 2p_2 b \ge 0. \end{split}$$

$$(4.2 - 13)$$

Excluding the special cases $b=b_1$ or $b_2, f(b_1)=f(b_2)=0$, we can integrate the first of the equations (4.2-11) directly, we then obtain $z=z(\xi)$ as an elliptic integral of the first kind, *cf.* [B2], pag. 72. Inverting, we get $\xi=\xi(z)$ as a periodic function; its domain of definition may be taken to be the whole z axis. Then, $\gamma(z)$ is obtained directly from the last of the equations (4.2-11) and $\varphi(z)$ from the second one. After some calculation we get the following solution of problem (I1):

$$\begin{split} & \xi(z) \!=\! [1 \!+\! (k_2 \!-\! 1) \mathrm{sn}^2(\zeta(z))]^{1/2}, \\ & \varphi(z) \!=\! p_2 z \!+\! \frac{b \!-\! p_2}{p_1} \, (k_1 \!-\! 1)^{-1/2} \, \varPi \left(\zeta(z), \, k_2 \!-\! 1\right), \\ & \gamma(z) \!=\! (1 \!+\! b^2)^{1/2} \!-\! p_1(k_2 \!-\! 1) \mathrm{sn}^2(\zeta(z)), \\ & \zeta(z) \!=\! z p_1(k_1 \!-\! 1)^{1/2}, \, z \!\in\! (-\infty, \infty). \end{split}$$

In these expressions the modulus k of $\operatorname{sn}(\zeta) = \operatorname{sn}(\zeta, k)$ and $\Pi(\zeta, k_2-1) = = \Pi(\zeta, k_2-1, k)$ is determined by

$$k^{2} = \frac{k_{2} - 1}{k_{1} - 1}, \ 0 < k < 1, \tag{4.2-15}$$

where k_1 and k_2 are the same as in Eq. (4.2–13) and f(b) > 0 holds. Further, the constants p_1 and p_2 are real and $p_1 > 0$, while p_2 is not restricted. The solution of problem (I1) given by Eq. (4.2–14) will be discussed in the next section.

In the special cases f(b)=0 we have

$$\xi(z) \equiv 1,$$

 $\varphi(z) = bz,$ (4.2-16)
 $\gamma(z) \equiv (1+b^2)^{1/2},$

where b is a root of the equation (4.2-10). In these two cases the string curve is a *circular helix*.

We now turn to problem (I2). From system Eq. (4.1-1) and the discussion in section 4.22 we obtain

$$\begin{split} &\gamma + p_1 \xi^2 = A, \\ &\varphi_z = p_2, \\ &\xi_z^2 = (A - p_1 \xi^2)^2 - 1 - p_2^2 \xi^2. \end{split} \tag{4.2-17}$$

It is clear that A > 1 (cf. note 28) and it is easily deduced that the preceding expressions will be physically significant for $\xi \in [-\xi_2, \xi_2]$ only, where $\xi = \xi_2$ is the smallest positive zero of ξ_z regarded as a



Fig. VII. The graph shows that the conditions $\xi_z^2 > 0$ and $\gamma > 1$ are satisfied for $|\xi| < \xi_2$. For $|\xi| > \xi_1$ one has $\gamma < 0$, cf. note 28.

function of ξ (see Fig. VII). Normalizing the string curve, we write $\xi_2=1$ and take $\xi=\varphi=0$ for z=0, *i. e.* we introduce the conditions Eqs. (4.1-4) and (4.1-5). Consequently any steady rotatory motion with positive reduced string tension such that the string curve intersects the axis of rotation is determined by the solution of problem (I2). From Eq. (4.2-17) we obtain

$$\begin{split} \xi_z^2 &= \left(\frac{p_1}{k}\right)^2 (1 - \xi^2)(1 - k^2 \xi^2), \\ \varphi_z &= p_2, \\ \gamma &= (1 + p_2^2)^{1/2} + p_1(1 - \xi^2), \end{split} \tag{4.2-18}$$

where the range of ξ is [-1, 1] and k is determined by

$$k = \left[1 + \left(\frac{p_2}{p_1}\right)^2 + \frac{2}{p_1} \left(1 + p_2^2\right)^{1/2}\right]^{-1/2}.$$
 (4.2-19)

After some elementary calculations we get the following solution of problem (I2):

$$\begin{split} \xi(z) &= \operatorname{sn}\left(z\frac{p_1}{k}\right), \\ \varphi(z) &= p_2 z, \\ \gamma(z) &= (1+p_2^2)^{1/2} + p_1 \operatorname{cn}^2\left(z, \frac{p_1}{k}\right), \\ z &\in (-\infty, \infty). \end{split}$$

$$(4.2-20)$$

The modulus k of sn $\left(z\frac{p_1}{k}\right)$ and cn $\left(z\frac{p_1}{k}\right)$ is determined by Eq. (4.2-19). p_1 and p_2 are real constants and p_1 is positive, while p_2 is not restricted.

4.24. Discussion of the Solutions of Problems (I1) and (I2)

In this section we shall briefly discuss the solutions of problems (I1) and (I2) with special regard to the possibilities of obtaining a complete determination of all the string properties, starting from the physical conditions in one or two points (cf. section 4.11). It is only to be expected that one-point problems are more easily approached than two-point problems. Three physical quantities will be considered known a priori, viz. m, ω , and H (the string mass per unit length, the angular velocity of the string curve, and the quantity of string mass flowing through every surface z = const. per unit of time). These quantities are constants in the case of an inextensible string in steady motion. We first discuss the solution of problem (I1) and start by transforming Eq. (4.2-14) into physical variables. We also add expressions for the arc length s of the string curve and the string tension T. From equations (1.2-8) and (3.1-8) we get $ds = a\gamma dz$ and from Eqs. (3.1-8) and (3.1-10) $T = T_0(\gamma \bar{\sigma} + 2p_1 \varrho \gamma^2 \mu^2)$. Now, $\bar{\sigma} \equiv 1$, $\varrho \equiv 1$ and $\gamma \mu \equiv h = \frac{p_2}{2p_1}$, so we obtain $T = T_0 \left(\gamma + \frac{p_2^2}{2p_1} \right)$. We recall that $x^1 = a\xi$, φ , $x^3 = az$ are the cylindrical coordinates of the string point observed from a frame of reference rotating with the string curve. With respect to such a frame, every string point has a constant velocity $uq = a\omega h = \frac{H}{m}$ (cf. section 3.22, point (1)). After some calculation we obtain the following arrangement of the physical string properties as functions of the axial coordinate z (cf. Eq. (4.2-14)): $x^{1}(z) = a [1 + (k_{2} - 1) \operatorname{sn}^{2}(\zeta(z))]^{1/2}$

$$\begin{split} &\varphi(z) \!=\! p_2 z + \frac{b \!-\! p_2}{p_1} (k_1 \!-\! 1)^{-1/2} \Pi(\zeta(z), \, k_2 \!-\! 1), \\ &s(z) \!=\! a \left\{ z(1 \!+\! b^2)^{1/2} \!-\! (k_1 \!-\! 1)^{1/2} \![\zeta(z) \!-\! E(\zeta(z))] \right\}, \quad (4.2 \!-\! 21) \\ &T(z) \!=\! T_0 \left[-\frac{p_2^2}{2p_1} \!+\! (1 \!+\! b^2)^{1/2} \!-\! p_1(k_2 \!-\! 1) \operatorname{sn}^2(\zeta(z)) \right], \\ &\zeta(z) \!=\! z p_1(k_1 \!-\! 1)^{1/2}, \, z \!\in\! (-\infty, \infty). \end{split}$$

According to Eqs. (4.2–15) and (4.2–13) the constants k, k_1 and k_2 may be expressed in terms of p_1 , p_2 and b. T_0 and a may also be expressed in terms of p_1 , p_2 (and, of course, in the *a priori* constants m, ω , H). We then get from Eqs. (3.1–9), (4.2–4) and the above equality $h = \frac{H}{a\omega m}$

$$a = \frac{p_1}{p_2} \frac{2H}{m\omega},$$
$$T_0 = \frac{p_1}{p_2^2} \frac{2H^2}{m}.$$

(4.2 - 22)

The three constants p_1 , p_2 and b are the dimensionless parameters of the string,²⁹ and if they are known, it then follows from the expressions Eq. (4.2-21) that all the physical string properties are known in every point of the string curve. Instead of these three parameters we may use a, T_0 and b which have the advantage of possessing simple physical interpretations (a is the smallest distance from any point of the string curve to the axis of rotation, T_0 is the axial component of the reduced string tension vector, and b is the cotangent of the angle of elevation of the string curve at a point of distance a from the axis of rotation). We prefer to use p_1 , p_2 and b, since they are dimensionless and yield simple expressions. In order to obtain the expressions Eq. (4.2-21) we have introduced a simple set of onepoint conditions, viz.

$$x^{1}(0) = a, \ x^{1}_{z}(0) = 0, \ \varphi(0) = 0,$$

$$\varphi_{z}(0) = b, \ s(0) = 0.$$
(4.2-23)

and we have also introduced the parameter $T_0.^{30}$ It is most improbable, however, that T_0 and such simple conditions as Eq. (4.2–23) should be available in a physical problem, cf. section 4.11. The main difficulty in solving a particular problem of steady rotatory motion of an inextensible string will consist in determining the string parameters p_1 , p_2 , b (or

²⁹ A more precise statement would be: p_1 , p_2 and b are the three parameters of the (inextensible) string in case of a steady rotatory motion such that the string curve does not intersect the axis of rotation.

³⁰ We could also introduce T_0 by means of an initial value of the string tension T, viz. $T(0) = \frac{H^2}{m} + T_0(1+b^2)^{1/2}$. This fact can be verified by Eqs. (4.2-21) and (4.2-22).

a, T_0 , b) and adjusting the string curve in such a way with respect to the cylindrical coordinate system that the available set of boundary conditions and other conditions are satisfied. For instance, let the motion satisfy conditions (3) of section 4.11. We then have to find string parameters p_1 , p_2 , b and z_A , z_B that satisfy

$$\begin{aligned} x^{1}(z_{A}) &= x_{A}^{1}, \quad x^{1}(z_{B}) = x_{B}^{1}, \\ \varphi(z_{B}) - \varphi(z_{A}) &= \delta, \\ z_{B} - z_{A} &= \frac{1}{a} C = \frac{p_{2}}{p_{1}} \frac{m\omega}{2H} C, \end{aligned}$$
(4.2–24)

where x_A^1 , x_B^1 , δ and C are known quantities and $x^1(z)$ and $\varphi(z)$ are the same as in Eq. (4.2-21). (It should be noted that C is the physical distance between the planes $z=z_A$ and $z=z_B$.) In order to be able to determine the unknown quantities we need one more condition, as predicted in point (6) of section 4.11. The determination of p_1 , p_2 , b, z_A , z_B will generally involve intricate computations, especially in such cases where several different string motions have to be joined.

It must be pointed out that we cannot expect the determination of the set p_1 , p_2 , b, z_A , z_B to be unique. It is known that infinitely many solutions are theoretically possible for a simple motion corresponding to a special case of problem (I2), cf. the paper by NERONOFF, [N1], containing a discussion on stability; see also [H6], [K2], and note 26. This state of things has nothing to do with the uniqueness of the solutions of the problems (I1) and (I2) as stated here.

So far, we have discussed the case of a steady rotatory motion, when the string curve does not intersect the axis of rotation (corresponding to the problem (I1)). We now turn to the case of an intersection between the string curve and the axis of rotation, corresponding to problem (I2). This is a most special case, in so far as there are small chances that any set of one-point or two-point conditions would produce such an intersection. In several textile devices, however, for example spinning machines, the string is forced to pass through a small guide eye placed centrally with reference to the axis of rotation. It is not difficult to show that the solution of problem (I1) turns into the solution of (I2) when we pass to the limits $a \rightarrow 0$, $b \rightarrow p_2$.³¹ The solu-

 $^{^{31}}$ It should be observed that *a* has different meanings in problems (I1) and (I2), respectively.

tion of (I2) consequently provides an approximate solution of (I1) for the kind of problem outlined above.

Transforming the solution Eq. (4.2-20) of the problem (I2) into *physical variables* and adding expressions for the arc length of the string curve s and the string tension T we get

$$\begin{split} x^{1}(z) &= a \, \operatorname{sn}\left(z \, \frac{p_{1}}{k}\right), \\ \varphi(z) &= p_{2}z, \\ s(z) &= a \, \left[-z \left(\frac{p_{2}^{2}}{p_{1}} + (1 + p_{2})^{1/2}\right) + \frac{1}{k} E\left(z \, \frac{p_{1}}{k}\right)\right], \quad (4.2 - 25) \\ T(z) &= T_{0} \left[\frac{p_{2}^{2}}{2p_{1}} + (1 + p_{2})^{1/2} + p_{1} \operatorname{cn}^{2}\left(z \, \frac{p_{1}}{k}\right)\right], \end{split}$$

where k is determined by Eq. (4.2-19) and we choose s(0)=0. The equalities Eq. (4.2-22) hold true in this case too. There are two independent parameters p_1 , p_2 in Eq. (4.2-25). Alternatively, we may use a, T_0 instead of p_1 and p_2 . The comments on Eq. (4.2-21) are in the main valid for Eq. (4.2-25) also, but the computational difficulties are considerably smaller for Eq. (4.2-25), since there are only two parameters to be determined and the elliptic integral of the third kind in the expression for $\varphi(z)$ has disappeared.

4.3. The Steady Rotatory Motion of an Elastic String in the Case of Positive Reduced String Tension and Small Strain

In this section we treat the basic problems (E1) and (E2), *i. e.* the analogues for an elastic string to the problems (I1) and (I2). We shall assume that the reduced string tension is positive, *i. e.* $\overline{T} > 0$, and that the strain of the string is small everywhere, *i. e.* $\alpha = kT_0 \ll 1$, *cf.* section 3.22. According to point (3) of section 3.22 we put $\bar{\sigma} = 1$ in the system Eqs. (3.2-4), (3.2-5). Excluding the first equation of Eq. (3.2-4) we have

$$\begin{aligned} &(\xi^{2}\varphi_{z})_{z} - 2p_{2}\xi\xi_{z} = 0, \\ &\gamma_{z} + 2p_{1}\varrho\xi\xi_{z} = 0, \\ &\xi_{z}^{2} + (\xi\varphi_{z})^{2} + 1 = \gamma^{2}, \end{aligned} \tag{4.3-1}$$

and from Eq. (3.2-5)

$$\varrho = \exp\left[-\frac{\beta}{p_1}\left(\gamma + \frac{p_2^2}{2p_1}\frac{1}{\varrho}\right)\right]. \tag{4.3-2}$$

The parameters p_1 , p_2 and β of these equations are connected with the a priori constants m_0 , ω , H, k and the characteristical quantities a, T_0 according to

$$p_{1} = \frac{1}{2} m_{0} \omega^{2} \frac{a^{2}}{T_{0}},$$

$$p_{2} = 2p_{1} h = \omega H \frac{a}{T_{0}},$$

$$\beta = \alpha p_{1} = \frac{1}{2} m_{0} \omega^{2} k a^{2}.$$
(4.3-3)

The system Eq. (4.3-1) consists of three ordinary, nonlinear differential equations for the functions $\xi(z)$, $\varphi(z)$, $\gamma(z)$. Eq. (4.3-2) is a transcendental equation which determines ϱ as a function of γ . The parameters are real and $p_1 > 0$, $p_2 \neq 0$, $0 \leq \beta \leq 1$. We exclude the special case $p_2=0$ (which implies zero tangential velocity), since this case will be treated in section 4.4. The function $\varrho(\gamma) = \varrho(\gamma, \beta)$ turns out to be multiple-valued mathematically, but if we take $\varrho(\gamma) \rightarrow 1$ as $\beta \rightarrow 0$, then $\varrho(\gamma)$ is analytic in γ and β for every γ and $p_1 > 0$, if β is sufficiently small.³² This is a consequence of a theorem on implicit analytic functions, see for instance [S2], p. 170.

Compared to the corresponding system Eq. (4.1-1) of the inextensible string Eq. (4.3-1) is further complicated by the presence of the variable $\varrho = \varrho(\gamma)$. This complication is not serious from the mathematical point of view, however, since Eq. (4.3-1) is analytic in its variables and parameters. The solution of a one-point problem in connection with Eq. (4.3-1) will consequently be analytic in the independent variable z, the initial values and the parameters (in the neighbourhood of $\beta = 0$). The solution therefore admits of series expansions after powers of β . This property is most valuable, since the exact solution of the problem (for arbitrary values of β), though it probably might be found, is too complicated to be of practical interest. We will therefore seek approximations, linear in β , to the

³² It should be observed that $\beta = 0$ means an inextensible string for which $\rho \equiv 1$.

solution of our problem. The search for linear approximations does not need any special justification; it corresponds to the common approximations in the classical theory of elasticity.

It would be natural to apply a linearization on approaching the problem of approximation, *i. e.* to add terms linear in β to the solutions of the corresponding problems for the inextensible string. Neglecting terms $O(\beta^2), \beta \rightarrow 0$, a system of linear differential equations would then result from Eq. (4.3-1). This system, however, is singular at the turning points $\xi_z=0$, as may be seen from Eq. (4.3-1). Such a difficulty might be surmounted by an application of the PLK method, *cf.*, for instance, [T2], but the attempts in this direction by the author have failed so far. Moreover, the coefficients of the differential system will consist of elliptic functions and such a system is not easily treated, however simple the boundary conditions may be.³³

In order to avoid the difficulties just pointed out we shall apply a method which is a most special one, since its application is confined to this very problem, or at least to a small class of problems.³⁴ The details of the method will be elucidated later on.

In subsequent deductions we shall frequently meet statements in the form, the function $f=f(x, p_1, p_2, \beta)$ is analytic in its variable xand the parameter β . It should be understood that the statement is true for every x of current interest, if β is in a sufficiently small neighbourhood of zero and the parameters p_1 , p_2 are real, and $p_1>0$.

We now return to the system Eqs. (4.3-1), (4.3-2). Expanding after powers of β and making some elementary calculations, we get

$$\begin{split} \varrho &= 1 - \frac{\beta}{p_1} \left(\gamma + \frac{p_2^2}{2p_1} \right) + \beta^2 G_1, \\ \gamma &= A - p_1(\xi^2 - 1) \left[1 - \frac{1}{2} \beta(2Q - \xi^2) \right] + \beta^2 G_2, \end{split} \tag{4.3-4}$$

where $G_1 = G_1(\gamma, \beta)$ is analytic in the variable γ and the parameter β , and further $G_1 = O(1), \beta \rightarrow 0$. $G_2 = G_2(\xi, \beta)$ is analytic in ξ and β and $G_2 = O(1), \beta \rightarrow 0$. A is an arbitrary constant and Q is a constant determined by

$$2Q = 1 + \left(\frac{p_2}{p_1}\right)^2 + \frac{2}{p_1}A.$$
(4.3-5)

³³ Cf. Chapter 6 where a related problem is discussed.

³⁴ A mathematician might well ask if it deserves to be termed a method.

From Eq. (4.3-1) we also obtain

$$\begin{split} \xi_z^2 &= \gamma^2 - 1 - \xi^{-2} (B + p_2 \xi^2)^2, \\ \varphi_z &= \xi^{-2} (B + p_2 \xi^2), \end{split} \tag{4.3-6}$$

where B is an arbitrary constant.

From now on we must consider the two problems (E1) and (E2) separately.

4.31. Problems (E1) and (E2)

We now formulate the basic problem (E1) for the steady rotatory motion of an elastic string, when the strain is small and the string curve does not intersect the axis of rotation.

Let the functions $\xi'(z, \beta)$, $\varphi'(z, \beta)$, $\gamma'(z, \beta)$ of the variable z and the parameter β be defined, for z on some interval [a', b'], a' < 0 < b', and for β in a neighbourhood of $\beta = 0$. Let these functions satisfy

a) The differential system Eq. (4.3-1), where $\varrho = \varrho(\gamma, \beta)$ is defined by Eq. (4.3-2).

b) The initial values

$$\begin{aligned} \xi'(0,\beta) &= 1, \ \varphi'(0,\beta) = 0, \\ \varphi'_{z}(0,\beta) &= b, \ \gamma'(0,\beta) = (1+b^{2})^{1/2}, \end{aligned}$$

$$(4.3-7)$$

where b is a real constant such that for some positive number M

$$f(b) = b^2 - 2p_1(1+b^2)^{1/2} - 2p_2b > M.$$
(4.3-8)

We call the functions $\xi(z, \beta)$, $\varphi(z, \beta)$, $\gamma(z, \beta)$ a solution of the problem (E1) on the interval $(a, b) \subset [a', b']$, a < 0 < b (cf. note 27), if, for $z \in (a, b)$ and in a neighbourhood of $\beta = 0$, they satisfy

c) The initial values Eq. (4.3-7).

d) The conditions

$$\xi = \xi', \ \varphi = \varphi', \ \gamma = \gamma' = O(\beta^2), \ \beta \to 0. \tag{4.3-9}$$

e) Some further conditions of continuity.

The following comments may be made concerning problem (E1): 1) The condition Eq. (4.3-8) means that $\xi'(0, \beta)=1$ is a local minimum of $\xi'(z, \beta)$ provided that β is small enough. This condition implies no specialization, *cf.* point (3) of section 4.12. 2) The conditions of continuity imposed on ξ , φ , γ will depend on the kind of string property we want to approximate. If, for instance, we want the approximation of $\frac{d^n \xi'}{dz^n}$, (linear in β), then $\xi(z, \beta)$ must be of the class C^n and the condition $\frac{d^n}{dz^n}$ $(\xi - \xi') = O(\beta^2), \beta \rightarrow 0$, must hold.

The method employed in this thesis will provide functions $\xi(z, \beta)$, $\varphi(z, \beta)$, $\gamma(z, \beta)$ which are analytic in z and β , and further supply approximations of derivatives of any order.

3) It will become clear in the next section that $\xi'(z, \beta)$ must be a periodic function of z; the domain of definition of ξ', φ', γ' may accordingly be taken to be the whole z axis. The functions ξ, φ, γ (solution of (E1)) may also be defined on the whole z axis and provide the required approximations valid for any finite z, if β is small enough.

We now proceed to the problem (E2) which is the counterpart of (E1) when the string curve intersects the axis of rotation. We use a formulation of the problem (E2) similar to that given for (E1), with the exception that we replace the conditions Eq. (4.3-7) by the following expressions:

$$\xi'(0,\beta) = \varphi'(0,\beta) = 0, \qquad (4.3-10)$$

$$\xi'_{z} = 0 \text{ for } \xi' = 1.$$
 (4.3–11)

Comments (2) and (3) on problem (E1) are valid for (E2), too, and Eq. (4.3-10) means that z=0 is a point of intersection between the string curve and the axis of rotation. The condition Eq. (4.3-11) means that the amplitude of $\xi'(z, \beta)$ is normalized to be 1 and does not imply any specialization, cf. point (3) in section 4.12.

4.32. The Solution of Problem (E1)

Proceeding now to the solution of problem (E1) it is evident that the functions $\xi'(z, \beta)$, $\varphi'(z, \beta)$, $\gamma'(z, \beta)$ must satisfy equations (4.3-4) and (4.3-6). We may here make the same investigations concerning the constants A and B as we did in section 4.23. Since the influence of the parameter β on γ' and ξ'_z may be made as small as we like, the result will be essentially the same as in the case of the inextensible string. We can therefore conclude that A and B must be such that ξ'_z , regarded as a function of ξ' and β , has three zeros $\xi'_1(\beta) > \xi'_2(\beta) >$ $> \xi_3(\beta) > 0$. These zeros are analytic in β and are real for any real β . The expressions Eqs. (4.3-4) and (4.3-6) are physically significant for $\xi' \in [\xi'_3, \xi'_2]$ only. We may put $\xi'_3(\beta) = 1$ which means that we let the string parameter a be the smallest distance between the string curve and the axis of rotation. Just as in problem (I1) we can introduce the initial values Eq. (4.3-7) without loss of general validity. The special case $\xi'_2 = \xi'_3$ is excluded by the inequality Eq. (4.3-8). We consider this helix case separately in section 4.34. We shall find it convenient to write $(\xi'_2)^2 = k_2 n'_2$ where $n'_2 = 1 + O(\beta), \beta \rightarrow 0$. (It should be observed that k_2 here is the same parameter as in Eq. (4.2-11).

It is our aim to find an approximation $\xi(z, \beta)$ of $\xi'(z, \beta)$, and the key to the problem is a quartic in $(\xi'_z)^2$ which provides an approximation, linear in β , of the expression for $(\xi'_z)^2$, *i. e.* the first of Eqs. (4.3-6). The next step in the derivation of this quartic consists in expressing ξ'_z as a function of ξ' and β in a suitable form. The expression so obtained must be valid for $(\xi')^2 \in [1, k_2 n'_2]$. After some elementary calculations we get the following expressions which (in the case of small strain) hold true for any steady rotatory motion of an elastic string which does not intersect the axis of rotation,

$$\begin{split} (\xi'\xi_{z}^{'})^{2} &= p_{1}^{2}(1+m\beta)((\xi')^{2}-1)(k_{2}n_{2}^{'}-(\xi')^{2}) \times \\ &\times (k_{1}n_{1}-(\xi')^{2})(1+\beta(\xi')^{2})(1+\beta^{2}G_{3}), \\ (\xi')^{2}\varphi_{z}^{'} &= b+p_{2}((\xi')^{2}-1), \\ \gamma' &= (1+b^{2})^{1/2}-p_{1}((\xi')^{2}-1)\left[1-\frac{1}{2}\beta(2Q_{1}-(\xi')^{2})\right]+\beta^{2}G_{2}, \end{split}$$

$$(4.3-12)$$

where $(\xi')^2 \in [1, k_2 n'_2]$, $G_2 = G_2(\xi', \beta)$ and $G_3 = G_3(\xi', \beta)$ are analytic in ξ' and β and are real for real values of ξ' and β . They possess the properties $G_2 = O(1)$, $G_3 = O(1)$, $\beta \rightarrow 0$. Moreover, $n_1 = n_1(\beta)$ and $n'_2 = n'_2(\beta)$ are analytic in β . The following expressions determine the parameters in Eq. (4.3-12):

$$\begin{array}{c} m = -(k_1 + k_4), \\ n_1 = n_1(\beta) = 1 - \beta \ \frac{k_2 k_4}{k_1 - k_2}, \\ n_2' = n_2'(\beta) = 1 + \beta \ \frac{k_1 k_3}{k_1 - k_2} + O(\beta^2), \ \beta \to 0, \end{array} \right\}$$
(4.3-13)

$$\begin{array}{c} k_{1} = Q_{1} + R, \quad k_{2} = Q_{1} - R, \\ k_{3} = Q_{2} + R, \quad k_{4} = Q_{2} - R, \\ 2 Q_{1} = 1 + \left(\frac{p_{2}}{p_{1}}\right)^{2} + \frac{2}{p_{1}} (1 + b^{2})^{1/2}, \\ 2 Q_{2} = 1 - \left(\frac{p_{2}}{p_{1}}\right)^{2}, \\ R = \frac{1}{p_{1}} [p_{1}^{2}Q_{1}^{2} - (b - p_{2})^{2}]^{1/2}. \end{array} \right\}$$

$$(4.3 - 14)$$

The constants m, k_1, k_2, k_3, k_4 are all real for $p_1 > 0$ and we have $k_1 > k_2 > 1$. These facts can be verified directly, but they can also be concluded from section 4.23 and inequality (4.3-8). (In section 4.23 k_1 and k_2 are real and have the same meaning as here, and if f(b) > 0 in Eq. (4.2-13), then $k_1 > k_2 > 1$.) Therefore, n_1 and n'_2 are real if β is real, and obviously $n_1 = 1 + O(\beta), n'_2 = 1 + O(\beta), \beta \rightarrow 0$. The inequalities $k_1n_1 > k_2n'_2 > 1$ then hold, if β is sufficiently small.

Regarding z as a function of $t \equiv (\xi')^2$, the fact that z_t is singular of the order $O(x^{-1/2})$, $x \rightarrow 0$, at x=1 and $x=k_2n'_2$ is readily obtained from the first of the equations (4.3-12). Within this interval z_t is finite and positive (by a proper choice of sign), if β is sufficiently small. If we then put $z(1,\beta)=0$, the function $z=z(t,\beta)$ is defined for $t \in [1, k_2 n'_2]$ and can be obtained by means of quadrature. Moreover, $z(t, \beta)$ is strictly increasing and the inverse function $(\xi')^2 \equiv$ $\equiv t \equiv t(z, \beta)$ consequently exists and is defined on $z \in [0, z(k_2n'_2, \beta)]$. Further, $t(z, \beta)$ may be re-defined as a periodic function of z with the period $2z(k_2n'_2,\beta)$. Once we have determined $(\xi')^2$ as a function of z and the parameter β , $\gamma' = \gamma'(z, \beta)$ will be obtained directly from the last of Eqs. (4.3-12). Integrating once with respect to z we get $\varphi' = \varphi'(z, \beta)$ from the second of the equations (4.3-12). In order to be able to determine the functions $\xi'(z, \beta)$, $\varphi'(z, \beta)$, $\gamma'(z, \beta)$ we must calculate the functions G_2 and G_3 that appear in Eq. (4.3-12), e.g. by computing their series expansions after powers of β . It would be very laborious to do so and we therefore restrict ourselves to the search for functions $\xi(z, \beta), \varphi(z, \beta), \gamma(z, \beta)$ satisfying conditions (c), (d) and (e) of section 4.31. In other words, we proceed to find a solution of problem (E1).

As a solution of the problem (E1) we choose functions $\xi(z, \beta)$, $\varphi(z, \beta)$, $\gamma(z, \beta)$ such that they satisfy

a) The system

$$\begin{split} \xi^2 &=\! t, \, \xi \!>\! 0, \\ t_z^2 &=\! 4 p_1^2 (1\!+\!m\beta)(t\!-\!1)(k_2 n_2\!-\!t)(k_1 n_1\!-\!t)(1\!+\!\beta t), \\ t \varphi_z &=\! b\!+\! p_2(t\!-\!1), \\ \gamma &=\! (1\!+\!b^2)^{1/2} \!-\! p_1(t\!-\!1) \big[1\!-\!\frac{1}{2} \,\beta(2Q_1\!-\!t) \big], \end{split} \tag{4.3-15}$$

where $t \in [1, k_2n_2]$; and n_2 is determined by

$$n_2 = 1 + \beta \frac{k_1 k_3}{k_1 - k_2}.$$
 (4.3–16)

Eqs. (4.3-13) and (4.3-14) are valid for the other parameters. b is assumed to satisfy Eq. (4.3-8).

b) The initial values Eq. (4.3-7), i. e.

$$t(0, \beta) = 1, \ \varphi(0, \beta) = 0.$$
 (4.3-17)

In order to be sure that ξ , φ , γ do constitute a solution of (E1) we must show that the conditions Eq. (4.3-9) are satisfied everywhere in the domain of definition of the functions. At first glance it might appear evident that these conditions should be satisfied, since the system Eq. (4.3-15) is obtained from Eq. (4.3-12) by omitting terms of the order $O(\beta^2)$, $\beta \rightarrow 0$, but owing to the singularities of the integrand in the integral which determines $z=z(t,\beta)$ things become somewhat complicated. We therefore discuss this question in greater detail in Appendix A1.

After some calculations we get the following solution of problem (E1):

$$\begin{split} \xi^{2}(z, \beta) &= 1 + (\varkappa_{2} - 1)(1 + \beta) \operatorname{sn}^{2}(\lambda z) \left[1 + \beta \varkappa_{2} - \beta(\varkappa_{2} - 1) \operatorname{sn}^{2}(\lambda z) \right]^{-1}, \\ \varphi(z, \beta) &= z [p_{2}(1 + \beta) - \beta b] + (b - p_{2})(1 + \beta) \frac{1}{\lambda} \Pi(\lambda z, \delta^{2}), \quad (4.3 - 18) \\ \gamma(z, \beta) &= (1 + b^{2})^{1/2} - p_{1}(\xi^{2} - 1) \left[1 - \frac{1}{2} \beta (2Q_{1} - \xi^{2}) \right], \\ where \ \xi > 0 \ and \ z \in [a, b] \subset (-\infty, \infty). \end{split}$$

The parameters λ , k, δ^2 appearing in Eq. (4.3-18) are determined by

$$\begin{split} \lambda &= \lambda(\beta) = \varkappa_0 [(\varkappa_1 - 1)(1 + \beta \varkappa_2)]^{1/2}, \\ k^2 &= k^2(\beta) = (\varkappa_2 - 1)(1 + \beta \varkappa_1) (\varkappa_1 - 1)^{-1}(1 + \beta \varkappa_2)^{-1}, \ k > 0, \quad (4.3 - 19) \\ \delta^2 &= \delta^2(\beta) = (\varkappa_2 - 1)(1 + \beta \varkappa_2)^{-1}, \end{split}$$

where k is the modulus of $\operatorname{sn}(\lambda z) = \operatorname{sn}(\lambda z, k)$ and $\Pi(\lambda z, \delta^2) = \Pi(\lambda z, \delta^2, k)$. For the parameters $\varkappa_0, \varkappa_1, \varkappa_2$ we have

$$\begin{split} &\varkappa_{0}^{2} = \varkappa_{0}^{2}(\beta) = p_{1}^{2}(1+m\beta) = p_{1}^{2}[1-\beta(k_{1}+k_{4})], \ \varkappa_{0} > 0, \\ &\varkappa_{1} = \varkappa_{1}(\beta) = k_{1}n_{1} = k_{1}[1-\beta k_{2}k_{4}(k_{1}-k_{2})^{-1}], \\ &\varkappa_{2} = \varkappa_{2}(\beta) = k_{2}n_{2} = k_{2}[1+\beta k_{1}k_{3}(k_{1}-k_{2})^{-1}], \end{split}$$

$$(4.3-20)$$

where k_1 , k_2 , k_3 , k_4 are given by Eq. (4.3-14) which also determines Q_1 . If the four parameters β , p_1 , p_2 , b are given, then the solution Eq. (4.3-18) of problem (E1) is completely determined. As was shown in a previous discussion $\varkappa_1 > \varkappa_2 > 1$ and consequently 0 < k < 1 for sufficiently small values of β .

For $\beta = 0$ the solution of (E1) turns into the solution of (I1). This trivial result is readily perceived from Eqs. (4.3–18) and (4.2–14).

By the method applied here we have obtained an approximation $\xi(z, \beta)$ of the function $\xi'(z, \beta)$, where ξ' is determined by Eq. (4.3-12) and the initial value $\xi'(0, \beta) = 1$. The approximation is true up to terms linear in β . It is also possible to obtain approximations of higher order, since we can write the first equation of Eq. (4.3-12)

$$(\xi'\xi'_z)^2 = Q_n(\xi',\beta)(1+\beta^n H),$$

where Q_n is a polynomial in $(\xi')^2$ the degree of which is increasing with n, and $H=H(\xi',\beta)$ is analytic in ξ' and β . Let $\xi=\xi(z,\beta)$ be the *n*th order approximation of $\xi'(z,\beta)$ with respect to β . As previously for n=1, we can determine ξ from the differential equation

$$(\xi\xi_z)^2 = Q_n(\xi,\beta)$$

and the initial value $\xi(0, \beta) = 1$. We obtain $\xi(z, \beta)$ as the inverse function of a hyperelliptic integral the class of which depends on *n*, *cf*. [B2], p. 252.

4.33. The Solution of Problem (E2)

We start by considering the functions $\xi'(z, \beta), \varphi'(z, \beta), \gamma'(z, \beta)$ which satisfy Eqs. (4.3-4) and (4.3-6), just as we did for problem (E1). In this case the string curve intersects the axis of rotation ($\xi=0$); the constant B in Eq. (4.3-6) must then be zero and we get $\varphi_z = p_z =$ =const. β is a small parameter, so we conclude, as for problem (I2) in section 4.23, that A is a constant such that ξ'_z , regarded as a function of ξ' and β , is physically significant for $\xi' \in [-\xi'_2, \xi'_2]$. At the end points of the interval $\xi'_z = 0$. Without loss of general validity we can take $\xi'_2 = 1$, and we then let the characteristical length a be the greatest distance between any string point and the axis of rotation. We also let the origin of our cylindrical coordinate system be a point of intersection between the string curve and the axis of rotation, and the conditions Eqs. (4.3-10) and (4.3-11) are then satisfied. After some calculation we get the following system from Eqs. (4.3-4) and (4.3-6), a system which (in case of small strain) is valid for any steady rotatory motion of an elastic string intersecting the axis of rotation:

$$\begin{split} &(\xi'\xi'_z)^2 = p_1^2(\xi')^2(1-(\xi')^2)(2Q-(\xi')^2)(q+\beta(\xi')^2)(1+\beta^2G_4),\\ &\varphi'_z = p_2,\\ &\gamma' = &(1+p_2^2)^{1/2} - p_1((\xi')^2-1)\big[1-\frac{1}{2}\beta(2Q-(\xi')^2)\big] + \beta^2G_2, \end{split}$$

where $(\xi')^2 \in [-1,1]$; $G_2 = G_2(\xi', \beta)$ and $G_4 = G_4(\xi', \beta)$ are analytic in ξ' and β . The parameters Q and q are determined in β , p_1 , p_2 by

$$2Q = 1 + \left(\frac{p_2}{p_1}\right)^2 + \frac{2}{p_1}(1+p_2^2)^{1/2},$$

$$q = 1 - \beta \left(1 + \frac{1}{p_1}(1+p_2^2)^{1/2}\right).$$
(4.3-22)

It should be observed that 2Q > 1, since by definition $p_1 > 0$, cf. Eq. (3.1-9). Then, for small values of β , $\xi'_z \neq 0$ for $\xi' \in (-1,1)$, and we may take $\xi'_z > 0$ in that interval. According to Eq. (4.3-10) we put z=0for $\xi'=0$, and we can obtain $z=z(\xi',\beta)$ from Eq. (4.3-21) by means of quadrature. $z(\xi',\beta)$ would be strictly increasing, and it consequently does possess an inverse $\xi' = \xi'(z,\beta)$. From the last of the equations (4.3-21) we obtain $\gamma'=\gamma'(z,\beta)$. $\varphi'=\varphi'(z,\beta)$ is readily derived from Eq. (4.3-21) and the condition Eq. (4.3-10). Since the evaluation of $\xi'(z, \beta)$ in this way would be a laborious procedure, we restrict ourselves to search for such solutions of the problem (E2) as specified in section 4.31.

As a solution of problem (E2) we choose functions $\xi(z, \beta)$, $\varphi(z, \beta)$, $\gamma(z, \beta)$ such that they satisfy

a) The system

$$\begin{split} \xi^2 &= t, \\ t_z^2 &= 4 p_1^2 t (1-t) (2Q-t) (q+\beta t), \\ \varphi_z &= p_2, \\ \gamma &= (1+p_2^2)^{1/2} - p_1 (t-1) \Big[1 - \frac{1}{2} \beta (2Q-t) \Big], \end{split} \tag{4.3-23}$$

where $t \in [0,1]$ and the parameters Q and q are the same as in Eq. (4.3-22). b) The initial values

$$t(0, \beta) = \varphi(0, \beta) = 0.$$
 (4.3-24)

In order to constitute a solution of the problem (E2) the functions ξ , φ , γ must also satisfy the conditions Eq. (4.3-9). The proof might be carried out in the same way as for problem (E1), *cf.* Appendix A1.

After some calculation we get the following solution of problem (E2):

$$\begin{split} \xi(z,\,\beta) &= \operatorname{sn}(\lambda z) \Big[\, 1 - \beta \operatorname{cn}^2(\lambda z) \Big]^{1/2}, \\ \varphi(z,\,\beta) &= p_2 z, \\ \gamma(z,\,\beta) &= (1 + p_2^2)^{1/2} + p_1 (1 - \xi^2) \Big[\, 1 - \frac{1}{2} \beta (2Q - \xi^2) \Big], \end{split}$$

$$(4.3 - 25)$$

where $z \in [a, b] \subset (-\infty, \infty)$.

The elliptic modulus k and the parameters λ and Q are determined by

$$\begin{aligned} k &= (2Q)^{-1/2} [1 + \beta (2Q - 1)]^{1/2}, \\ \lambda &= p_1 (2Q)^{1/2} \left[1 - \frac{1}{2} \frac{\beta}{p_1} (1 + p_2^2)^{1/2} \right], \\ 2Q &= 1 + \left(\frac{p_2}{p_1} \right)^2 + \frac{2}{p_1} (1 + p_2^2)^{1/2}. \end{aligned}$$

$$(4.3 - 26)$$

If the three parameters β , p_1 , p_2 are given, then the solution Eq. (4.3-25) of the problem (E2) is completely determined.

For $\beta = 0$ the solution of (E2) turns into the solution of (I1), as might be expected.

4.34. A Special Case of Problem (E1)

On excluding condition Eq. (4.3-8), the problem (E1) may have a solution for which the string curve is a *circular helix* at certain values of the parameters p_1 , p_2 and b.

Introducing the initial values Eq. (4.3-7), we find that the string curve is a helix, if p_1 , p_2 and b are such that the equation

$$\frac{1}{2}f(b)(1+b^2)^{-1/2} + \beta \left[(1+b^2)^{1/2} + \frac{1}{2}p_2^2 p_1^{-1} \right] - \beta^2 G_1((1+b^2)^{1/2},\beta) = 0$$
(4.3-27)

holds true. In Eq. (4.3–27) f(b) is the same function as in Eq. (4.3–8), and $G_1(\gamma, \beta)$ is the same function as in Eq. (4.3–4), *i. e.* G_1 is analytic in γ and β , and $G_1=O(1), \beta \rightarrow 0$. The equation (4.3–27) means that $\xi_{zz}=0$ for $\xi=1, \varphi_z=b, \gamma=(1+b^2)^{1/2}$, and it is easily derived from the first of the equations (3.2–4) and also from Eq. (4.3–1). Eq. (4.3–27) may also be considered to be the condition for $k_2n'_2=1$ in Eq. (4.3–12). After some elementary calculations we find

$$b = b_0 + \beta b_1 + O(\beta^2), \quad \beta \to 0,$$
 (4.3-28)

where b_0 and b_1 satisfy

$$\begin{split} f(b_0) &= 0, \\ b_1 &= -\frac{1 + b_0^2}{4b_0} \left[b_0^2 + 2p_1(1 + b_0^2)^{1/2} \right]^2 \left[2p_1 + b_0^2(1 + b_0^2)^{1/2} \right]^{-1}. \end{split} \tag{4.3-29}$$

 $f(b_0)=0$ always has one positive and one negative root (*i. e.* for any p_2 and $p_1>0$), cf. section 4.23.³⁵

There is one value of b_1 , of course, for each of the roots of $f(b_0)=0$, and Eq. (4.3-29) shows that they are both finite. In the case of an elastic string we may regard b_1 as a linear correction to the values $b=b_0$ which we obtained for an inextensible string. There is no difficulty in

³⁵ It should be noted that b_1 is not the same constant here as in section 4.23, where it is the positive root of f(b)=0.

calculating higher order terms in the series expansion of b after powers of β (the elastic parameter).

To sum up, we may claim that the string curve may be a helix for any pair of permissible values of p_1 and p_2 , if b assumes either of the two values satisfying Eq. (4.3-27). These two values of b are different in sign, and if b is positive the curve will be a right-handed helix and vice versa. b is the cotangent of the helix angle.

4.35. Discussion of the Solutions of Problems (E1) and (E2)

The expressions Eq. (4.3–18) and Eq. (4.3–25) give the solutions to the basic problems (E1) and (E2) for the steady rotatory motion of an elastic string in the case of small strain and the absence of external forces. The solutions are correct up to terms linear in the elastic parameter β . The four parameters, β , p_1 , p_2 , b, constitute the solution of (E1); the parameter b does not appear in the solution of (E2).

Compared with the solutions of the corresponding problems for the inextensible string, (I1) and (I2), the solutions of (E1) and (E2) are more complicated. In principle, however, the difficulties connected with the determination of the string properties starting from physical boundary conditions are the same regarding both the elastic string and the inextensible string.

The evaluation of the physical string properties may be carried out in a manner similar to that used in section 4.24. The determination of the total string mass in an arbitrary interval on the string curve, *i. e.* the evaluation of the integral $\int m_0 a \varrho \gamma dz$, will cause no difficulties; $\varrho = \varrho(\gamma)$ is given by Eq. (4.3-4) and $\gamma = \gamma(\xi)$ by Eq. (4.3-15) or (4.3-23). (This particular problem does not occur in case of an inextensible string since in that case the string mass is proportional to the length of the string curve.)

The physical string properties will be completely determined if we are able to determine the parameters p_1 , p_2 , b (or, equivalently, a, T_0 , b) on condition that the *a priori* constants m_0 , ω , H, k are known. It should be observed that the elastic parameter β can be determined in p_1 and p_2 (and the *a priori* constants) by means of Eq. (4.3-3); the latter equation also gives the connection between a, T_0 on the one hand and p_1 , p_2 on the other. In the case the string curve intersects the axis of rotation, problem (E4), parameter bdisappears.

There are consequently *three* parameters (*two* in the case of intersection) to be determined from the boundary conditions and other available physical conditions. This situation is quite similar to that occurring for the inextensible string. It is also to be expected that for a particular string problem we can obtain a first order approximation of the solution, if we regard the string as inextensible. It must be pointed out, however, that the parameter values p_1 , p_2 , b, obtained in this way, must be modified when elasticity is taken into account.

4.4. The Steady Rotatory Motion Without Tangential Velocity of an Elastic String

When the string rotates without tangential velocity, an arbitrary string point describes a circle which has its centre on the z axis. It is evident from section 3.1 that u=0, $\overline{T}=T$ hold true everywhere on the string, or, in dimensionless variables, $\mu=0$, $\bar{\sigma}=\sigma$. When no external forces act on the string, it follows from section 3.2 that $\bar{\sigma}=\sigma=$ const. and as previously we can take $\bar{\sigma}=1$. This means that the string parameter T_0 is the component of the string tension vector on the z axis. From Eqs. (3.2-4) and (3.2-7) the following equations then result

$$\exp\left(\frac{\beta}{p_1}\gamma\right) + \beta\xi^2 = A,$$

$$\xi^2 \varphi_z = B,$$

$$\xi_z^2 + (\xi \varphi_z)^2 + 1 = \gamma^2.$$

(4.4-1)

In these equations A > 0 and B are arbitrary real constants and p_1 , β are parameters, cf. Eq. (4.3-3). It should be noted that no restriction is laid upon the magnitude of the strain in the string. Two different cases now appear, corresponding to (E1) and (E2) (the equivalent problems in case of small strain and non-zero tangential velocity).

1) $B \neq 0$; the string curve has no point of intersection with the axis of rotation. We conclude from Eq. (4.4-1) that in this case A and B must be such that

$$\begin{split} &\xi_z = 0 \quad \text{if } \xi = \xi_1 \quad \text{or } \xi = \xi_2, \\ &\xi_z^2 > 0 \quad \text{if } \xi \in (\xi_1, \, \xi_2), \end{split}$$

where $\xi_2 > \xi_1 > 0$. We can take $\xi_1 = 1$, as we did in problems (I1) and (E1), and it then implies that the characteristical length a in the problem is the smallest distance between the string curve and the axis of rotation. The basic one-point problem which results from $B \neq 0$ will be called (E3).

2) B=0; the string curve intersects the axis of rotation in one or more points. We conclude from Eq. (4.4-1) that in this case $\varphi = \text{const.}$, *i.e.* the string describes a plane curve. Except for the trivial solution $\xi \equiv 0$, a real solution of Eq. (4.4-1) can only exist, if A is such that

$$\begin{aligned} \xi_z^2 > 0 \quad \text{for } \xi \in (-\xi_1, \xi_1), \\ \xi_z = 0 \quad \text{for } |\xi| = \xi_1. \end{aligned}$$

We put $\xi_1 = 1$ and the characteristical length a is then the greatest distance between the string curve and the axis of rotation. The basic one-point problem connected with this kind of motion will be called (E4).

4.41. Problem (E3)

(E3) Problem: To find functions $\xi(z)$, $\varphi(z)$, $\gamma(z)$ defined on some interval (a_1, a_2) , $a_1 < 0 < a_2$ and constants A and B such that the following conditions obtain

- a) The system Eq. (4.4-1) is satisfied.
- b) The initial conditions

$$\xi(0)=1, \ \varphi(0)=0, \ \varphi_{z}(0)=b, \ \gamma(0)=(1+b^{2})^{1/2}$$
 (4.4-2)

are satisfied. b is a real constant which satisfies the inequality³⁶

$$f(b) \equiv b^2 - 2p_1(1+b^2)^{1/2} \exp\left(-\frac{\beta}{p_1} (1+b^2)^{1/2}\right) > 0.$$
 (4.4-3)

Otherwise b is an arbitrary quantity. p_1 and β are parameters of the system connected with the *a priori* constants m_0 , ω , k and the characteristical quantities a and T_0 according to the expressions Eq. (4.3-3).

³⁶ This inequality proves that $\xi(0) = 1$ is a local minimum. The helix case f(b) = 0 is treated separately.

The solution of problem (E3) is easily obtained. Eqs. (4.4-1), (4.4-2) imply that $t(z) \equiv (\xi(z))^2$ must satisfy a separable differential equation

$$t_z^2 = 4g(t),$$

where

$$g(t) = t \left[\frac{p_1}{\beta} \log(C - \beta(t - 1)) \right]^2 - t - b^2,$$

$$C = \exp\left(\frac{\beta}{p_1} (1 + b^2)^{1/2}\right).$$
(4.4-4)

It will be evident from previous discussions that g(t) > 0 for $t \in (1, t_2)$, where t_2 depends on the parameters p_1 , β and the initial value b. In the neighbourhood of t=1 and $t=t_2$ g(t) satisfies

$$\begin{split} g(1\!+\!\varepsilon)\!=\!\varepsilon g_t(1)\!+\!O(\varepsilon^2), & \varepsilon\!\!\rightarrow\!\!+0, \\ g(t_2\!-\!\varepsilon)\!=\!-\varepsilon g_t(t_2)\!+\!O(\varepsilon^2), & \varepsilon\!\!\rightarrow\!\!+0, \end{split}$$

where $g_t(1) > 0$, $g_t(t_2) < 0$. z = z(t) can consequently be obtained by quadrature. Taking z(1)=0, cf. Eq. (4.4-2), we get

$$z(t) = \frac{1}{2} \int_{1}^{t} (g(\tau)^{-1/2} d\tau. \qquad (4.4 - 5)$$

z(t) is defined on $[1,t_2]$. We now introduce $L \equiv z(t_2)$, *i. e.*

$$2L = \int_{1}^{t_{1}} (g(\tau)^{-1/2} d\tau.$$
(4.4-6)

Now, since z(t) is strictly increasing, it possesses an inverse $F(z) \equiv t(z) \equiv (\xi(z))^2$ which is defined on [0, L]. By the introduction of F(-z) = F(z), F(z+2L) = F(z), F(z) is extended to be an even periodic function with the period 2L. As a solution of the problem (E3) we then obtain

$$\begin{split} \xi(z) &= (F(z))^{1/2}, \\ \varphi(z) &= b \int_{0}^{z} (F(\zeta))^{-1} d\zeta, \\ \gamma(z) &= \frac{p_{1}}{\beta} \log[C - \beta(F(z) - 1)], \end{split} \tag{4.4-7}$$

where C is a constant determined by Eq. (4.4-4) and F(z)=t(z) is the inverse of z(t) which is defined by Eqs. (4.4-5) and (4.4-4).

The period 2L of $\xi(z)$ is determined by Eq. (4.4-6), where $t=t_2$ is the second zero of g(t) (the first is t=1). The physical distance $x^1(z)=a\xi(z)$ between the string points and the axis of rotation is immediately obtained from Eq. (4.4-7), and the derivation of the string tension T(z), and the arc length s(z) of the string curve can easily be carried out, cf. section 4.24. It must be observed, however, that the integral defining z(t) (the inverse of F(z)) is not an elementary one.

We now turn to the *helix case* which has been exluded so far by the condition Eq. (4.4-3). It is clear that the equation f(b)=0 has two solutions, $b=\pm b_1$.³⁷ If $b=\pm b_1$, we have

$$\begin{split} \xi(z) &\equiv 1, \\ \varphi(z) &= \pm b_1 z, \\ \gamma(z) &\equiv (1 + b_1^2)^{1/2}. \end{split} \tag{4.4-8}$$

In this case the string curve will obviously be a circular helix.

4.42. Problem (E4)

(E4) Problem: To find functions $\xi(z)$, $\varphi(z)$, $\gamma(z)$, defined on some interval (a_1, a_2) , $a_1 < 0 < a_2$, and constants A and B such that the following conditions obtain:

- a) The system (4.4-1) is satisfied.
- b) The initial conditions

$$\xi(0) = \varphi(0) = 0$$
 (4.4-9)

are satisfied.

c) The condition

$$\xi_z = 0 \text{ for } \xi = 1$$
 (4.4–10)

is satisfied.

The solution of this problem is obtained by straight-forward calculation. As mentioned previously, the constant B must be zero

³⁷ If the string has a non-vanishing tangential velocity, the negative and positive roots are different in magnitude, owing to the acceleration of Coriolis.

and the string curve is a plane one, *i. e.* $\varphi = \varphi(z) = \text{const.}$ The solution of problem (E4) may be written

$$\begin{split} &\xi(z)\!=\!F_{0}(z),\\ &\varphi(z)\!\equiv\!0,\\ &\gamma(z)\!=\![1\!+\!F_{0}^{2}(z)]^{1/2}, \end{split} \tag{4.4-11}$$

where $F_0(z)$ is the inverse function of $z(\xi)$; the latter is defined by

$$z(\xi) = \int_{0}^{\xi} (g_{0}(\tau))^{-1/2} d\tau,$$

$$g_{0}(\tau) = \left[\frac{p_{1}}{\beta} \log\left(\exp \frac{\beta}{p_{1}} + \beta(1 - \tau^{2})\right)\right]^{2} - 1.$$
(4.4-12)

By the extension $F_0(-z) = -F_0(z), F_0(2L_0-z) = F_0(z)$, where

$$L_0 = z(1) = \int_0^1 (g_0(\tau))^{-1/2} d\tau, \qquad (4.4 - 13)$$

we may take the domain of definition of the solution Eq. (4.4–11) to be the whole z axis. $\xi(z) = F_0(z)$ is obviously an odd, periodic function with the period $4L_0$.

4.43. Discussion of the Solutions of Problems (E3) and (E4)

Fundamental theorems on ordinary differential systems indicate that the solutions of problems (E3) and (E4) are unique and analytic in the independent variable z, the elastic parameter β , the parameter p_1 , and the initial value b. The solutions are of general validity in the sense that, if a problem concerning a steady rotatory motion without tangential velocity of a string does have a solution, the latter is then contained in the solution of either problem (E3) or problem (E4).³⁸

No restrictions have been imposed with regard to the magnitude of the strain in the string, but the expressions obtained are naturally only valid for a string which is linearly elastic in our sense of the term, see section 1.25.

Evidently, other kinds of elasticity will give rise to different expres-

³⁸ On condition that no external forces act on the string in the domain in question.

sions, but if the strain in a point of the string is completely determined by the string tension in that point, then a system essentially identical to Eq. (4.4—1) is valid. It is only the function $\exp(\beta p_1^{-1}\gamma)$ in the first equation that has to be changed for another function of the variable γ . The function $\exp\left(\frac{\beta}{p_1}\gamma\right) = \frac{p_1}{\beta}\int \exp\left(\frac{\beta}{p_1}\gamma\right)\gamma_z dz$ comes from the third equation in the system Eq. (3.2—4) and change of the elastic condition simply leads to another function of γ (other than the exponential one) under the integral sign. The ultimate consequence of a change to another kind of elasticity (than the linear one) is that the functions g(t) and $g_0(t)$ that occur in the solutions of problems (E3) and (E4) have to be changed for other related functions.

The practical difficulties in finding a solution of a particular string problem connected with problem (E3) or (E4), *i. e.* to find the characteristical quantities a, T_0 (or alternatively the parameters p_1 , β , cf. Eq. (4.3-3)) and the initial value b, are related to those discussed in section 4.24. An additional complication is caused by the presence of non-elementary functions.

There must exist a close connection between the solutions of the problems (E1) and (E2) on the one hand, and (E3) and (E4) on the other, since, putting $p_2=0$ in the former case, we get approximations linear in β for the latter. It is also possible, of course, to obtain these approximations directly from the solutions of (E3) and (E4). We cannot, however, expand the functions $(g(\tau))^{-1/2}$ or $(g_0(\tau))^{-1/2}$ after powers of β and then integrate termwise, since the intervals of integration contain singular points. The solutions of (E3) and (E4), in the form stated here, are consequently not suitable in the case of problems involving small strain.

5. The Steady Rotatory Motion of an Inextensible String in the Case of Negative Reduced String Tension and in the Absence of External Forces

The symbols and notations used in this chapter are essentially the same as those in Chapter 4.

5.1. Introduction

There are two different kinds of steady rotatory string motion possible, as shown in section 3.22. The first one is well-known in textile mechanics, and is characterized by the fact the so-called reduced string tension $T-mu^2q^2$ is positive. This type of steady motion was studied in some detail in chapter 4. The second type, characterized by negative reduced string tension, has apparently not been observed or discussed earlier.

It will become clear that motion with negative reduced string tension is possible only if the tangential velocity uq (cf. section 1.22) is higher than the highest circumferential velocity occurring in any real string point, see Fig. VIII. By a real string point we mean a



Fig. VIII. The broken curve (circle) is the path of a point on the string curve, fixed with respect to a co-rotating frame. It should be noted that this path does not coincide with the path of any string point. The circumferential velocity is tangential to the above curve.

point, where the string tension is not negative, *i. e.* $T \ge 0$. (String points with T < 0 may occur in formal solutions of string problems, but the solution has no physical significance in such points.)

The author suggests that motions where the above condition is satisfied do not occur in industrial practice, although they are easy to generate. As no theoretical objections have been advanced against the existence of the steady rotatory motion with negative reduced tension, the author has considered it proper briefly to investigate this type of motion in spite of the above-mentioned suggestion. We will only discuss motions analogous to those of problem (I2), *i. e.* motions of an inextensible string such that the string curve intersects the axis of rotation. For this kind of motion we must distinguish between two cases, as will become apparent later on. If we had to consider motions in which the string does not intersect the axis of rotation, *i. e.* problems analogous to (I1), matters would become more complicated. Moreover, problems for an elastic string with negative reduced string tension would involve tedious calculations.

5.2. The String Equations. The Two Cases

We shall start from the string equations in the form of Eq. (3.2-4). Since we shall deal solely with an inextensible string, we put $\varrho = \varrho(z) \equiv 1$, and just as in Chapter 4, we introduce $p_2 = 2p_1h = 2p_1\gamma\mu$. In accordance with remark (3) in section 3.22 we take $\bar{\sigma} = \text{const.} = -1$. Observing that a change of sign for $\bar{\sigma}$ in Eq. (3.2-4) has the same effect as a change of sign for p_1 and p_2 , it becomes evident that we can apply the deductions carried out for problem (I2). We also have to take into consideration that the string tension σ (the dimensionless form) must not be negative; this condition was automatically satisfied in problem (I2). By appropriate changes of sign and adding the condition $\sigma \geq 0$, we then get from Eqs. (4.2-17)

$$\begin{split} \xi_{z}^{2} &= (A + p_{1}\xi^{2})^{2} - 1 - p_{2}^{2}\xi^{2}, \\ \varphi_{z} &= -p_{2}, \\ \gamma &= A + p_{1}\xi^{2}, \\ \sigma &= 2p_{1}\gamma\mu^{2} - 1 = p_{2}^{2} \frac{1}{2p_{1}\gamma} - 1 \ge 0, \end{split}$$
(5.2-1)

where A > 1 is a constant of integration. It is true that any steady rotatory motion of an inextensible string must satisfy Eq. (5.2-1) provided that the string intersects the axis of rotation and that the external forces are neglected. We now have to consider two different cases, depending on the magnitude of the constant A. If A is sufficiently small compared to p_1 and p_2 , ξ_z will become zero for two positive values of ξ^2 , and we then get a situation similar to that in problem (I2). For large values of $A \xi_z^2$ will be positive for any (real) ξ^2 , and we then get entirely different solutions of Eq. (5.2-1). A further complication is caused by the condition $\sigma \geq 0$.

The "critical" value of A will be called A_0 ; it is determined by

$$A_{0} = \frac{p_{1}}{p_{2}^{2}} + \frac{1}{4} \frac{p_{2}^{2}}{p_{1}}.$$
 (5.2–2)

 $\sigma \ge 0$ implies $p_2^2 > 2p_1$, hence $A_0 > 1$. We now have the two cases:

a) $A_0 > A > 1$. In this case there are two values ξ_1 and $\xi_2, \xi_2 > \xi_1 > 0$, depending on p_1 , p_2 and A, and such that $\xi_z^2 = 0$ for $|\xi| = \xi_1$, and $|\xi| = = \xi_2$. It turns out that $\sigma > 0$ holds for $|\xi| \le \xi_1$, and Eq. (5.2–1) is consequently physically significant for $|\xi| \le \xi_1$, if $A_0 > A$.

b) $A > A_0$. In this case $\xi_z^2 > 0$ for every real ξ , but $\sigma \ge 0$ only holds for $|\xi| \le \xi_3$ where ξ_3 depends on p_1 , p_2 and A.

We call the basic boundary value problems connected with these two cases (IN2a) and (IN2b). I means inextensible string, N negative reduced string tension, and 2 that the string intersects the axis of rotation, *cf.* (I2). On considering the two cases for given values of p_1 and p_2 we find that at the point of intersection γ is smaller in case (a) than in case (b). Now, $\frac{1}{\gamma}$ is the direction cosine of the string tangent with respect to the *z* axis, and the string consequently intersects the *z* axis at a smaller angle in case (a) than in case (b).

5.21. Problem (IN2a)

We can without specialization take the smallest zero of ξ_z to be $\xi_1=1$. This means that the characteristical length a of the problem is chosen as the greatest distance between any string point and the axis of rotation.

(IN2a) Problem: To find functions $\xi(z)$, $\varphi(z)$, $\gamma(z)$, $\sigma(z)$ defined on some interval (-b, b), a constant A, and the parameters p_1 , p_2 such that

a) The system Eq. (5.2-1) is satisfied.

b) The initial values

$$\xi(0) = \varphi(0) = 0 \tag{5.2-3}$$

are satisfied.

c) The condition

$$\xi_z = 0$$
 for $\xi = 1$ (5.2-4)

is satisfied.

The existence and uniqueness of the solution follows from basic theorems on differential systems, cf. section 4.22. The solution is readily obtained. From Eqs. (5.2-1) and (5.2-4) we get

$$A = (1 + p_2^2)^{1/2} - p_1 \tag{5.2-5}$$

The inequality $\sigma \geq 0$ is satisfied for every ξ of interest, if

$$p_2^2 \ge 2p_1(A+p_1) = 2p_1(1+p_2^2)^{1/2}$$

On solving this inequality with respect to p_2 we get

$$p_2^2 \ge 2p_1[p_1 + (1+p_1^2)^{1/2}] > 4p_1^2.$$
 (5.2-6)

A necessary condition for $\sigma \ge 0$ to hold is consequently that $|p_2| > 2p_1$, or, expressed in physical quantities, it is necessary that the tangential velocity uq satisfies

$$|uq| = a\omega |h| = \frac{a\omega}{2p_1} |p_2| > a\omega.$$

 $a\omega$ is the highest circumferential velocity of *any* point on the string, and this velocity must be lower than the tangential velocity (which is the same for every string point). Factoring the expression for ξ_z^2 in Eq. (5.2-1) we get by means of Eq. (5.2-5)

$$\xi_z^2 = p_1^2 \left(1 - \xi^2\right) \left[1 + \left(\frac{p_2}{p_1}\right)^2 - \frac{2}{p_1} \left(1 + p_2^2\right)^{1/2} - \xi^2\right]$$

It is evident that $\xi_z^2 > 0$ for $|\xi| < 1$, and also that ξ_z^2 has a simple zero for $|\xi|=1$, if we exclude equality in Eq. (5.2-6). Using the

results from problem (I2) in section 4.23 we get the following *solution* of problem (IN2a):

$$\begin{split} \xi(z) &= \operatorname{sn}\left(z \; \frac{p_1}{k}\right), \\ \varphi(z) &= -p_2 z, \\ \gamma(z) &= (1 + p_2^2)^{1/2} - p_1 \operatorname{cn}^2\left(z \; \frac{p_1}{k}\right), \\ \sigma(z) &= p_2^2 \frac{1}{2p_1 \gamma(z)} - 1, \end{split}$$
(5.2-7)

where $z \in (-\infty, \infty)$ and the modulus k of the elliptic functions is determined by

$$k = \left[1 + \left(\frac{p_2}{p_1}\right)^2 - \frac{2}{p_1} \left(1 + p_2^2\right)^{1/2}\right]^{-1/2}.$$
 (5.2-8)

The restriction Eq. (5.2-6) must be satisfied by p_1 and p_2 . If equality holds in Eq. (5.2-6) then $\xi \rightarrow 1 \Rightarrow z \rightarrow \infty$ as then the elliptic modulus $k \rightarrow 1$. The solution Eq. (5.2-7) is closely related to the solution of problem (I2), but has an important change of sign for the last term in the expression for k. We can transform the solution into physical variables without difficulty and also determine the length of the string curve, as we did in section 4.24 for problem (I2).

5.22. Problem (IN2b)

We can without specialization take $\xi_3=1$. This means that the characteristical length a of the problem is the distance between the axis of rotation and that point on the string, where the string tension T (or σ in the dimensionless form) becomes zero.

(IN2b) Problem: To find functions $\xi(z)$, $\varphi(z)$, $\gamma(z)$, $\sigma(z)$, defined on some interval (-b, b), a constant A, and the parameters p_1 and p_2 such that

a) The system Eq. (5.2-1) is satisfied.

b) The initial values

$$\xi(0) = \varphi(0) = 0$$
 (5.2-3)

are satisfied.

c) The conditions

$$\sigma = 0$$
 and $\xi_z = \varepsilon > 0$ for $\xi = 1$ (5.2-9)

are satisfied.

The condition $\xi_z = \varepsilon > 0$ in Eq. (5.2-9) has been introduced in order to guarantee that $\xi(z)$ has no turning point or, equivalently, that $A > A_0$. The solution of the problem is readily obtained. The constants A and ε can be determined in terms of the parameters p_1 and p_2 .

From Eqs. (5.2-1) and (5.2-9) we get

$$A = p_2^2 \frac{1}{2p_1} - p_1,$$

$$\varepsilon^2 = p_2^4 \frac{1}{4p_1^2} - p_2^2 - 1.$$
(5.2-10)

From the last expression we have

$$p_2^2 = 2p_1[p_1 + (1 + p_1^2 + \varepsilon^2)^{1/2}] > 2p_1[p_1 + (1 + p_1^2)^{1/2}] > 4p_1^2.$$
 (5.2-11)

A necessary condition for (IN2b) to have a solution is then that the tangential velocity of the string is higher than the highest circumferential velocity occurring in any real point of the string (*i. e.* any point with $\sigma \geq 0$). This condition also appeared in problem (IN2a). After some calculation we get the tollowing solution of problem (IN2b)

$$\begin{split} \xi(z) &= C \left[\frac{1 - \operatorname{cn} \zeta(z)}{1 + \operatorname{cn} \zeta(z)} \right]^{1/2}, \ z > 0, \\ \xi(-z) &= -\xi(z), \\ \varphi(z) &= -p_2 z, \\ \gamma(z) &= p_2^2 \frac{1}{2p_1} - p_1 [1 - \xi^2(z)], \\ \sigma(z) &= p_2^2 \frac{1}{2p_1 \gamma(z)} - 1, \\ \zeta(z) &= 2p_1 C z, \ z \in [-b, b]. \end{split}$$
(5.2–12)



Fig. IX. Graph showing the main feature of the string curve in the case (IN2b). The vertical asymptote of $\xi(z)$ is $z=(p_1C)^{-1}K(k)$. The point z=b, $\xi=1$ is a point of inflection.

In these expressions we have the elliptic modulus k. The constants k, C, and b are determined by

$$k = \left[\frac{1}{2} (1 + C^{-2})\right]^{1/2},$$

$$C^{4} = 1 + \left(\frac{\varepsilon}{p_{1}}\right)^{2}, C > 1,$$

$$b = \frac{1}{2p_{1}C} F(\varphi_{1}),$$

$$\varphi_{1} = \arccos\left(\frac{C^{2} - 1}{C^{2} + 1}\right),$$
(5.2-13)

where $F(\varphi_1) = F(\varphi_1, k)$ is the incomplete elliptic integral of the first kind. The constant ε is determined in terms of p_1 , p_2 by Eq. (5.2–10), and p_1 , p_2 are restricted by the fact that ε must be real. The transformation of the solution into physical variables is easily carried out, cf section 4.24. It is not difficult to show that the second derivative ξ_{zz} is negative for $z \in (0, b)$ and zero at z=b; the delineation of the function $\xi(z)$ is given by Fig. IX.

6. A Vibration Problem of the Inextensible String

Symbols and conventions:

τ

ξ

Y

y_i, i=1,...,7 Dimensionless variables describing a small time-dependent motion of the string and defined as in Chapter 3 for i=1,...,5 and by Eq. (6.2-3) for i=6,7.
z Dimensionless, independent variable, cylindrical co-

Dimensionless, independent variable, cylindrical coordinate defined by Eq. (3.1-6).

Independent variable, dimensionless time, defined by Eq. (3.1-6).

Variable of a steady rotatory motion, dimensionless radial distance, defined by Eq. (3.1-8).

Variable of a steady rotatory motion. $\frac{1}{\gamma}$ is the direction cosine of the string tangent with respect to the axis of rotation (the z axis).

 $p_1=2p$ String parameter, defined as in Chapter 3. p is used as perturbation parameter for the vibration problem. x Independent variable used instead of z and defined by Eq. (6.2-2).

 $g_{y}(x,p), r=1, ..., 6$ Functions defined by Eqs. (6.2-6) and (6.2-7).

 $g_{\nu,i}(x), i=1, 2, 3, \ldots$ Coefficients in the series expansion of $g_{\nu}(x, p)$ after powers of p.

- $\alpha = \alpha_{\nu} = \alpha_{\nu}(p), \ \nu = 1, 2, 3, ...$ Dimensionless natural, circular frequency of the vibratory motion, defined by Eq. (6.2-9). Ordinal number ν usually omitted.
- $\alpha_0 \equiv \alpha_{1,0}$ First term in the series expansion of $\alpha(p)$ after powers of p, zero order approximation.

 $\alpha_i, i=1, 2, 3, \ldots$ General term of the above expansion, pertubation of the order *i*.

a_1, a_2	End points of the basic interval of the x axis. $[a_1, a_2]$ is the interval of definition of the vibratory motion.
$A_{\mu}(x), B_{\mu}(x), \mu =$	1, 3, 6, 7 Dependent variables of the (separated) vibratory motion, see Eq. $(6.2-9)$.
$Y_{\mu}(x) = Y_{\mu}(x, p),$	$\mu\!=\!1,3,6,7$ Dependent variable used instead of A_{μ} and $B_{\mu}.$
$\boldsymbol{Y}_{\mu,0}(\boldsymbol{x})$	First term in the series expansion of $Y_{\mu}(x, p)$ after powers of p , zero order approximation.
$Y_{\mu,i}(x), i = 1, 2, 3$, General term of the above expansion, pertubation of the order i .
D, L, M	Differential operators, depending on p , defined by Eq. (6.2 -12).
$L_0 = M_0$	Differential operator obtained from L by putting $p=0$, first order 'approximation' of L and M .
$L_i, M_i, i = 1, 2, 3$	$B_{i,\ldots}$ 'Pertubation' of the order <i>i</i> of the differential operators <i>L</i> and <i>M</i> . L_i , M_i are first order differential operators.
C_0	Boundary value, see Eq. $(6.2-14)$.
$\lambda_n, n=1, 2, 3, \ldots$. Eigenvalue of a Sturm-Liouville eigenvalue problem, defined by Eq. $(6.3-13)$.
$u_n = u_n(x)$	Orthonormal eigenfunctions of the above problem.
$a_{0,n}, b_{0,n}$	Coefficients in the series expansions of $Y_{1,0}(x)$ and $Y_{6,0}(x)$ after the eigenfunctions $u_n(x)$, see Eq. (6.3 -15).
d_n	Expansion coefficient, defined by Eq. $(6.3-16)$.
E_n, H_n	Functions of λ_n and α_0 , defined by Eq. (6.3-16).
$(f, g) = \int_{a_1}^{a_2} fg dx$	Inner product of the functions $f(x)$, $g(x)$.
$S(\alpha_0)$	Function of α_0 , defined by Eq. (6.3-20).
$\Phi_{\eta,0}, \eta = 1, \ldots, 4$	Functions of x and α_0 , see Eq. (6.3-23).
$Y'_{\mu,1}(x), Y''_{\mu,1}(x), \mu = 1, 6$ Functions defined by Eq. (6.3-25).	
a' _{1, n}	Expansion coefficient, defined by Eq. $(6.3-30)$.
$T(\alpha_0)$	Function of α_0 defined by Eq. (6.3-33).
$\Phi_{\eta,i}, \eta=1,\ldots,4$; $i=1, 2, 3, \ldots$ Functions of x and of perturbations α_i up to the order <i>i</i> , see Eq. (6.3-34).

Symbols not included in the above list will occur now and then in the text.

Greek subscripts are used to denote numbering of various functions, Latin subscripts are used to denote ordinal numbers of approximations, perturbations, and also for eigenvalues, eigenfunctions and expansion coefficients. Latin superscripts denote powers.

The summation convention is used for Latin but not for Greek indices.

6.1. Introduction

In this chapter we shall study a problem of vibration in connection with the linear differential system Eq. (3.2-10). In this system the independent variables are z (the dimensionless axial distance of a point on the string curve) and τ (the dimensionless time). There are five dependent variables $y_i(z, \tau)$ and the system is valid for a small time-dependent motion superimposed on a special kind of steady rotatory motion of an inextensible string. The coefficients of the system are constituted by the periodic functions $\xi(z)$, $\gamma(z)$, and a real, positive parameter p_1 , they consequently do not depend on the time τ .

The functions $\xi(z)$ and $\gamma(z)$ are determined by the steady rotatory motion on which the time-dependent motion in question is superimposed. The steady motion was studied in a more general form in Chapter 4, and the functions $\xi(z)$ and $\gamma(z)$ may be taken from Eq. (4.2-20) if we put $p_2=0$; in that equation it means that the string has no tangential velocity at the steady motion. The string curve of the steady motion is a plane one and it intersects the axis of rotation at equidistant points.

The system Eq. (3.2-10) is linear and homogeneous and has coefficients with simple properties; yet most problems of technical interest connected with that system are quite complicated. This will be evident, if we recall that the system is not hyperbolic (the string is inextensible) and that curves $\tau = \text{const.}$ are characteristics, *cf.* Chapter 2. We shall therefore confine ourselves to a simple vibration problem, which we will describe more precisely in section 6.2.

In order to be able to obtain results of practical interest we must restrict our investigations to the case of small values of the parameter p_1 . It will be clear from Eq. (3.1-9) that a small value of p_1 means that the string tension T is large everywhere compared to the quantity

 $m_0 a^2 \omega^2$,³⁹ This is actually true in many problems of practical interest and our specialization is therefore justified. The treatment of the vibration problem for small values of the parameter p_1 is characterized by the use of the method of *separation of variables* and then the application of *a pertubation method* to the resulting system of ordinary, linear differential equations.

The system Eq. (3.2-10) must have a meaning for $p_1=0$, if the perturbation method is to work. It should be observed that the functions $\xi(z)$ and $\gamma(z)$, which constitute the coefficients of the system, depend explicitly on p_1 , cf. Eq. (4.2-19). It can easily be verified that the (real) period of the functions $\xi(z)$ and $\gamma(z)$ is 4K(k), where K(k) is the complete elliptic integral of the first kind. A simple investigation of Eqs. (4.2-19) and (4.2-20) shows that for $p_2=0$, $K=O(p_1^{-1/2})$, $p_1\rightarrow 0$. This property of K has the physical significance that if the string tension tends to infinity, then the ratio between the period and the amplitude (which is the characteristical length) of the string curve (in the steady motion) also tends to infinity. In order to overcome this difficulty we shall introduce $x=\mathrm{am}\left(z\frac{p_1}{k}\right)$ as a new independent

variable instead of z in the system Eq. (3.2-10). $\operatorname{am}(u) = \operatorname{am}(u, k)$ is the elliptic amplitude function, cf. [B2]. It will readily be seen that ξ , γ , and the coefficients of the system will be trigonometric functions of x with the period 2π .

We shall not concern ourselves with questions regarding the existence and uniqueness of the solution of the vibration problem, neither shall we discuss the convergence of the perturbation method applied. Consequently, we cannot affirm that the problem is properly stated; the results obtained are, of course, merely formal.

It may be remarked, however, that circumstances are favourable in so far as the coefficients of the ordinary differential system to be treated by the perturbation method will be analytic in the perturbation parameter $p=\frac{1}{2}p_1$ and the independent variable x, provided that p is sufficiently small.

³⁹ It will readily be observed that small values of p_1 correspond to large values of T_0 , and by Eq. (3.1-8) to large values of T; it should be noted that $\frac{q}{a} = \gamma \ge 1$ and that a is the greatest distance between the z axis and any point on the string curve (in the steady motion).
6.2. Transformation of the Basic System The Vibration Problem

The differential system Eq. (3.2-10) shall now be transformed into a more suitable form. Let us introduce the parameter p defined by

$$p = \frac{1}{2}p_1, \tag{6.2-1}$$

the independent variable x defined by

$$x = am\left(z\frac{p_1}{k}\right),\tag{6.2-2}$$

and the dependent variables $y_6(x, \tau)$ and $y_7(x, \tau)$ defined by

$$y_6 = y_2 \xi,$$

 $y_7 = y_5 \gamma \sqrt{p}.$ (6.2-3)

The following condition must hold in order to ensure the existence of $y_5(x, \tau)$ when $p \rightarrow 0$

$$y_7 = O(\sqrt{p}), p \to 0.$$
 (6.2-4)

Keeping in mind that the parameter p_2 of Eq. (4.2-20) is zero (the string has no tangential velocity in the steady motion on which the time-dependent motion is superimposed), we get the following system after some elementary calculation

$$\begin{split} y_{1xx} + g_1(y_1 - y_{.\tau\tau} + 2y_{6\tau}) + 3pg_2 \sin x \cos x \, y_{1x} - g_1 \sin x \, y_3 &= 0, \\ y_{6xx} + g_1(y_6 - y_{6\tau\tau} - 2y_{1\tau}) - pg_3 \sin x \cos x \, y_{6x} - 4g_4 \cos x \, y_7 &= 0, \\ y_{7x} + 2pg_5 \cos x \, y_{1x\tau} &= 0, \\ y_{3x} - 2g_4 y_{7\tau} &= 0, \\ y_4 - 4pg_6 \cos x \, y_{1x} &= 0. \end{split}$$
(6.2-5)

In this system $g_{\nu}=g_{\nu}(x, p), \nu=1, \ldots, 6$ are functions of the independent variable x and the parameter p. They are defined by the following expressions

$$g_{1} = G_{1}^{-1}G_{2}, \quad g_{2} = (G_{1}G_{2})^{-1}(1 + \frac{2}{3}p\cos^{2}x),$$

$$g_{3} = G_{1}^{-1}, \quad g_{4} = (G_{1})^{-1/2}, \quad (6.2-6)$$

$$g_{5} = G_{1}^{1/2}G_{2}^{-1}, \quad g_{6} = G_{1}G_{2}^{-1},$$

104

where

$$G_1 = G_1(x, p) = 1 + p \cos^2 x,$$

$$G_2 = G_2(x, p) = 1 + 2p \cos^2 x.$$
(6.2-7)

The functions $g_{\nu}(x, p)$ are analytic in x and p for every x, if p is sufficiently small. It is then clear that all the functions g_{ν} may be expanded after powers of p, and that the series obtained in this way are absolutely and uniformly convergent and posses a common, finite radius of convergence for every x. Moreover, the functions g_{ν} satisfy

 $g_{\nu}(x, p) = 1 + O(p), p \to 0, \nu = 1, \dots, 6.$ (6.2-8)

The first four equations of the system (6.2-5) constitute a system of linear, homogeneous partial differential equations for the functions $y_{\mu}(x, \tau), \mu = 1,3,6,7.$

The last equation of the system (6.2-5) defines the function $y_4(x, \tau)$ and can be excluded from the system. The coefficients of the differential system are trigonometric functions and (regular) analytic in the whole complex x plane, if p is sufficiently small. As mentioned previously the differential system is not hyperbolic. The unknown functions $y_{\mu}(x, \tau)$ represent small, time-dependent motions superimposed on a steady rotatory motion of the string. It will be recognized from section 3.21 and Eq. (6.2-3) that y_1 and y_6 correspond to the time-dependent changes of the radial and angular distances of the string points. Further, y_3 determines the string tension, and y_7 the tangential velocity. Finally, y_4 determines the direction cosine of the string tangent vector with respect to the axis of rotation, hence its connection with y_{1x} , cf. the last equation of Eq. (6.2-5).

We now turn to the vibration problem which we will formulate in the following way:

Problem: To find real constants α and functions $A_{\mu}(x)$, $B_{\mu}(x)$, $\mu=1,3,6,7$, defined on an interval $[a_1, a_2]$, and such that

a) The differential system Eq. (6.2-5) is satisfied, for $x \in (a_1, a_2)$ by

$$y_{\mu}(x,\tau) = A_{\mu}(x) \cos(\alpha \tau) + B_{\mu}(x) \sin(\alpha \tau).$$
 (6.2–9)

b) The homogeneous boundary conditions

$$y_{\nu}(a_1, \tau) = y_{\nu}(a_2, \tau) \equiv 0, \ \nu = 1, 6, 7$$
 (6.2–10)

are satisfied.



Fig. X. The figure illustrates the situation of the vibration problem. The end points of the string are fixed in the co-rotating plane at $x=a_1$ and $x=a_2$. The broken curve contained in the co-rotating plane is the string curve of the underlying steady motion.

The problem which we have set just now implies that we study an inextensible string on the interval $a_1 \leq x \leq a_2$. The end-points of the string are fixed in a plane rotating with constant angular velocity ω round the x axis which is contained in the plane and fixed in space, cf. Fig. X. The end-points of the string then move in two fixed circles which are assumed to be non-coplanar. The string has a small vibratory motion with the dimensionless period $\frac{2\pi}{\alpha}$ around its steady motion curve (the broken curve in the figure). We seek the possible modes of vibration and the natural frequencies α .

A few preliminary remarks may be made concerning the problem. The string curve of the steady motion is normalized so as to have the maximum distance a to the axis of rotation (it may happen, of course, that no turning point of the string curve falls in the interval (a_1, a_2)), and the curve intersects the axis of rotation at x=0. The effects of gravity and air resistance are disregarded. The length and shape of the string in steady motion is determined by Eq. (4.2-25), where we

have to put $p_2=0$, $p_1=2p$ and $\operatorname{sn}\left(z \; \frac{p_1}{k}\right)=\sin x$ as in Eq. (6.2-2). By varying the magnitudes of a, p, a_1 , a_2 , we can cover any possible type of the described motion. Later on we shall confine ourselves to small values of p, i. e. to flat string curves.

It should be observed that no boundary condition is imposed on the function $y_3(x, \tau)$ which corresponds to the string tension. This appears reasonable from the physical point of view.

We now return to the system Eq. (6.2-5) and substitute the ex-

$$LY_{1} + 2\alpha g_{1}Y_{6} - g_{1} \sin x Y_{3} = 0,$$

$$MY_{6} + 2\alpha g_{1}Y_{1} - 4g_{4} \cos x Y_{7} = 0,$$

$$DY_{7} - 2\alpha pg_{5} \cos x DY_{1} = 0,$$

$$DY_{3} - 2\alpha g_{4}Y_{7} = 0,$$

(6.2-11)

where we have used the symbols $Y_{\mu} = Y_{\mu}(x)$ for the unknown functions, and D, L, M symbolize the following differential operators

$$D = \frac{d}{dx},$$

$$L = D^2 + 3pg_2 \sin x \cos x \ D + g_1(1 + \alpha^2),$$

$$M = D^2 - pg_3 \sin x \cos x \ D + g_1(1 + \alpha^2).$$
(6.2-12)

The functions $g_{\nu} = g_{\nu}(x, p)$ are the same as in Eq. (6.2-6).

From the boundary conditions Eq. (6.2-10) we obtain

$$Y_{\mu}(a_1) = Y_{\mu}(a_2) = 0, \ \mu = 1, 6, 7.$$
 (6.2–13)

We have now to consider the sin $(\alpha\tau)$ terms which are obtained by the substitution of the expressions Eq. (6.2-9) in the system Eq. (6.2-5). It can easily be verified that the functions $B_1(x)$, $B_3(x)$, $-A_6(x)$, $-A_7(x)$ must satisfy the ordinary differential equations (6.2-11) and naturally the boundary conditions Eq. (6.2-13), too.

The essential properties of the vibratory motion in question are consequently determined by the solution of the boundary value problem constituted by the system Eq. (6.2-11) and the boundary conditions Eq. (6.2-13). It will prove convenient, however, to add one more boundary condition to the problem. We take this condition to be

$$Y_3(a_1) = C_0, \tag{6.2-14}$$

where $C_0 \neq 0$ is an arbitrary constant not depending on the parameter p.

We conclude from section 3.21 that this condition implies that the dimensionless reduced string tension $\bar{\sigma}$ at the end point $x=a_1$ of the string oscillates around its steady-state value $\bar{\sigma}_0=1$ according to the

expression $\varepsilon C_0 \cos(\alpha \tau)$. (A sin $(\alpha \tau)$ term may also be present, of course.) We may normalize the oscillation and take $C_0=1$, and the small parameter ε is then interpreted as the amplitude of the oscillation of the dimensionless reduced string tension at $x=a_1$. It will become clear from the perturbation method, as applied to the boundary value problem in the case of small parameter values p, that the functions $Y_{\mu}(x)$ become proportional to C_0 , while the natural frequencies α do not depend on C_0 . These circumstances are not surprising, if we recall that we are concerned with a small, linearized motion.

It may be expected that our boundary value problem will only have a solution, if α belongs to a particular set of real numbers which we will call the set of natural frequencies, or the Nf set. The determination of this set is the most important part of the solution, while the functions $Y_{\mu}(x)$ are of secondary interest. At this stage we know very little about the Nf set, but may expect it to be enumerably infinite with no point of accumulation other than infinity. It can easily be verified that if a number α belongs to the Nf set, then $-\alpha$ also does so. The Nf set must depend on the parameter p, too, but in what manner we do not know.

6.3. Outline of a Perturbation Method

6.31. Preliminary Considerations

In the following considerations we shall understand it as implied that α is a natural frequency; if necessary we shall indicate a particular natural frequency by means of a Greek subscript, e. g. α_{ν} . We now turn to the boundary value problem constituted by the differential system Eq. (6.2-11) and the boundary conditions Eqs. (6.2-13), (6.2-14); p being a small (real) parameter. We shall assume that the natural frequencies $\alpha = \alpha(p)$ and the functions $Y_{\mu}(x) = Y_{\mu}(x, p)$ have power series in p such that they permit all the subsequent derivations for $x \in (a_1, a_2)$ and p in a neighbourhood of p=0. We may consequently write

$$\begin{aligned} &\alpha(p) = p^{i} \alpha_{i}, \\ &Y_{u}(x, p) = p^{i} Y_{u, i}(x), \end{aligned} \tag{6.3-1}$$

where we have $\mu = 1,3,6,7$; $i=0,1,2,3,\ldots$, and summation on $i.^{40} \alpha_i$ and $Y_{\mu,i}(x)$ do not depend on p. Then, for the differential operators

⁴⁰ p^i of course means the *i*th power of p.

L, M and the coefficient functions $g_{\nu}(x, p)$ of the system Eq. (6.2-11), we get

$$L = p^{i}L_{i}, \quad M = p^{i}M_{i}, \quad g_{\nu}(x, p) = p^{i}g_{\nu, i}(x), \quad (6.3-2)$$

where $r=1, \ldots, 5$; $i=0, 1, 2, 3, \ldots L_i$, M_i are differential operators of the first order; L_i , M_i and $g_{\nu,i}(x)$ can be calculated from Eqs. (6.2-6), and (6.2-12), and they may consequently be regarded as known. We shall call α_n and $Y_{\mu,n}(x)$ perturbations of the order n for $\alpha(p)$ and $Y_{\mu}(x, p)$, respectively. The finite sums $p^i \alpha_i$ and $p^i Y_{\mu,i}$, $i=0,1,\ldots,n$, are called approximations of the order n for $\alpha(p)$ and $Y_{\mu}(x, p)$. The same denominations are used for L, M and $g_{\nu}(x, p)$. It should be observed that Greek subscripts are used for the numbering of the functions Y_1, Y_3 , \ldots ; g_1, g_2, \ldots , while Latin subscripts indicate the ordinal numbers of the perturbations of these functions, of α , and of the differential operators. The summation convention is applied to Latin subscripts and superscripts.

The following expressions for the zero order approximations are easily obtained:

$$L_0 = M_0 = D^2 + 1 + \alpha_0^2,$$

 $g_{\nu, 0}(x) \equiv 1, \ \nu = 1, \dots, 5.$
(6.3-3)

For the perturbations of the order i we have

$$\begin{split} & L_i = L'_{i-1} + (1 + \alpha_0^2) g_{1,i} + 2 \alpha_0 \alpha_i, \\ & M_i = M'_{i-1} + (1 + \alpha_0^2) g_{1,i} + 2 \alpha_0 \alpha_i, \end{split}$$

where L'_{i-1} and M'_{i-1} are (linear) differential operators of the first order with coefficients which are constituted by α_{i-1} and lower order perturbations of α and of the known functions $g_{\nu,k}(x)$, $k=0,1,\ldots,$ i-1. L_i and M_i are consequently linear in α_i , which is an important property.

Substituting our series expansions in the system Eq. (6.2-11) and arranging the terms in suitable order we get

$$\begin{aligned} p^{i}[L_{i-k}Y_{1,k} + 2\alpha_{k}g_{1,i-l-k}Y_{6,l} - g_{1,i-k}\sin xY_{3,k}] &= 0, \\ p^{i}[M_{i-k}Y_{6,k} + 2\alpha_{k}g_{1,i-l-k}Y_{1,l} - 4g_{4,i-k}\cos xY_{7,k}] &= 0, \\ p^{i}[DY_{7,i} - 2\alpha_{k}g_{5,i-l-k-1}\cos xDY_{1,l}] &= 0, \\ p^{i}[DY_{3,i} - 2\alpha_{k}g_{4,i-l-k}Y_{7,l}] &= 0, \end{aligned}$$

$$(6.3-5)$$

where we have summation on *i* from 0 to ∞ , on *k*, *l* from zero to *i*, and where functions $g_{\nu,n}(x) \equiv 0$ for n < 0; $\nu = 1, \ldots, 5$.

From the boundary conditions Eq. (6.2-13) we have

$$p^{i}Y_{\mu,i}(a_{1}) = p^{i}Y_{\mu,i}(a_{2}) = 0, \ \mu = 1, 6, 7.$$
 (6.3-6)

From the boundary condition Eq. (6.2-14) we finally obtain

$$\begin{array}{l} Y_{3, 0}(a_{1}) = C_{0}, \\ Y_{3, i}(a_{1}) = 0, \ i \geq 1. \end{array}$$
 (6.3-7)

The last equation is a consequence of our assumption that C_0 does not depend on p. The boundary value problem now consists in finding the perturbations α_i and $Y_{\mu,i}(x)$, $\mu=1,3,6,7$, of every order i. As implied previously, it should be understood that the functions $Y_{\mu,i}(x)$ shall be defined for $x \in [a_1, a_2]$. We now let our conditions be satisfied in such a way that the coefficients for p^i in the system Eq. (6.3-5) and the boundary conditions Eq. (6.3-6) are zero for every value of i. We then get a boundary value problem for every i.

6.32. The Zero Order Approximation

Putting i=0 in the system Eq. (6.3-5) we get

$$L_{0}Y_{1,0} + 2\alpha_{0}Y_{6,0} = \sin x Y_{3,0},$$

$$L_{0}Y_{6,0} + 2\alpha_{0}Y_{1,0} = 4\cos x Y_{7,0},$$

$$DY_{7,0} = 0,$$

$$DY_{3,0} = 2\alpha_{0}Y_{7,0},$$
(6.3-8)

where the second order differential operator L_0 is determined by Eq. (6.3-3). The boundary conditions become

$$Y_{\mu,0}(a_1) = Y_{\mu,0}(a_2) = 0, \ \mu = 1, 6, 7,$$

$$Y_{3,0}(a_1) = C_0.$$
(6.3-9)

It is evident that $Y_{3,0}$ and $Y_{7,0}$ satisfy⁴¹

$$\begin{split} &Y_{3,0}(x)\!\equiv\!C_0\,,\\ &Y_{7,0}(x)\!\equiv\!0. \end{split} \tag{6.3-10}$$

⁴¹ The condition Eq. (6.2-4) is obviously satisfied, if $Y_{7,0} \equiv 0$, since then $Y_7(x, p) = = O(p), p \rightarrow 0$.

It remains to find α_0 , $Y_{1,0}(x)$, $Y_{6,0}(x)$ such that they satisfy the differential system

$$L_{0}Y_{1,0} + 2\alpha_{0}Y_{6,0} = C_{0}\sin x,$$

$$L_{0}Y_{6,0} + 2\alpha_{0}Y_{1,0} = 0,$$
(6.3-11)

and the boundary conditions

$$Y_{\mu,0}(a_1) = Y_{\mu,0}(a_2) = 0, \ \mu = 1,6.$$
 (6.3-12)

This problem can be treated as a Sturm-Liouville problem. The eigenvalues λ_n and the normalized eigenfunctions $u_n(x)$, of the related Sturm-Liouville eigenvalue problem

$$\begin{array}{l} D^2 u \! + \! \lambda u \! = \! 0, \\ u(a_1) \! = \! u(a_2) \! = \! 0 \end{array} \tag{6.3-13}$$

are easily found to be

$$\lambda_n = \left(\frac{n\pi}{a_2 - a_1}\right)^2,$$

$$u_n(x) = \left(\frac{2}{a_2 - a_1}\right)^{1/2} \sin\left(\sqrt{\lambda_n} \left(x - a_1\right)\right),$$
(6.3-14)

where $n=1, 2, 3, \ldots$ As shown in Appendix A2, the functions $Y_{1,0}(x)$ and $Y_{6,0}(x)$ can be expanded in series of the orthonormal functions $u_n(x)$.⁴²

After some elementary calculations we obtain the following results for the zero order approximations α_0 , $Y_{\mu,0}(x)$, $\mu=1,3,6,7$:

$$Y_{1,0}(x) = C_0 \sum_{1}^{\infty} a_{0,n} u_n(x),$$

$$Y_{6,0}(x) = C_0 \sum_{1}^{\infty} b_{0,n} u_n(x),$$

$$Y_{3,0}(x) \equiv C_0,$$

$$Y_{7,0}(x) \equiv 0,$$

(6.3-15)

 42 These series become Fourier sine series. The same method applies also in the case of a general Sturm-Liouville operator instead of $L_0.$

where the expansion coefficients $a_{0,n}$, $b_{0,n}$ are determined by

$$\begin{aligned} a_{0,n} &= d_n E_n H_n, \\ b_{0,n} &= -2\alpha_0 d_n H_n, \\ d_n &= (u_n, \sin x) = \int_{a_1}^{a_1} u_n(x) \sin x dx, \\ E_n &= 1 + \alpha_0^2 - \lambda_n, \\ H_n &= (E_n^2 - 4\alpha_0^2)^{-1}. \end{aligned}$$
(6.3–16)

 λ_n and $u_n(x)$ are determined by Eq. (6.3-14). The computation of d_n is not carried out here; it is, however, an elementary operation. $(H_n)^{-1} = E_n^2 - 4x_0^2 \pm 0$ constitutes a necessary condition for the existence of $Y_{1,0}(x)$ and $Y_{6,0}(x)$. This condition can be written

$$(1+\alpha_0^2-\lambda_n)^2-4\alpha_0^2 \neq 0.$$
 (6.3-17)

The results can be summarized as follows:

1) No conditions except those of Eq. (6.3-17) are obtained for the set of zero order approximations α_0 of the natural frequencies α .

2) The functions $Y_{\mu,0}(x)$, $\mu=1,3,6,7$ are uniquely determined for given α_0 . $Y_{1,0}$ and $Y_{6,0}$ are obtained as sums of Fourier sine series and become proportional to C_0 . The latter series converge absolutely and uniformly, *cf.* [C4], p. 293.

3) It can immediately be concluded from Eq. (6.3-15) that neither $Y_{1,0}(x) \equiv 0$ nor $Y_{6,0}(x) \equiv 0$, $x \in [a_1, a_2]$, is possible. The vibration therefore cannot possibly take place in a plane, neither can it be purely circumferential.

We shall make no further comments on these results at present, but turn instead to the first order perturbation $Y_{7,1}(x)$. Putting i=1 we get from the third equation of the system Eq. (6.3-5) and the boundary conditions Eq. $(6.3-6)^{43}$

$$DY_{7,1} - 2x_0 \cos x \ DY_{1,0} = 0,$$

$$Y_{7,1}(a_1) = Y_{7,1}(a_2) = 0.$$
(6.3-18)

These equations determine $Y_{7,1}(x)$ and give a further condition.

⁴³ The third equation of the system simply implies conservation of string mass, *i.e.* string length, in the interval $\lceil a_1, a_2 \rceil$.

Integrating the first expression over the basic interval $[a_1, a_2]$ and regarding $Y_{1,0}(a_1) = Y_{1,0}(a_2) = 0$ we get⁴⁴

$$(Y_{1,0}, \sin x) = 0.$$
 (6.3–19)

Substituting the series for $Y_{1,0}(x)$ in Eq. (6.3–19), reversing the order of summation and integration, and employing the notation d_n , we get the following equation, expressing a necessary condition for the convergence of our perturbation method

$$\begin{split} S(\alpha_0) &= \sum_{n=1}^{\infty} d_n^2 E_n H_n = \\ &= \sum_{n=1}^{\infty} d_n^2 (1 + \alpha_0^2 - \lambda_n) [(1 + \alpha_0^2 - \lambda_n)^2 - 4\alpha_0^2]^{-1} = 0. \quad (6.3 - 20) \end{split}$$

This equation can be regarded as a condition for α_0 , and consequently every α_0 satisfying Eq. (6.3-20) is the zero order approximation of a natural frequency α . The coefficients d_n are easily obtained from their defining equation in Eq. (6.3-16). There exist two different cases depending on the length of the basic interval $[a_1, a_2]$:

a) If the basic interval is such that $a_2 - a_1 \neq k\pi$, k being an arbitrary, positive integer, we have

$$d_n = \sqrt{2} \,\lambda_n (a_2 - a_1)^{-1/2} (\lambda_n - 1)^{-1} \,(\sin a_1 - (-1)^n \sin a_2), \quad (6.3 - 21)$$

where, from Eq. (6.3-14), $\lambda_n = \left(\frac{n\pi}{a_2 - a_1}\right)^2$. b) If we have $a_2 - a_1 = k\pi$ we get

$$d_n = \sqrt{2 \lambda_n} (k\pi)^{-1/2} (\lambda_n - 1)^{-1} \sin a_1 (1 - (-1)^{n+k}), \ n \neq k,$$

$$d_k = \frac{1}{2} \sqrt{2 k\pi} \cos a_1,$$
 (6.3-22)

where $\lambda_n = -\frac{n}{k}$.

8

Equation $S(x_0)=0$ gives rise to the following comments:

1) The infinite series $S(\alpha_0)$ converges as Σn^{-4} .

2) $S(x_0)$ is an even function; the set of zero order approximations of the natural frequencies is therefore symmetric with respect to zero.

⁴⁴ The case $\alpha_0 = 0$ which provides $Y_{7,1} \equiv 0$ does not interest us, since it implies steady motion, cf. Eq. (6.2-9).

3) The condition Eq. (6.3-17) is automatically satisfied for those α_0 which satisfy $S(\alpha_0)=0$.

4) The sum of the series defining $S(\alpha_0)$ can be obtained by a method known from the theory of functions of a complex variable, *cf*. [T1], p. 114. The result is a finite sum of products of trigonometric functions (of α_0 , a_1 and a_2) and algebraic functions (of α_0).

5) The author would advance the conjecture that the set of real x_0 satisfying $S(x_0)=0$ is enumerably infinite and has no other point of accumulation than infinity. At least this comes true in special cases of case (b). In the general case (a) the evaluation of $S(x_0)$ is rather tedious and it will not be reproduced here.

We have now succeeded in finding the zero order approximations α_0 , $Y_{\mu,0}(x)$, $\mu=1,3,6,7$ of our boundary value problem, at least in principle. We now turn to the problems involved in the determination of the pertubations of the first and higher orders.

6.33. Pertubations of the First and Higher Orders

We shall first approach the determination of the first order perturbations α_1 , $Y_{\mu,1}(x)$, $\mu = 1,3,6$.

The perturbation $Y_{7,1}(x)$ is already known from Eq. (6.3-18) and it is clear that it is proportional to C_0 , since this is the case of $Y_{1,0}(x)$. In the system Eq. (6.3-5) we put i=1 in the first, second and fourth equations. In the third equation of the system we put i=2. The following system results:

$$\begin{split} L_{0}Y_{1,1} + & 2x_{0}Y_{6,1} = -2x_{1}(x_{0}Y_{1,0} + Y_{6,0}) + \sin xY_{3,1} + \varPhi_{1,0}, \\ L_{0}Y_{6,1} + & 2x_{0}Y_{1,1} = -2x_{1}(x_{0}Y_{6,0} + Y_{1,0}) + 4\cos xY_{7,1} + \varPhi_{2,0}, \\ DY_{3,1} = & 2x_{0}Y_{7,1} + & 2x_{1}Y_{7,0} + \varPhi_{3,0}, \\ DY_{7,2} = & 2\cos x(x_{1}DY_{1,0} + & x_{0}DY_{1,1}) + \varPhi_{4,0}, \end{split}$$
(6.3–23)

where we let $Y_{7,0}(x)=0$ conformably to Eq. (6.3-10).

The following boundary conditions have to be satisfied:

$$\begin{split} &Y_{\mu,1}(a_1) = Y_{\mu,1}(a_2) = 0, \ \mu = 1,6, \\ &Y_{3,1}(a_1) = 0, \\ &Y_{7,2}(a_1) = Y_{7,2}(a_2) = 0. \end{split} \tag{6.3-24}$$

114

The functions $\Phi_{\eta,0}, \eta=1, \ldots, 4$ in the system Eq. (6.3-23) are known functions of x and x_0 and proportional to C_0 , they consequently do not contain α_1 or the unknown functions $Y_{7,2}(x)$ and $Y_{\mu,1}(x)$, $\mu=1,3,6$. Evidently we have one system of this kind for every α_0 in the set of zero order approximations of the natural frequencies. The differential operators occurring in the left members of the differential system Eq. (6.3-23) are wholly identical to those in the system Eq. (6.3-8) which governs the zero order approximations.

Regarding α_1 as an arbitrary parameter the system can be integrated step by step. The right member of the third equation in the system (6.3-23) contains only known quantities. Taken together with the boundary condition $Y_{3,1}(a_1)=0$ we are able to obtain a unique determination of $Y_{3,1}(x)$ which becomes proportional to C_0 and does not depend on α_1 . Now, the right members of the first two equations of the differential system Eq. (6.3-23) are known (if we consider α_1 as known). We can therefore consider these two equations and the boundary conditions as a Sturm-Liouville problem for the unknown functions $Y_{1,1}(x)$ and $Y_{6,1}(x)$. We can write the solutions in the form

$$Y_{1,1}(x) = \alpha_1 Y'_{1,1} + Y''_{1,1},$$

$$Y_{6,1}(x) = \alpha_1 Y'_{6,1} + Y''_{6,1},$$
(6.3-25)

where the functions $Y'_{\mu,1} = Y'_{\mu,1}(x)$ and $Y''_{\mu,1} = Y''_{\mu,1}(x)$, $\mu = 1,6$, do not depend on x_1 . These functions can be expressed in terms of absolutely and uniformly convergent series of the eigenfunctions $u_n(x)$ of the problem Eq. (6.3-13). At the moment we shall take no interest in the explicit computation of these series. It now remains to determine x_1 . The last equation of the system (6.3-23) together with the boundary conditions $Y_{7,2}(a_1) = Y_{7,2}(a_2) = 0$ determine the functions $Y_{7,2}(x)$ and also provide a further condition which is a linear algebraic equation for x_1 . Integrating the differential equation over the basic interval $[a_1, a_2]$ and taking into account the boundary conditions, we obtain⁴⁵

$$2\alpha_{1}(Y_{1,0}, \sin x) + 2\alpha_{0}\alpha_{1}(Y'_{1,1}, \sin x) + + 2\alpha_{0}(Y''_{1,1}, \sin x) + (\varPhi_{4,0}, 1) = 0.$$
 (6.3-26)

⁴⁵ It should be observed that $Y'_{1,1}$ and $Y''_{1,1}$ are both zero at the end points of the basic interval, a_1 and a_2 .

The first term is zero according to Eq. (6.3-19), and Eq. (6.3-26) is a linear equation for α_1 since none of the inner products depend on α_1 . We get a value α_1 for every particular α_0 in the set of zero order approximations of the natural frequencies, provided that the following condition is satisfied:

$$(Y'_{1,1}, \sin x) \neq 0.$$
 (6.3–27)

This condition is extremely important to our perturbation method, and we shall therefore try to give it a more explicit formulation. First we must determine the function $Y'_{1,1}(x)$. Substituting the expressions of Eq. (6.3-25) in the first two equations of the system Eq. (6.3-23) and extracting the terms depending on α_1 (observe that $Y_{3,1}(x)$ and $Y_{7,1}(x)$ do not depend on α_1), we get the following equations for $Y'_{1,1}(x)$ and $Y'_{6,1}(x)$:

$$\begin{split} & L_0 Y'_{1,1} + 2 \alpha_0 Y'_{6,1} = -2(\alpha_0 Y_{1,0} + Y_{6,0}), \\ & L_0 Y'_{6,1} + 2 \alpha_0 Y'_{1,1} = -2(\alpha_0 Y_{6,0} + Y_{1,0}). \end{split} \tag{6.3-28}$$

The boundary conditions to be satisfied are

$$Y'_{\mu,1}(a_1) = Y'_{\mu,1}(a_2) = 0, \ \mu = 1,6.$$
 (6.3–29)

This Sturm-Liouville problem is essentially the same as the previous one which determined $Y_{1,0}(x)$ and $Y_{6,0}(x)$. After some calculations we get according to the results elaborated in Appendix A2

$$Y'_{1,1}(x) = \sum_{n=1}^{\infty} a'_{1,n} u_n(x), \qquad (6.3-30)$$

where

$$\begin{aligned} &a_{1,n}' = H_n(E_n d_{1,n}' - 2x_0 d_{2,n}'), \\ &d_{1,n}' = -2x_0(Y_{1,0}, u_n) - 2(Y_{6,0}, u_n), \\ &d_{2,n}' = -2x_0(Y_{6,0}, u_n) - 2(Y_{1,0}, u_n). \end{aligned}$$
(6.3-31)

In Eq. (6.3-31) E_n and H_n are determined by Eq. (6.3-16). The coefficients $d'_{1,n}$ and $d'_{2,n}$ are the expansion coefficients of the right

members of Eq. (6.3-28) with respect to u_n . It can be concluded from Eqs. (6.3-15) and (6.3-16) that

$$(Y_{1,0}, u_n) = C_0 a_{0,n} = C_0 E_n H_n d_n,$$

$$(Y_{6,0}, u_n) = C_0 b_{0,n} = -2C_0 \alpha_0 H_n d_n.$$

After some calculation we get

$$a_{1,n}' = -2C_0 \alpha_0 H_n^2 d_n (E_n^2 - 4E_n + 4\alpha_0^2), \qquad (6.3 - 32)$$

where $d_n = (u_n, \sin x)$ according to Eq. (6.3-16). From the condition Eq. (6.3-27) and from Eq. (6.3-30) we obtain by termwise integration

$$\sum_{n=1}^{\infty} a'_{1, n}(\sin x, u_n) = \sum_{n=1}^{\infty} a'_{1, n} d_n \neq 0.$$

Substituting Eq. (6.3-32) in the last expression and deleting some non-essential factors we obtain

$$T(\alpha_0) \equiv \sum_{n=1}^{\infty} d_n^2 H_n^2 (E_n^2 - 4E_n + 4\alpha_0^2) \neq 0$$
 (6.3-33)

as the condition for the existence of the first order perturbations x_1 . In the inequality (6.3-33) E_n , H_n are given by Eq. (6.3-16), and d_n is given by Eq. (6.3-21) or Eq. (6.3-22).

The following comments are relevant to the above considerations:

1) The infinite series in Eq. (6.3-33) converges as Σn^{-6} .

2) The sum $T(\alpha_0)$ of the series can be obtained in the same way as $S(\alpha_0)$, cf. section 6.32. When $T(\alpha_0)$ is obtained in this way it has the form of a finite sum of products of trigonometric functions (of α_0 , a_1 and a_2) and algebraic functions (of α_0).

3) It is a necessary condition for our perturbation method to work in the case of a particular value of x_0 that $T(x_0) \neq 0$. If it happens that the set of real zeros of $S(x_0)$ and the set of real zeros of $T(x_0)$ are disjoint (for given boundary coordinates a_1 and a_2), then it is possible to determine all the first order perturbations x_1 . It is the author's conjecture that this comes true, since it seems most unlikely that the two functions $S(x_0)$ and $T(x_0)$ should possess coincidental zeros. The proof of this conjecture seems to constitute a difficult problem. Summarizing our results regarding the first order perturbations, we assert that x_1 , $Y_{\mu,1}(x)$, $\mu=1,3,6$ do exist and are uniquely determinable, if the condition $T(x_0) \neq 0$ is satisfied. Then the second order perturbation $Y_{7,2}(x)$ also does exist and is uniquely determinable.

We now turn to the perturbations of the arbitrary order i for α , $Y_{\mu}(x)$, $\mu=1,3,6$, and the order i+1 for $Y_{7}(x)$. Assuming that all lower order perturbations are determined, we get from system Eq. (6.3-5)

$$L_{0}Y_{1,i} + 2\alpha_{0}Y_{6,i} = -2\alpha_{i}(\alpha_{0}Y_{1,0} + Y_{6,0}) + \sin xY_{3,i} + \Phi_{1,i-1},$$

$$L_{0}Y_{6,i} + 2\alpha_{0}Y_{1,i} = -2\alpha_{i}(\alpha_{0}Y_{6,0} + Y_{1,0}) + 4\cos xY_{7,i} + \Phi_{2,i-1},$$

$$DY_{3,i} = 2\alpha_{0}Y_{7,i} + \Phi_{3,i-1},$$

$$DY_{7,i+1} = 2\cos x(\alpha_{i}DY_{1,0} + \alpha_{0}DY_{1,i}) + \Phi_{4,i-1},$$
(6.3-34)

where the functions $\Phi_{\eta, i-1}, \eta=1, \ldots, 4$ are known functions of x and of perturbations α_k up to and including the order i-1. The boundary conditions to be satisfied are the same as those for the first order perturbations, *i. e.* the conditions Eq. (6.3-24) are valid with obvious changes of the second subscripts.

The system Eq. (6.3-34) is essentially the same as Eq. (6.3-23), and it is not difficult to show that all the important results obtained for the first order perturbations also apply to perturbations of arbitrary order. In particular, the condition $T(\alpha_0) \neq 0$ is the condition of existence of the perturbation α_i . The latter fact is easy to prove from the system Eq. (6.3-34), since the term depending on α_i in the function $Y_{1,i}(x)$ will become exactly the same as the term depending on α_1 in the function $Y_{1,1}(x)$, *i. e.* equal to $Y'_{1,1}(x)$.

We now conclude our discussion of the perturbation method applied to the vibration problem as set in section 6.2 with the following summary:

It is possible to obtain perturbations of arbitrary order of a particular natural frequency α and the functions $Y_{\mu}(x)$, $\mu=1,3,6,7$ provided that the condition $T(\alpha_0) \neq 0$ is valid for the zero order approximation α_0 of the natural frequency in question. The zero order approximations α_0 are the real roots of $S(\alpha_0)=0$. $S(\alpha_0)$ is defined by Eq. (6.3-20) and $T(\alpha_0)$ by Eq. (6.3-33). The zero order approximations and the first order perturbations are not difficult to calculate for given boundary coordinates a_1 and a_2 . The explicit calculation of higher order perturbations will be laborious. We do not know whether the perturbation

method converges or not. Our results are consequently only formal although the perturbations are determined without ambiguity. It is to be expected that the perturbation method discussed here will apply to more general problems. One might, for instance, study a small time-dependent motion of an inextensible string which is initially in a state of steady rotatory motion with no tangential velocity and initially possessing a plane string curve. The perturbations causing the timedependent motion may be acting on one end of the string or along the string and may depend on the time, they may, for instance, be periodic functions of the time. In its initial state the string may be deflected from its steady state curve and possess a velocity in relation to a steady rotating frame. Starting from the system Eq. (3.2-10), as we did in the vibration problem, we can apply the Laplace transformation to the time variable (instead of separating the variables), and then apply the perturbation method to the resulting ordinary differential system.

Our choice to study a vibratory motion as an illustration of this kind of motions is based on the fact that it is the simplest time-dependent motion to be found.

Appendix A1

In section 4.32 we solved the problem (E1). The solution was regarded as an approximation, linear in the elastic constant β and valid for small values of β , of the solution of the basic one-point problem for the steadily rotating and linearly elastic string, when the string does not intersect the axis of rotation. The functions constituting the approximate solution were denoted $\xi(z, \beta), \varphi(z, \beta), \gamma(z, \beta)$ and the functions to be approximated were denoted $\xi'(z, \beta) \varphi'(z, \beta), \gamma'(z, \beta)$. In this appendix we shall prove that the conditions Eq. (4.3-9) are satisfied, *i. e.* that

$$\xi - \xi', \varphi - \varphi', \gamma - \gamma' = O(\beta^2), \quad \beta \to 0$$
 (A1-1)

hold true if z occurs in an arbitrary, bounded interval. The functions ξ , φ , γ satisfy the differential system Eq. (4.3–15) and the initial values $\xi(0, \beta)=1, \varphi(0, \beta)=0$. The functions ξ', φ', γ' satisfy the differential system Eq. (4.3–12) and the same initial values. It was shown that ξ and ξ' are periodic in z and have positive upper and lower bounds. Furthermore, all the functions discussed are analytic in z and in the parameters of the systems for every z on the real axis, on condition that β is small enough. This is true, because the system Eq. (4.3–12) and (4.3–15) is analytic in its variables and parameters for $\xi > 0$ and sufficiently small values of β . For the moment we assume that $\xi(z, \beta)$ and $\xi'(z, \beta)$ satisfy the condition in Eq. (A1–1) for some arbitrary z interval (a_1, a_2) . We then obtain

$$\xi^2 - (\xi')^2 = (\xi + \xi')(\xi - \xi') = O(\beta^2), \quad \beta \to 0$$
 (A1-2)

and from Eqs. (4.3-12), (4.3-15)

$$|(\varphi - \varphi')_z| = |b - p_2| |\xi^{-2}(\xi')^{-2}| |\xi^2 - (\xi')^2| < M_1 \beta^2.$$
 (A1-3)

 M_1 is a positive number which does not depend on z, and the inequality holds if β is sufficiently small. By the initial conditions $\varphi(0, \beta) =$

 $=\varphi'(0, \beta)=0$ we have at once $|\varphi-\varphi'| < |z|M_1\beta^2$ or, equivalently, $\varphi-\varphi'=O(\beta^2), \beta \rightarrow 0$ for $z \in (a_1, a_2)$. From Eqs. (4.3–12), (4.3–15) we also get $\gamma-\gamma'=O(\beta^2), \beta \rightarrow 0$.

It remains to show that $\xi - \xi' = O(\beta^2)$, $\beta \to 0$, is valid. The first equations of the two systems (4.3-12) and (4.3-15) may be written

$$(\xi'\xi'_z)^2 = [k_2(n_2 + \lambda K) - (\xi')^2][1 + \delta G_3]Q(\xi',\beta), \qquad (A1-4)$$

$$(\xi\xi_z)^2 = (k_2n_2 - \xi^2)Q(\xi, \beta),$$
 (A1-5)

where $\lambda = \delta = \beta^2$. $K = K(\beta)$ is analytic in β , and $G_3 = G_3(\xi, \beta)$ is analytic in z and β . Moreover, K = O(1), $G_3 = O(1)$, $\beta \rightarrow 0$. $Q(\xi', \beta)$ is a polynomial of the third degree in $(\xi')^2$ and analytic in β . Regarding λ and δ as parameters of the equation and adding the initial value $\xi'(0, \beta, \lambda, \delta) = 1$ to Eq. (A1-4) we have an initial value problem which determines the function $\xi' = \xi'(z, \beta, \lambda, \delta)$. This function becomes periodic in z; analytic in z and in the parameters in the neighbourhood of $\beta = \lambda = \delta = 0$, for every real z. We can consequently expand ξ' in a power series of β, λ, δ with coefficients which are analytic in z. We then obtain

$$\begin{aligned} \xi'(z,\,\beta,\,\lambda,\,\delta) = &f(z) + \beta g_1(z) + \lambda g_2(z) + \delta g_3(z) + \\ &+ O(\beta^2) + O(\beta\lambda) + O(\beta\delta), \quad \beta,\,\lambda,\,\delta \to 0, \end{aligned}$$
 (A1-6)

where f, g_1, g_2, g_3 are analytic. With $\xi(0, \beta) = 1$, we get from Eq. (A1-5)

$$\xi(z,\beta) = f(z) + \beta g_1(z) + O(\beta^2), \quad \beta \to 0. \tag{A1-7}$$

Evidently the functions f and g_1 in Eq. (A1-7) are the same as in Eq. (A1-6), since for $\lambda = \delta = 0$ $\xi' = \xi$ must hold for every z and β of interest. (The differential equations (A1-4) and (A1-5) coincide for $\lambda = \delta = 0$, and ξ and ξ' satisfy the same initial value.)

From Eqs. (A1-6) and (A1-7) we obtain

$$\xi - \xi' = -\lambda g_2(z) - \delta g_3(z) + O(\beta^2) + O(\beta\lambda) + O(\beta\delta).$$

Recalling that $\lambda = \delta = \beta^2$ we readily get the desired expression $\xi - \xi' = O(\beta^2), \beta \rightarrow 0$, which holds for every real z.

It is also concluded that the derivatives (with respect to z), of arbitrary orders, of the functions $\xi - \xi'$, $\varphi - \varphi'$, $\gamma - \gamma'$ are of the order $O(\beta^2)$.

Appendix A2

In section 6.32 we were faced with the problem of solving the following boundary value problem for two functions $y_1(x)$ and $y_2(x)$:

$$L_{0}y_{1}+2\alpha y_{2}=C \sin x,$$

$$L_{0}y_{2}+2\alpha y_{1}=0,$$

$$y_{i}(a_{1})=y_{i}(a_{2})=0, i=1,2,$$
(A2-1)

where α and C are real constants and the differential operator L_0 is defined by

 $L_0 = D^2 + 1 + \alpha^2,$ $D = \frac{d}{dx}.$

Problem (A2-1) is a special case of the problem

$$Ly_{1} + \alpha_{1}\varrho y_{2} = f_{1},$$

$$Ly_{2} + \alpha_{2}\varrho y_{1} = f_{2},$$

$$y_{i}(a) = y_{i}(b) = 0, \ i = 1, 2.$$
(A2-2)

 α_1, α_2 are real constants, $f_1 = f_1(x)$ and $f_2 = f_2(x)$ are arbitrary, continuous functions of x, and L is a Sturm-Liouville operator defined by

$$L = D(pD) + \lambda \rho, \tag{A2-3}$$

where p=p(x) and $\varrho=\varrho(x)$ are positive and continuous for $x\in[a, b]$. λ is a real parameter. Essentially, the following derivations remain valid, if we change L for the general Sturm-Liouville operator D(pD)+ $+\lambda \varrho - q$, where q=q(x) is a continuous function. For obvious reasons, we may call the problem (A2-2) a Sturm-Liouville problem. The author has not been able to find any treatment of such a problem in the literature, although it is quite possible that such a treatment in fact does exist, and we therefore find it worth-while to discuss it in some detail. Let $u_n(x)$ and λ_n , $n=1, 2, 3, \ldots$, be the normalized eigenfunctions and the eigenvalues of the related problem

$$Lu=0,$$

 $u(a)=u(b)=0.$ (A2-4)

With the notation $(f,g) \equiv \int_{a}^{b} \varrho(x)f(x)g(x)dx$ we then get by the orthonormality of the eigenfunctions

$$(u_n, u_k) = \begin{cases} 0 \text{ if } n \neq k, \\ 1 \text{ if } n = k. \end{cases}$$
(A2-5)

Let $c_{1,n}$ and $c_{2,n}$ be the wanted expansion coefficients, with respect to the eigenfunctions, for the unknown functions $y_1(x)$ and $y_2(x)$, *i. e.* let us write

$$y_i = c_{i,n} u_n,$$

 $c_{i,n} = (y_i, u_n),$ (A2-6)

where i=1,2 and we have summation on n from one to infinity. Multiplying the differential equations in Eq. (A2-2) by u_n and integrating over the basic interval we get

$$\begin{aligned} \int u_n [D(pDy_1) + \lambda_{\varrho} y_1] dx + \alpha_1 \int \varrho y_2 u_n dx = \int f_1 u_n dx, \\ \int u_n [D(pDy_2) + \lambda_{\varrho} y_2] dx + \alpha_2 \int \varrho y_1 u_n dx = \int f_2 u_n dx. \end{aligned}$$
 (A2-7)

Now, from Eq. (A2-4) and the boundary conditions $y_i(a) = y_i(b) = 0$, and by repeated integration by parts, we obtain

$$\int u_n D(pDy_i) dx = -\lambda_n(y_i, u_n), \ i=1,2.$$

With the notation

$$\int f_i u_n dx = d_{i_n}, i = 1,2$$
 (A2-8)

we have from (A2-7)

$$\begin{aligned} &(\lambda - \lambda_n) c_{1, n} + \alpha_1 c_{2, n} = d_{1, n}, \\ &(\lambda - \lambda_n) c_{2, n} + \alpha_2 c_{1, n} = d_{2, n}. \end{aligned}$$
 (A2-9)

From (A2-9) we get immediately

$$c_{1,n} = H_n[d_{1,n}(\lambda - \lambda_n) - \alpha_1 d_{2,n}],$$

$$c_{2,n} = H_n[d_{2,n}(\lambda - \lambda_n) - \alpha_2 d_{1,n}],$$
(A2-10)

where

$$H_n = [(\lambda - \lambda_n)^2 - \alpha_1 \alpha_2]^{-1}.$$
 (A2-11)

The equations (A2-10) determine the expansion coefficients $c_{1,n}$ and $c_{2,n}$ if H_n is finite, *i. e.* if

$$(\lambda - \lambda_n)^2 - \alpha_1 \alpha_2 \neq 0. \tag{A2-12}$$

The method can without difficulty be extended to the following Sturm-Liouville problem for n functions $y_i(x)$:

$$Ly_i + \varrho \alpha_{i, k} y_k = f_i,$$

$$y_i(a) = y_i(b) = 0,$$

where $i=1,2,\ldots,n$; $k=1,2,\ldots,n$; $k\neq i$, and we have summation on k. $\alpha_{i,k}$ are arbitrary real constants and L, ϱ , f_i are defined as for problem (A2-2).

Returning to problem (A2-1) we find that this special case of problem (A2-2) is characterized by

$$L = L_0 = D^2 + 1 + \alpha^2, \ \lambda = 1 + \alpha^2, \ \alpha_1 = \alpha_2 = 2\alpha,$$

$$\rho(x) \equiv 1, \ f_1(x) = C \sin x, \ f_2(x) \equiv 0.$$
(A2-13)

The basic interval is $[a_1, a_2]$. The eigenvalues λ_n and eigenfunctions $u_n(x)$ of the corresponding problem Eq. (A2-4) are easily obtained, and the determination of the expansion coefficients $c_{1,n}$ and $c_{2,n}$ result from straight-forward calculations.

124

Bibliography

- [B1] BEGHIN, H., Cours de mécanique théorique et appliquée. Paris 1952.
- [B2] BYRD, P. F., FRIEDMAN, M. D., Handbook of elliptic integrals for engineers and physicists. Berlin-Göttingen-Heidelberg 1954.
- [B3] BECKERT, H., Über quasilineare hyperbolische Systeme partielle Differentialgleichungen erster Ordnung mit zwei unabhängigen Variablen. Ber. Verh. Sächs. Akad. Wiss. Leipzig, Math. — Nat. Kl., 97 (1950), no. 5.
- [C1] COURANT, R., FRIEDRICHS, K. O., Supersonic flow and shock waves. New York 1948.
- [C2] COURANT, R., LAX, P., On nonlinear partial differential equations with two independent variables. Comm. Pure Appl. Math., 2 (1949), 255-273.
- [C3] COURANT, R., HILBERT, D., Methoden der mathematischen Physik. Bd. II. Berlin 1937.
- [C4] COURANT, R., HILBERT, D., Methods of mathematical physics. Vol. I. New York 1953.
- [C5] CAUGHEY, T. K., Whirling of a heavy chain. Proc. Third U. S. Nat. Congr. Appl. Mech., 1958, 101-108.
- [C6] CINQUINI CIBRARIO, M., Sopra la teoria delle caratteristiche per i sistemi di equazioni quasilineari alle derivate parziali del primo ordine. Ann. Scuola Norm. Super. Pisa (3), 3 (1949), 161-197.
- [D1] DETRY, J., La détermination théorique de la forme et des tensions du ballon dans la filature au continu à anneaux. Revue Textilis, 12 (1956), 17-27, 32-40, 50-59.
- [E1] ERICKSEN, J. L., TRUESDELL, C., Exact theory of stress and strain in rods and shells. Arch. Ratl. Mech. Anal., 1 (1957/58), 295-323.
- [H1] HILL, R., The mathematical theory of plasticity. Oxford 1950.
- [H2] HORN, J., Gewöhnliche Differentialgleichungen beliebiger Ordnung. Leipzig 1905.
- [H3] HAMEL, G., Theoretische Mechanik. Berlin-Göttingen-Heidelberg 1949.
- [H4] HAMEL, G., Über die Bewegung eines unausdehnbaren Fadens. Mat. Z., 48 (1942), 27-47.
- [H5] HAMEL, G., Über ein für die Textiltechnologie wichtiges Problem aus der Mechanik dünner Fäden. Ing. Arch., 4 (1933), 84-97.
- [H6] HALL, H. W., HUNTER, W., On the form assumed by a steadily rotating thread. Phil. Mag., VII Series, 25 (1938), 193-234.
- [K1] KAMKE, E., Differentialgleichungen. Bd. 2. Leipzig 1944.
- [K2] KOLODNER, I. I., Heavy rotating string a non-linear eigenvalue problem. Comm. Pure Appl. Math., 8 (1955), 395-408.
- [L1] LAX, P., Nonlinear hyperbolic equations. Comm. Pure Appl. Math., 6 (1953), 231-258.

- [M1] MACK, C., Theory of the spinning balloon. Quart. J. Mech. Appl. Math., XI (1958), 196-207.
- [M2] MACK, C., Theory of winding in centrifuge spinning. Quart. J. Mech. Appl. Math., IX (1956), 75-83.
- [M3] MCCONNEL, A. J., Application of tensor analysis. New York 1957. (Published under the title 'Applications of the absolute differential calculus'. Glasgow 1947.)
- [N1] NERONOFF, N., Sur les positions d'équilibre dans un problème de dynamique. Comm. Inst. Sci. Math. Méc. Univ. Kharkoff, IV (1938), 89-97.
- [P1] PAILLOUX, H., Sur le mouvement d'un fil où glisse un petit anneau. C. R. 208 (1939), 325-327.
- [P2] PETROWSKI, I. G., Vorlesungen über die Theorie der gewöhnlichen Differentialgleichungen. Leipzig 1954.
- [P3] PAILLOUX, H., Contribution à l'étude des systèmes déformables. Ann. Fac. Sci. Univ. Toulouse, IV^e Série, 1 (1937), 1-125.
- [R1] LORD RAYLEIGH, Theory of sound. Vol. I. London 1944.
- [R2] ROUTH, E. J., Dynamics of a system of rigid bodies. 6th ed. New York 1955.
- [S1] SAUER, R., Anfangswertprobleme bei partiellen Differentalgleichungen. 2. Aufl. Berlin-Göttingen-Heidelberg 1958.
- [S2] SAKS, S., ZYGMUND, A., Analytic functions. Warszawa-Wrocław 1952.
- [S3] SZABO, I., Höhere technische Mechanik. 2. Aufl. Berlin-Göttingen-Heidelberg 1958.
- [S4] STEVENSON, A. C., A note on the vibrations of a rotating rope. Phil. Mag., VII Series, 24 (1937), 293-299.
- [T1] TITCHMARSH, E. C., The theory of functions. Oxford 1932.
- [T2] TSIEN, H. S., The Poincaré Lighthill Kuo method. Adv. Appl. Mech., IV (1956), 281-349.
- [W1] WRIGHT, C. E., Note on the paper 'A note on the vibrations of the rotating rope'. Phil. Mag., VII Series, 25 (1938), 387-388.
- [Z1] ŽDANOVIČ, V. F., Solution by the Fourier method of non selfadjoint mixed problems for hyperbolic systems on the plane I. Mat. Sb. (N. S.), 47 (1959), 307-354.

126





GÖTEBORG Elanders boktryckeri aktiebolag 1961